NAESB Retail Quadrants Open Industry Meetings

January 29, 2002

Washington DC



NAESB Retail Quadrants Open Industry Meeting Washington DC – January 29, 2002 Table of Contents

Tab 1. Welcome & Opening Remarks

- Meeting Announcement and Background
- Proposed Agenda
- Meeting Specifics & Directions
- Meeting Participants
- Antitrust Guidelines

Tab 2. Review of 1/17/02 Meeting

- Agenda
- Meeting Participants
- Meeting Minutes
- Summary of 1/17/02 Comments

Tab 3. Joint Retail Gas & Retail Electric Working Group Activity Summary

- Participant List
- 1/22/02 Conference Call Agenda
- 1/22/02 Conference Call Minutes
- 1/22/02 Work Paper 1: Reformatted REQ Procedures
- 1/22/02 Work Paper 2: Reformatted RGQ Procedures
- 1/22/02 Work Paper 3: NAESB Bylaws Organization

Tab 4. Proposed NAESB Retail Electric Quadrant Procedures

- Revised Retail Electric Quadrant Procedures v2.2 1/24/02 DRAFT
- REQ Procedures Comments Tracking Summary v2.2 1/24/02
- REQ Procedures 1/13/02 DRAFT

Tab 5. Proposed NAESB Retail Gas Quadrant Procedures

- Revised Retail Gas Quadrant Procedures 1/25/02 DRAFT
- RGQ Procedures Revisions Tracking Summary 1/25/02
- RGQ Procedures 1/4/02 DRAFT

Tab 6. Proposed NAESB REQ Segments Procedures

- REQ Supplier Segment Procedures 12/12/01 DRAFT
- REQ Distributor Segment Procedures 12/4/01 DRAFT
- REQ End User Segment Procedures 12/21/01 DRAFT
- REQ Services Segment Procedures 1/14/02 DRAFT

Tab 7. Proposed NAESB RGQ Segments Procedures

- RGQ Supplier Segment Procedures 12/12/01 DRAFT
- RGQ Distributor Segment Procedures 1/25/02 REVISED
- RGQ End User Segment Procedures 12/21/01 DRAFT
- RGQ Services Segment Procedures 1/14/02

Tab 8. Next Steps

- NAESB Letter of Intent
- NAESB Application with Revocable Proxy
- Contacts for NAESB Quadrants and Segments Formation

TAB 1

WELCOME & OPENING REMARKS



Please Save the Date

January 17, 2002 and January 29, 2002 NAESB Retail Quadrants Open Industry Meeting 9:30 AM - 5:00 PM EDT

Two meetings will be held in January in Washington DC, hosted by AGA, for all industry stakeholders to coordinate efforts related to the creation of the North American Energy Industry Standards Board (NAESB). The meetings will focus on structure and governance for the Retail Gas Quadrant, the Retail Electric Quadrant and the Segments of each of the two Quadrants. These efforts are aimed at organizing the two Quadrants of NAESB by March 2002.

<u>January 17, 200</u>2: The purpose is 1) to introduce interested participants to the bylaws of NAESB and the Retail Gas Quadrant Procedures proposed by AGA and APGA and the Retail Electric Quadrant proposed UBP-RSB; and 2) to develop consensus on Quadrant procedures.

<u>January 29, 200</u>2: The objective of this meeting would be to 1) work on developing Retail Gas and Retail Electric Quadrants procedures; 2) review and incorporate Segment procedures for each of the Retail Quadrants; and 3) discuss the transition process toward forming the two Retail Quadrants of NAESB.

All stakeholders and interested parties are invited to attend and/or participate. Teleconferencing capabilities will be offered to those unable to attend in person. Included in this packet is a summary of recent developments related to this endeavor, a preliminary agenda, registration form, and information on meeting logistics. Dial-in specifics, a final meeting agenda, and other meeting materials will be distributed to registrants by the close of business, January 4, 2002.

Please RSVP by <u>January 4, 200</u>2. Please return the attached registration form to Lucy Castaneda-Land by FAX at 202-824-7087. We hope to see you in Washington DC in January. Meeting Location:

> American Gas Association 400 N. Capitol Street, 4th Floor Washington DC 20001 Phone: 202-824-7126 FAX: 202-824-7087

Background on Recent Developments Leading up to this Meeting

On December 5, 2001 the GISB Board of Directors approved bylaws for the new North American Energy Standards Board (NAESB), which will be formed on January 1, 2002. The NAESB bylaws provide for four industry branches to operate within the new organization, representing quadrants of the energy industry—wholesale gas, retail gas, wholesale electric, and retail electric. By passing this resolution, the GISB Directors expect to have both Retail Gas and Retail Electric quadrants formed by March 2002 and operating by June 2002. Thus to meet the March 2002 deadline, all stakeholders of the Retail Gas Quadrant and of the Retail Electric Quadrant will have to complete the following steps:

- 1. All stakeholders will complete Quadrant procedures as addendum to the NAESB bylaws, which the NAESB Board would consider for approval at its March 2002 meeting.
- 2. The Segments of each Quadrant (e.g. Suppliers, Distributors, End Users and Service Providers) will develop Segment procedures, to be incorporated into the Quadrant addendum.
- 3. Each of the Quadrants, with the NAESB office, will conduct a 30-day Membership recruitment drive.
- 4. Once a minimum threshold of Members is attained (that being, forty per Quadrant and five per Segment, according to the NAESB bylaws), the Segments of each Quadrant will elect their representatives to the Quadrant Executive Committee and to the NAESB Board of Directors.
- 5. The newly elected NAESB Board members are expected to join their counterparts from the Wholesale Gas Quadrant to approve Quadrant and Segment Procedures at the March 2002 Board meeting, or during a conference call soon after.

To meet those milestones for a Retail Gas and Retail Electric quadrant within the timeline envisioned by the GISB leadership, two open industry meetings will be held, on January 17 and January 29, 2002, in Washington DC. The meetings will allow stakeholders to begin coordinating activities for forming the two NAESB retail quadrants.

NAESB Retail Quadrants Open Industry Meeting Washington DC – AGA Offices January 29, 2002 9:30 AM – 5:00 PM EDT

Proposed Agenda

A conference bridge will be provided to meeting registrants who are unable to attend in person.

<u>Meeting Purpos</u>e: 1) Review Retail Gas and Retail Electric Quadrants procedures; 2) present Segment procedures for each of the Retail Quadrants; and 3) discuss the transition process toward forming the two Retail Quadrants of NAESB. Jim Buccigross, Vice President - Energy Industry, Group 8760 LLC, will moderate the meeting.

- 9:30 a.m. Registration and Continental breakfast
- 9:45 a.m. Welcome & Administrative Matters
 - Antitrust Guidelines
 - Participant/caller guidelines
 - Brief Introductions
- 10:00 a.m. Brief Review of January 17, 2002 NAESB Retail Quadrants Industry Meeting
- 10:30 a.m. Review Retail Gas Quadrant Procedures Document (comments and revisions to the document since 1/17/02 meeting)
- Noon Working lunch
- 12:30 p.m. Review Retail Electric Quadrant Procedures Document (comments and revisions to the document since 1/17/02 meeting)
- 2:00 p.m. Break out Session by Segment Stakeholders - Discussion and Review of Segment Procedures for the Retail Gas Quadrant and for the Retail Electric Quadrant
- 4:00 p.m. Transition Issues and Next steps
- 5:00 p.m.* Adjourn

Hotels within the Vicinity of AGA

Following is a list of hotels within a one-block radius of the AGA offices. Please request the AGA rate when reserving rooms. The rates reflect a negotiated volume rate that will be in effect through December 2002.

Hyatt Regency Capitol Hill \$199 single/double 400 New Jersey Avenue, NW Washington, DC (202) 737-1234 (800) 233-1234

The Hotel George \$184 single \$209 double 15 E Street, NW Washington, DC 20001 (202) 347-4200 (800) 576-8331 e-mail: rooms@hotelgeorge.com

Phoenix Park Hotel\$179 single \$199 double20 North Capitol Street, NWWashington, DC 20001(202) 638-6900(800) 824-5419

Hotel availability in the Capitol Hill area is very tight for the evenings of Monday and Tuesday because most Congressional visits are on Tuesday or Wednesday morning. March, April and May are very tight because many associations hold legislative conferences during that time. September and October are also very tight months due to citywide meetings.

Directions to AGA

For driving directions, please the AGA web site: aga.org/AboutAGA/DirectionstoAGA.

<u>Walking from Union Station Metro Stop to A</u>GA: AGA is located diagonally across from the Union Station Metro Stop—five minutes by foot. Use the Union Station Shops Exit to leave the station. Walk 0.21 miles or 1 block west/right on Massachusetts Ave., NE, until you reach the corner of the Post Office/Postal museum. Turn left on N. Capitol St. NW, walk 2 blocks S. on N. Capitol St., NW.

<u>Directions from National Airport (DCA) via Me</u>tro: The National Airport Metro Stop is on site at the airport. Take the Yellow Line in the direction of Mount Vernon Square. Get off 5 stops later at Gallery Place/China Town Metro Stop. Go up to the upper level. Take the Red Line train in the direction of Glenmont. Get off 2 stops later at Union Station Stop. Approximate travel time: 30 minutes.

<u>Directions from Dulles Airport (IAD) via Washington Flyer Shuttle and Metro:</u> Take the Washington Flyer shuttle to Union Station from the airport to West Falls Church Metro Stop (Frequency: Every half-hour). Take the Orange line train in the direction of New Carrolton. Get off 10 stops later at Metro Center station. Go up to the upper level. Take the Red Line train in the direction of Glenmont. Get off 3 stops later at Union Station Metro Stop. Approximate travel time: 1 hour & 20 minutes.

<u>Directions from Baltimore Washington International Airport (BWI) via Marc</u> train: A) Take the free airport shuttle from BWI to the BWI Airport train station (10-minute ride). Take the Marc train Penn Line toward Washington DC. Get off at Union Station train stop in Washington DC.

B) Or take the SuperShuttle van from BWI to AGA. Van travels door-to-door, with several stops along the way.

NAESB Retail Quadrants Meeting Participants — January 29, 2002

An Asterisk indicates that the individual will participate via conference call.

ACKERMAN, Eric (Eric)

Manager, Regulatory Policy Edison Electric Institute 701 Pennsylvania Avenue, N.W. Suite #300 Washington, DC 20004-2608 Tel: (202) 508-5528 Fax: (202) 508-5445 Email: eackerman@eei.org

ALEXANDER, Barbara R. (Barbara)

Consumer Affairs Consultant NASUCA/AARP 83 Wedgewood Drive Winthrop, ME 04364-9663 Tel: (207) 395-4143 Fax: (207) 395-4143 Email: barbalex@ctel.net

ANDERSON, John (John)

Executive Director Electric Consumers Council 1333 H Street, NW West Tower - 8th Floor Washington, DC 20001-0000 Tel: (202) 682-1390 Fax: (202) 289-6370 Email: janderson@elcon.com

ANTHONY, Jeffrey E. (Jeff)

Manager-Retail Access Implementation Wisconsin Electric - Wisconsin Gas 231 W. Michigan Room 321 Milwaukee, WI 53203-2918 Tel: (414) 221-2481 Fax: (414) 221-3985 Email: jeff.anthony@wepco.com

ARNAOUT, Mariam (Mariam)

American Gas Association 400 North Capitol Street, NW 4th Floor Washington, DC 20001-0000 Tel: (202) 824-7127 Fax: (202) 824-9137 Email: marnaout@aga.org

BAILEY, Janice L. (Janice)*

Manager, Customer Programs Niagara Mohawk Power Corporation James A. O'Neill Building 1304 Buckley Road North Syracuse, NY 13212-Tel: (315) 460-4041 Fax: (315) 460-4007 Email: baileyj@NiagaraMohawk.com

BARKAS, William (Bill)*

Manager, Retail State Gov. Rls. Dominion Retail One Chatham Center Pittsburgh, PA 15219-3441 Tel: (412) 316-7078 Fax: (412) 316-7500 Email: William_L._Barkas@dom.com

BARKLIND, Sharla (Sharla)

Assistant General Counsel NARUC 1101 Vermont Avenue, NW Suite 200 Washington, DC 20005-3521 Tel: (202) 898-1350 Fax: (202) 898-2213 Email: sbarklind@naruc.org

BAUER, Michael (Mike)

Corporate Marketing KeySpan 175 E. Old Country Road Hicksville, NY 11788-0000 Tel: (516) 545-3855 Fax: (516) 545-3252 Email: mbauer@keyspanenergy.com

BOSWELL, William P. (Bill)*

Partner McGuireWoods L.L.P. 23rd Floor, Dominion Tower 625 Liberty Avenue Pittsburgh, PA 15222-3110 Tel: (412) 667-7901 Fax: (412) 667-7952 Email: wboswell@mcguirewoods.com

BOURBONNAIS, William L. (Bill)

Manager, Rates and Economic Evaluations Wisconsin Public Service Corp. P.O. Box 19001 Green Bay, WI 54307-9001 Tel: (920) 433-1573 Fax: (920) 433-5734 Email: wbourbo@wpsr.com

BRAGG, Audrey (Audrey)

Specialist Energy Acquisitions Washington Gas 6801 Industrial Road Springfield, VA 22151-0000 Tel: (703) 750-7991 Fax: (703) 750-7945 Email: abragg@washgas.com

BRAKE, Ernie F (Ernie)* Staff

Atlanta Gas Light Company 1219 Caroline Street Atlanta, GA 30307-0000 Tel: (404) 584-4161 Fax: (404) 584-4187 Email: ebrake@aglresources.com

BUCCIGROSS, Jim (Jim)

Vice President - Energy Industry Group 8760 LLC 3125 Independence Drive Suite 200 Birmingham, AL 35209-4159 Tel: (508) 238-0345 Fax: (205) 802-7669 Email: jhb@8760.com

BURCH, Katheryn (Katheryn)

Project Coordinator Duke Energy Gas Transmission 5400 Westheimer Court Houston, TX 77056-5310 Tel: (713) 627-5765 Fax: (713) 989-3181 Email: klburch@duke-energy.com

BUSTO, Carol (Carol)

Customer Choice Coordinator Dominion Virginia Power P. O. Box 26666 Richmond, VA 23261-6666 Tel: (804) 771-3807 Email: carol_busto@dom.com

CAMP, Yvette (Yvette)*

Research Analyst Southern Company 270 Peachtree Street Bin 949 Atlanta, GA 30308 Tel: (404) 506-0227 Fax: (404) 506-5190 Email: yscamp@southernco.com

CAVE, Robert (Bob)

President American Public Gas Association 11094-D Lee Highway, Suite 102 Fairfax, VA 22030-5014 Tel: (703) 352-3890 Fax: (703) 352-1271 Email: bcave@apga.org

CHEZAR, Dolores D. (Dolores)

Director, Regulatory Policy KeySpan One MetroTech Center Brooklyn, NY 11201-3831 Tel: (718) 403-2987 Fax: (718) 246-2927 Email: dchezar@keyspanenergy.com

COCHRANE, Anna V. (Anna)

Federal Regulatory Affairs Director CMS Energy Corporation 1016 16th Street, N.W. Suite 500 Washington, DC 20036-5703 Tel: (202) 293-5794 Fax: (202) 223-6178 Email: avcochrane@cmsenergy.com

NAESB Retail Quadrants Meeting Participants — January 29, 2002

An Asterisk indicates that the individual will participate via conference call.

DARNELL, David (David)

President & CEO Systrends, Inc. 7855 S. River Parkway Suite 111 Tempe, AZ 85284-Tel: (480) 756-6777 Fax: (480) 756-9755 Email: david.darnell@systrends.com

DAVIS, Dale (Dale)

Consultant Williams Gas Pipeline-Transco 2800 Post Oak Blvd., - Level 19 P.O. Box 1396 Houston, TX 77251-1396 Tel: (713) 215-2396 Fax: (713) 215-3946 Email: Dale.M.Davis@Williams.com

DEBROFF, Scott (Scott)

Vice President-Govt. & Reg. Affrs. Exo Link 4709 Pine Ridge Road Harrisburg, PA 17110-3239 Tel: (717) 230-8122 Fax: (717) 230-8125 Email: scott.debroff@exolink.com

DOYLE, Michele (Michele)

Manager, Transportation Services Consolidated Edison Co. of N.Y. 4 Irving Place New York, NY 10003-Tel: (212) 460-8172 Fax: (212) 529-1452 Email: doylem@coned.com

DRIGGS, Charles L. (Charlie)

Manager, Gas Operations and Planning Conectiv P. O. Box 231 Wilmington, DE 19899-0231 Tel: (302) 429-3194 Fax: (302) 429-3207 Email: charlie.driggs@conectiv.com

EICHENLAUB, David (Dave)

Assistant Director - EAF State Corporation Commission P. O. Box 1197 Richmond, VA 23218-1197 Tel: (804) 371-9050 Fax: (804) 371-9935 Email: deichenlaub@scc.state.va.us

FULTON, Donna (Donna)

Director, Federal Regulatory Policy Northeast Utilities 701 Pennsylvania Avenue, N.W. Suite 300 Washington, DC 20004-2608 Tel: (202) 508-5306 Fax: (202) 508-5309 Email: fultodj@nu.com

GAARDE, Steven (Steven)

Director of Pricing and Rate Administration Consumers Energy Co. 212 W. Michigan Avenue Jackson, MI 49201-2236 Tel: (517) 788-1096 Fax: (517) 788-0756 Email: slgaarde@cmsenergy.com

GALLAGHER, Lynn (Lynn)

Counsel Swidler Berlin Shereff Friedman, LLP 3000 K Street, N.W. Suite 300 Washington, DC 20007-5101 Tel: (202) 424-7556 Fax: (202) 424-7643 Email: lmgallagher@swidlaw.com

GEBHARDT, Rachel H. (Rachel)*

Regulatory Administrator TECO Peoples Gas System 702 North Franklin Street Tampa, FL 33572-9200 Tel: (813) 228-4646 Fax: (813) 228-4742 Email: rhgebhardt@tecoenergy.com

GOODMAN, Craig G. (Craig)

President Nat'l. Energy Marketers Assoc. 3333 K Street, N.W. Suite 425 Washington, DC 20007-3500 Tel: (202) 333-3288 Fax: (202) 333-3266 Email: cgoodman@energy marketers.com

GRAY, Charles D. (Charles)

Exec. Director & General Counsel Nat'l Assoc. of Regulatory Utility Commissioners 1101 Vermont Av., NW Suite 200 Washington, DC 20005-3521 Tel: (202) 898-2208 Fax: (202) 898-2213 Email: cdgray@naruc.org

HACKEMACK, Ronald P. (Ron)

Senior Regulatory Project Manager TXU Electric & Gas 1601 Bryan Street Dallas, TX 75201-Tel: (214) 812-4003 Fax: (214) 812-5014 Email: ron.hackemack@txu.com

HILL, A. Karen (Karen)

Vice President, Regulatory Affairs American Gas Association 400 North Capitol Street, NW 4th Floor Washington, DC 20001-1511 Tel: (703) 841-8400 Fax: (703) 841-8406 Email: khill@aga.org

JONES, Veronica (Veronica)

Energy Manager Defense Energy Support Center 8725 John J. Kingman Rd., Suite 4950 Fort Belvoir, VA 22060-6222 Tel: (703) 767-8552 Fax: (703) 767-8757 Email: vjones@desc.dla.mil

KALISCH, Bertram (Bert)

Vice President, Government Relations American Public Gas Association 11094-D Lee Highway, Suite 102 Fairfax, VA 22030-5014 Tel: (703) 352-3890 Fax: (703) 352-1271 Email: bkalisch@apga.org

KHAN, Misty Ann (Misty Ann)

Vice President New Business Dev. Telerx 723 Dresher Road Horsham, PA 19044-2205 Tel: (713) 463-9661 Fax: (713) 463-9764 Email: khan@telerx.com

KIRK, Jeffrey L. (Jeff)

General Manager, Gas Supply Allegheny Power 414 Summers Street Charleston, WV 25301-1621 Tel: (304) 347-0540 Fax: (304) 367-2068 Email: jkirk2@alleghenypower.com

KISELEWICH, Ruth (Ruth)

Director, Supplier Account Mgmt. Baltimore Gas and Electric Co. 7225 Windsor Road Baltimore, MD 21244-2642 Tel: (410) 265-4003 Fax: (410) 265-4899 Email: ruth.c.kiselewich@bge.com

KOOGLER, David F (David)

Project Director Dominion Virginia Power P. O. Box 26666 Richmond, VA 23261-6666 Tel: (804) 771-3429 Fax: (804) 771-3083 Email: David_Koogler@dom.com

NAESB Retail Quadrants Meeting Participants — January 29, 2002

An Asterisk indicates that the individual will participate via conference call.

KRAFT, Camille (Camille)*

Attorney BTU Watch 2939 Ferndale Houston, TX 77098-1117 Tel: (713) 533-9180 Fax: (713) 533-0179 Email: ckraft@btuwatch.com

LENNON, Rose T. (Rose)

Attorney, Office of General Counsel Washington Gas 1100 H Street, N.W. Washington, DC 20080-0001 Tel: (202) 624-6623 Fax: (202) 624-6789 Email: rlennon@washgas.com

MAASSEL, Mark T. (Mark)

Vice President, Regulatory & Government Affairs NiSource Inc. 801 East 86th Avenue Merrillville, IN 46410-6271 Tel: (219) 647-6400 Fax: (219) 647-6430 Email: mtmaassel@nisource.com

MARKLE, Frank H. (Frank)

General Counsel UGI Corporation P.O. Box 858 Valley Forge, PA 19482-0858 Tel: (610) 337-1000 ext. 3148 Fax: (610) 992-3258 Email: marklef@ugicorp.com

MASON, Jean (Jean)

Customer Enrollment Supervisor Ameren Services 607 East Adams Street Springfield, IL 62739-Tel: (314) 554-3153 Fax: (314) 206-0600 Email: jmason@ameren.com

MCGILL, Christopher (Chris)

Managing Director, Policy Analysis American Gas Association 400 North Capitol Street, NW Washington, DC 20001-1511 Tel: (202) 824-7132 Fax: (202) 824-7087 Email: cmcgill@aga.org

MCGLYNN, Anne (Anne)

Director, Electric Supplier Svcs. ComEd 1919 Swift Drive Oak Brook, IL 60523-0000 Tel: (630) 891-7127 Fax: (630) 684-2830 Email: anne.mcglynn@exeloncorp.com

MCGRATH, Mike (Mike)

Group Director Energy Services Edison Electric Institute 701 Pennsylvania Avenue, N.W. Suite #300 Washington, DC 20004-2608 Tel: (202) 508-5552 Fax: (202) 508-5335 Email: mmcgrath@eei.org

MCQUADE, Rae (Rae)

Executive Director North American Energy Standards Board 1100 Louisiana Suite 3625 Houston, TX 77002-5227 Tel: (713) 356-0060 ext. Ext. Fax: (713) 356-0067 Email: gisb1@aol.com

MENHORN, Cynthia (Cynthia)* Staff

Allegheny Energy, Inc. 800 Cabin Hill Drive Greensburg, PA 15601-1650 Tel: (714) 838-6654 Fax: (724) 830-5174 Email: cmenhor@alleghenyenergy.com

MINNEMAN, Jim (Jim)

Director, Business Support PPL Solutions Two North Ninth Street Allentown, PA 18101-1179 Tel: (610) 774-5774 Fax: (610) 774-7229 Email: jmminneman@pplweb.com

MORAN, Terrence (Terry)

Issues Manager Public Service Electric & Gas Co. 80 Park Plaza T-7 Newark, NJ 07102-4106 Tel: (973) 430-7105 Fax: (973) 624-4107 Email: Terrence.Moran@pseg.com

NOVAK, Michael E. (Mike)

Assistant General Manager National Fuel Gas Distribution Corporation 10 Lafayette Square Room 1700 Buffalo, NY 14203-1824 Tel: (716) 857-7884 Fax: (716) 857-7687 Email: novakm@natfuel.com

OAKS, Timothy J. (Timothy)

UGI Utilities, Inc. 100 Kachel Boulevard Suite 400 Reading, PA 19607-9674 Tel: (610) 796-3521 Fax: (610) 796-3595 Email: toaks@ugi.com

OPPENHEIM, Bill (Bill)

Manager - Gas Aquisition and Pricing PECO Energy Company 300 Front Street West Conshohocken, PA 19428-2723 Tel: (610) 832-6414 Fax: (610) 832-6499 Email: william.oppenheim@peco-energy.com **PAULSON, S. Laurence (Laurie)** Principal Hoffman-Paulson Associates 3314 Rosemary Lane Hyattsville, MD 20782-1032 Tel: (301) 779-6863

PENN, Deborah (Deborah)

Fax: (301) 779-9263

Email: hoffson@aol.com

Vice President, Information Systems American Public Power Association 2301 M Street, N.W. Washington, DC 20037-1427 Tel: (202) 467-2956 Fax: (202) 467-2920 Email: debpenn@appanet.org

RANTALA, Stacey (Stacey)

Nat'l. Energy Marketers Assoc. 3333 K Street Suite 425 Washington, DC 20007-3500 Tel: (202) 333-3288 Fax: (202) 333-3266 Email: srantala@energymarkets.com

REICHERT, Anne (Anne)*

Senior Analyst New York State Electric & Gas Corporation Corporate Drive Kirkwood Industrial Park Binghamton, NY 13905-5224 Tel: (607) 762-7742 Fax: (607) 762-8645 Email: apreichert@nyseg.com

RINGENBACH, Thomas (Tom)

Manager, Business Standards American Electric Power 700 Morrison Road Gahanna, OH 43230-6642 Tel: (614) 883-7542 Fax: (614) 883-7088 Email: tjringenbach@aep.com

ROBERT, Lisa (Lisa)

Energy Manager Defense Energy Support Center 8725 John J. Kingman Road Suite 4950 Fort Belvoir, VA 22060-6222 Tel: (703) 767-8546 Fax: (703) 767-8757 Email: lrobert@desc.dla.mil An Asterisk indicates that the individual will participate via conference call.

ROBERTS, George (George)

Director & VP - Reg. Rls. & Strategy Schlumberger Communications 5430 Metric Place Norcross, GA 30092-2550 Tel: (770) 368-3461 Fax: (770) 368-3490 Email: Roberts7@slb.com

ROSENTHAL, Robert (Bob)

Director, Bureau of Fixed Utly. Svcs. Pennsylvania Public Utilities Commission P.O. Box 3265 Harrisburg, PA 17120-Tel: (717) 783-5242 Fax: (717) 787-4750 Email: rrosental@puc.state.pa.us

ROTHMAN, Jeffrey P (Jeff)

Section Manager, Retail Access Consolidated Edison Co. of N.Y. 4 Irving Place Room 509 New York, NY 10003-3502 Tel: (212) 780-8655 Fax: (212) 228-3436 Email: rothmanj@coned.com

SAPPENFIELD, Keith M II (Keith)

Regional Director, US Regulatory Affrs. PanCanadian Energy Services, Inc.` 1200 Smith Street Houston, TX 77002-4302 Tel: (713) 331-5182 Fax: (713) 331-5008 Email: ksappenfield@pcenergy.com

SCHEEL, Mark (Mark)

Dynegy, Inc. 1000 Louisiana Suite 5800 Houston, TX 77002-5000 Tel: (713) 507-6787 Fax: (713) 507-6834 Email: mark.scheel@dynegy.com

SCHNEIDER, Richard (Rich)

Transaction Manager, Retail Settlement Public Service Electric & Gas Co. P.O. Box 570 80 Park Plaza, L1A Newark, NJ 07102-4106 Tel: (973) 430-6697 Fax: (973) 624-2891 Email: richard.schneider@pseg.com

SHAH, Nicole (Nicole)

Staff Reliant Energy Entex 1111 Louisiana Street 22nd Floor Houston, TX 77002-5230 Tel: (713) 207-8924 Fax: (713) 207-9595 Email: nshah@reliantenergy.com

SHANNING, Robert E. (Robert)*

Director, Gas Supply & Transportation Management Niagara Mohawk Power Corporation James A. O'Neill Building 1304 Buckley Road North Syracuse, NY 13212-4302 Tel: (315) 460-4024 Fax: (315) 460-4007 Email: shanningr@niagaramohawk.com

SIEGEL, Brandon (Brandon)

Senior Consultant Systrends, Inc. RD 1 #71 Greensburg, PA 15601-8734 Tel: (724) 527-3592 Fax: (724) 527-3594 Email: brandon.siegel@systrends.com

SIEMER, Marc S (Marc)

Manager, Trading Systems Cinergy Corp. 7200 Industrial Road Florence, KY 41042-2910 Tel: (859) 372-5796 Fax: (859) 372-5750 Email: msiemer@cinergy.com

SPECHT ADUSEI-POKU, Marilyn (Marilyn)*

Senior Attorney NiSource Energy Distribution Cos. 333 S. Erie Street Toledo, OH 43602-1609 Tel: (419) 252-8152 Fax: (419) 252-8161 Email: mspecht@nisource.com

SULLIVAN, Stephen (Steve)

Regulatory Manager, Gas Supply Consolidated Edison Co. of N.Y. 4 Irving Place Room 2415-S New York, NY 10003-3502 Tel: (212) 460-2733 Fax: (212) 420-8744 Email: sullivans@coned.com

UNRUH, Julie (Julie)

Reliant Resources P.O. Box 4567 Houston, TX 77210-4567 Tel: (713) 207-5973 Fax: (713) 207-9595 Email: junruh@reliant.com

VAN PELT, Kimberly (Kim)

GISB Coordinator CMS Panhandle Eastern Pipe Line 5444 Westheimer Suite 516 Houston, TX 77056-Tel: (713) 989-7354 Fax: (713) 989-1121 Email: kvanpelt@cmsenergy.com

WEAVER, Alonzo (Alonzo)

Vice President, Operations Memphis Light, Gas & Water 220 South Main Street Memphis, TN 38104-0000 Tel: (901) 528-4481 Fax: (901) 528-4086 Email: aweaver@mlgw.org

WEBSTER, Marc P. (Marc)*

Specialist-Gas Pricing New York State Electric & Gas Corporation Corporate Drive Kirkwood Industrial Park Binghamton, NY 13902-5224 Tel: (607) 762-8075 Fax: (607) 762-8405 Email: mpwebster@nyseg.com

WILEN, Eric (Eric)*

Senior Analyst New York State Electric & Gas Corporation Corporate Drive Kirkwood Industrial Park Binghamton, NY 13905-5224 Tel: (607) 762-7430 Fax: (607) 762-8645 Email: ejwilen@nyseg.com

WILLIAMS, Shyletha (Shyletha)

Traffic Management Specialist Defense Energy Support Center 8725 John J. Kingman Road Suite 4950 Fort Belvoir, VA 22060-6217 Tel: (703) 767-8559 Fax: (703) 767-8757 Email: swilliams@desc.dla.mil

WOJICK, David (Dave)

Correspondent Electricity Daily P.O. Box 30203 Alexandria, VA 22310-8203 Tel: (540) 858-3503 Email: dwojick@shentel.net An Asterisk indicates that the individual will participate via conference call.

YACKER, Marc (Marc)

Director of Govt. and Public Affrs. Electric Consumers Council 1333 H Street, NW West Tower - 8th Floor Washington, DC 20001-0000 Tel: (202) 682-1390 Fax: (202) 289-6370 Email: myacker@elcon.org

YOUNG, Randall (Randy)

Director, Regulatory Compliance Gulf South Pipeline Company, LP 20 Greenway Plaza Houston, TX 77046-2011 Tel: (713) 544-4715 Fax: (713) 544-4184 Email: randy.young@gulfsouthpl.com

ZAVODNICK, Steve M. (Steve)

Senior Engineer Baltimore Gas and Electric Co. 1699 Leadenhall Street Baltimore, MD 21230-4854 Tel: (410) 291-4908 Fax: (410) 291-5104 Email: steven.m.zavodnick@bge.com

AMERICAN GAS ASSOCIATION ANTITRUST COMPLIANCE GUIDELINES

Introduction

The American Gas Association and its member companies are committed to full compliance with all laws and regulations, and to maintaining the highest ethical standards in the way we conduct our operations and activities. Our commitment includes strict compliance with federal and state antitrust laws, which are designed to protect this country's free competitive economy.

Responsibility for Antitrust Compliance

Compliance with the antitrust laws is a serious business. Antitrust violations may result in heavy fines for corporations, and in fines and even imprisonment for individuals. While the General Counsel's Office provides guidance on antitrust matters, you bear the ultimate responsibility for assuring that your actions and the actions of any of those under your direction comply with the antitrust laws.

Antitrust Guidelines

In all AGA operations and activities, you must avoid any discussions or conduct that might violate the antitrust laws or even raise an appearance of impropriety. The following guidelines will help you do that:

- **Do** consult counsel about any documents that touch on sensitive antitrust subjects such as pricing, market allocations, refusals to deal with any company, and the like.
- **Do** consult with counsel on any non-routine correspondence that requests an AGA member company to participate in projects or programs, submit data for such activities, or otherwise join other member companies in AGA actions.
- **Do** use an agenda and take accurate minutes at every meeting. Have counsel review the agenda and minutes before they are put into final form and circulated.

- **Do not** have discussions with other member companies about:
 - your company's prices for products or services, or prices charged by your competitors
 - costs, discounts, terms of sale, profit margins or anything else that might affect those prices
 - the resale prices your customers should charge for products you sell them
 - allocating markets, customers, territories or products with your competitors
 - limiting production
 - whether or not to deal with any other company
 - any competitively sensitive information concerning your own company or a competitor's.
- **Do not** stay at a meeting, or any other gathering, if those kinds of discussions are taking place.
- Do not discuss any other sensitive antitrust subjects (such as price discrimination, reciprocal dealing, or exclusive dealing agreements) without first consulting counsel.
- **Do not** create any documents or other records that might be misinterpreted to suggest that AGA condones or is involved in anticompetitive behavior.

We're Here to Help

Whenever you have any question about whether particular AGA activities might raise antitrust problems, contact the General Counsel's Office.

> American Gas Association Office of General Counsel and Corporate Secretary December 1997

TAB 2

REVIEW OF 1/17/02 MEETING

NAESB Retail Gas Quadrants Open Industry Meeting Washington DC – AGA Offices January 17, 2002 9:30 AM – 5:00 PM EDT*

Proposed Agenda

A conference bridge will be provided to meeting registrants who are unable to attend in person.

<u>Meeting Purpos</u>e: 1) To introduce interested participants to the bylaws of NAESB and the Retail Gas Quadrant Procedures proposed by AGA and APGA and the Retail Electric Quadrant proposed UBP-RSB; and 2) to develop consensus on Quadrant procedures. Jim Buccigross, Vice President - Energy Industry, Group 8760 LLC, will moderate the meeting.

- 9:30 a.m.Registration and Continental Breakfast
- 10:00 a.m. Welcome & Administrative Items (Jim Buccigross)
 - Antitrust Guidelines
 - Housekeeping Items and Participant/Caller Guidelines
 - Introductions
- 10:15 a.m. NAESB Update (Bill Boswell, McGuire Woods LLP)
 - Introduction to the NAESB Bylaws
 - Implications of the GISB Board Resolution of 12/5/01 in Terms of Milestones and Timelines
 - Q&A/Discussion
- Noon Working Lunch
- 12:30 p.m. Review of Retail Gas Quadrant Procedures and Retail Electric Quadrant Procedures (Jim Buccigross) Q&A with the Panel (Bill Boswell, for Dominion; Charlie Driggs; Conectiv, Rose Lennon, Washington Gas; Mike Novak, National Fuel Gas Distribution; and Keith Sappenfield, PanCanadian Energy Services Inc.)
- 3:00 p.m. Initial Comments on and Revisions to the Retail Gas Quadrant and Retail Electric Quadrant Procedures

4:45 p.m. Preview of January 29 Meeting and Next Steps in Preparation

- Input and Comments on the Retail Gas Quadrant Procedures Document
- Input and Comments on the Retail Electric Quadrants Procedures
- Segment Procedures to be Drafted by the Individual Segments

5:00 p.m.* Adjourn

* The meeting will be extended to 6:00 PM if the discussion warrants.

NAESB Retail Quadrants Meeting Participants — January 17, 2002

*An asterisk indicates that the individual will participate via conference call.

ACKERMAN, Eric (Eric)

Manager, Regulatory Policy Edison Electric Institute 701 Pennsylvania Avenue, N.W. Suite #300 Washington, DC 20004-2608 Tel: (202) 508-5528 Fax: (202) 508-5445 Email: eackerman@eei.org

ALEXANDER, Barbara R. (Barbara)

Consumer Affairs Consultant NASUCA/AARP 83 Wedgewood Drive Winthrop, ME 04364-9663 Tel: (207) 395-4143 Fax: (207) 395-4143 Email: barbalex@ctel.net

ANDERSON, John (John)

Executive Director Electric Consumers Council 1333 H Street, NW West Tower - 8th Floor Washington, DC 20001-0000 Tel: (202) 682-1390 Fax: (202) 289-6370 Email: janderson@elcon.com

ANTHONY, Jeffrey E. (Jeff)

Manager-Retail Access Implementation Wisconsin Electric - Wisconsin Gas 231 W. Michigan Room 321 Milwaukee, WI 53203-2918 Tel: (414) 221-2481 Fax: (414) 221-3985 Email: jeff.anthony@wepco.com

ARNAOUT, Mariam (Mariam)

Manager, Gas Transportation Programs American Gas Association 400 North Capitol Street, NW 4th Floor Washington, DC 20001-0000 Tel: (202) 824-7127 Fax: (202) 824-9137 Email: marnaout@aga.org

ARY, Ursula (Ursula)

Manager, Information & Customer Programs American Public Power Association 2301 M Street, N.W. Washington, DC 20037-1427 Tel: (202) 467-2980 Fax: (202) 467-2910 Email: uary@appanet.org

BAILEY, Janice L. (Janice)*

Manager, Customer Programs Niagara Mohawk Power Corporation James A. O'Neill Building 1304 Buckley Road North Syracuse, NY 13212-Tel: (315) 460-4041 Fax: (315) 460-4007 Email: baileyj@NiagaraMohawk.com

BARKAS, William (Bill)*

Manager, Retail State Gov. Rls. Dominion Retail One Chatham Center Pittsburgh, PA 15219-3441 Tel: (412) 316-7078 Fax: (412) 316-7500 Email: William_L._Barkas@dom.com

BAUER, Michael (Mike)

Corporate Marketing KeySpan 175 E. Old Country Road Hicksville, NY 11788-0000 Tel: (516) 545-3855 Email: mbauer@keyspanenergy.com

BOSWELL, William P. (Bill)

Partner McGuireWoods L.L.P. 23rd Floor, Dominion Tower 625 Liberty Avenue Pittsburgh, PA 15222-3110 Tel: (412) 667-7901 Fax: (412) 667-7952 Email: wboswell@mcguirewoods.com

BOURBONNAIS, William L. (Bill)

Manager, Rates and Economic Evaluations Wisconsin Public Service Corporation P.O. Box 19001 Green Bay, WI 54307-9001 Tel: (920) 433-1573 Fax: (920) 433-5734 Email: wbourbo@wpsr.com

BRAGG, Audrey (Audrey)

Specialist Energy Acquisitions Washington Gas 6801 Industrial Road Springfield, VA 22151-0000 Tel: (703) 750-7991 Fax: (703) 750-7945 Email: abragg@washgas.com

BRAKE, Ernie F (Ernie)*

Staff Atlanta Gas Light Company 1219 Caroline Street Atlanta, GA 30307-0000 Tel: (404) 584-4161 Fax: (404) 584-4187 Email: ebrake@aglresources.com

BUCCIGROSS, Jim (Jim)

Vice President - Energy Industry Group 8760 LLC 3125 Independence Drive Suite 200 Birmingham, AL 35209-4159 Tel: (508) 238-0345 Fax: (205) 802-7669 Email: jhb@8760.com

BURNHAM, Andrew J. (Andy)*

Rate Analyst Consumers Energy Co. 212 West Michigan Avenue Room H441 Jackson, MI 49201-2236 Tel: (517) 788-0951 Fax: (517) 788-0756 Email: ajburnham@cmsenergy.com

BUSTO, Carol (Carol)*

Customer Choice Coordinator Dominion Virginia Power P. O. Box 26666 Richmond, VA 23261-6666 Tel: (804) 771-3807 Email: carol_busto@dom.com

CAMP, Yvette (Yvette)

Research Analyst Southern Company 270 Peachtree Street Bin 949 Atlanta, GA 30308 Tel: (404) 506-0227 Fax: (404) 506-5190 Email: yscamp@southernco.com

CHEZAR, Dolores D. (Dolores)

Director, Regulatory Policy KeySpan One MetroTech Center Brooklyn, NY 11201-3831 Tel: (718) 403-2987 Fax: (718) 246-2927 Email: dchezar@keyspanenergy.com

CIANFLONE, Nancy C. (Nancy)

Director, Marketing Services KeySpan One MetroTech Center Brooklyn, NY 11201-3831 Tel: (718) 403-2505 Fax: (718) 596-7802 Email: ncianflone@keyspanenergy.com

COCHRANE, Anna V. (Anna)

Federal Regulatory Affairs Director CMS Energy Corporation 1016 16th Street, N.W. Suite 500 Washington, DC 20036-5703 Tel: (202) 293-5794 Fax: (202) 223-6178 Email: avcochrane@cmsenergy.com *An asterisk indicates that the individual will participate via conference call.

COOPER, Bill (Bill)

Counsel House Commerce Committee House Commerce Committee 2125 Rayburn HOB Washington, DC 20510-Tel: (202) 225-2927 Fax: (202) 225-2899 Email: bill.cooper@mail.house.gov

DAVIS, Dale (Dale)

Consultant Williams Gas Pipeline-Transco 2800 Post Oak Blvd., - Level 19 P.O. Box 1396 Houston, TX 77251-1396 Tel: (713) 215-2396 Fax: (713) 215-3946 Email: Dale.M.Davis@Williams.com

DEBROFF, Scott (Scott)

Vice President-Govt. & Reg. Affrs. Exo Link 4709 Pine Ridge Road Harrisburg, PA 17110-3239 Tel: (717) 230-8122 Fax: (717) 230-8125 Email: scott.debroff@exolink.com

DOYLE, Michele (Michele)

Manager, Transportation Services Consolidated Edison Company of New York, Inc. 4 Irving Place New York, NY 10003-Tel: (212) 460-8172 Fax: (212) 529-1452 Email: doylem@coned.com

DRIGGS, Charles L. (Charlie)

Manager, Gas Operations and Planning Conectiv P. O. Box 231 Wilmington, DE 19899-0231 Tel: (302) 429-3194 Fax: (302) 429-3207 Email: charlie.driggs@conectiv.com

EICHENLAUB, Dave (Dave)

Assistant Director - EAF State Corporation Commission P. O. Box 1197 Richmond, VA 23218-1197 Tel: (804) 371-9050 Fax: (804) 371-9935 Email: deichenlaub@scc.state.va.us

GALLAGHER, Lynn (Lynn)

Counsel Swidler Berlin Shereff Friedman, LLP 3000 K Street, N.W. Suite 300 Washington, DC 20007-5101 Tel: (202) 424-7556 Fax: (202) 424-7643 Email: lmgallagher@swidlaw.com

GEBHARDT, Rachel H. (Rachel)

Regulatory Administrator TECO Peoples Gas System 702 North Franklin Street Tampa, FL 33602-4429 Tel: (813) 228-4646 Fax: (813) 228-4742 Email: rhgebhardt@tecoenergy.com

GOODMAN, Craig G. (Craig)

President National Energy Marketers Association 3333 K Street, N.W. Suite 425 Washington, DC 20007-3500 Tel: (202) 333-3288 Fax: (202) 333-3266 Email: cgoodman@energy marketers.com

HACKEMACK, Ronald P. (Ron)

Senior Regulatory Project Manager TXU Electric & Gas 1601 Bryan Street Dallas, TX 75201-Tel: (214) 812-4003 Fax: (214) 812-5040 Email: ron.hackemack@txu.com

HILL, A. Karen (Karen)

Vice President, Regulatory Affairs American Gas Association 400 North Capitol Street, NW 4th Floor Washington, DC 20001-1511 Tel: (703) 841-8400 Fax: (703) 841-8406 Email: khill@aga.org

KALISCH, Bertram (Bert)

Vice President, Government Relations American Public Gas Association 11094-D Lee Highway, Suite 102 Fairfax, VA 22030-5014 Tel: (703) 352-3890 Fax: (703) 352-1271 Email: bkalisch@apga.org

KHAN, Misty Ann (Misty Ann)*

Vice President New Business Dev. Telerx 723 Dresher Road Horsham, PA 19044-2205 Tel: (713) 463-9661 Fax: (713) 463-9764 Email: khan@telerx.com

KIRK, Jeffrey L. (Jeff)

General Manager, Gas Supply Allegheny Power 414 Summers Street Charleston, WV 25301-1621 Tel: (304) 347-0540 Fax: (304) 367-2068 Email: jkirk2@alleghenypower.com

KISELEWICH, Ruth (Ruth)

Director, Supplier Account Mgmt. Baltimore Gas and Electric Co. 7225 Windsor Road Baltimore, MD 21244-2642 Tel: (410) 265-4003 Fax: (410) 265-4899 Email: ruth.c.kiselewich@bge.com

KRAFT, Camille (Camille)*

Attorney BTU Watch 2939 Ferndale Houston, TX 77098-1117 Tel: (713) 533-9180 Fax: (713) 533-0179 Email: ckraft@btuwatch.com

MAASSEL, Mark T. (Mark)

Vice President, Regulatory & Government Affairs NiSource Inc. 801 East 86th Avenue Merrillville, IN 46410-6271 Tel: (219) 647-6400 Fax: (219) 647-6430 Email: mtmaassel@nisource.com

MARKLE, Frank H. (Frank)

General Counsel UGI Corporation P.O. Box 858 Valley Forge, PA 19482-0858 Tel: (610) 337-1000 ext. 3148 Fax: (610) 992-3258 Email: marklef@ugicorp.com

MCGILL, Christopher (Chris)

Managing Director, Policy Analysis American Gas Association 400 North Capitol Street, NW Washington, DC 20001-1511 Tel: (202) 824-7132 Fax: (202) 824-7087 Email: cmcgill@aga.org

MCGLYNN, Anne (Anne)

Director, Electric Supplier Svcs. ComEd 1919 Swift Drive Oak Brook, IL 60523-0000 Tel: (630) 891-7127 Fax: (630) 684-2830 Email: anne.mcglynn@exeloncorp.com

NAESB Retail Quadrants Meeting Participants — January 17, 2002

*An asterisk indicates that the individual will participate via conference call.

MCQUADE, Rae (Rae)

Executive Director North American Energy Standards Board 1100 Louisiana Suite 3625 Houston, TX 77002-5227 Tel: (713) 356-0060 ext. Ext. Fax: (713) 356-0067 Email: gisb1@aol.com

MENHORN, Cynthia (Cynthia)*

Staff Allegheny Energy, Inc. 800 Cabin Hill Drive Greensburg, PA 15601-1650 Tel: (728) 386-654 Fax: (724) 830-5174 Email: cmenhor@alleghenyenergy.com

MINNEMAN, Jim (Jim)

Director, Business Support PPL Solutions Two North Ninth Street Allentown, PA 18101-1179 Tel: (610) 774-5774 Fax: (610) 774-7229 Email: jmminneman@pplweb.com

MORAN, Terrence (Terry)

Issues Manager Public Service Electric & Gas Company 80 Park Plaza T-7 Newark, NJ 07102-4106 Tel: (973) 430-7105 Fax: (973) 624-4107 Email: Terrence.Moran@pseg.com

NOVAK, Michael E. (Mike)

Assistant General Manager National Fuel Gas Distribution Corporation 10 Lafayette Square Room 1700 Buffalo, NY 14203-1824 Tel: (716) 857-7884 Fax: (716) 857-7687 Email: novakm@natfuel.com

OPPENHEIM, Bill (Bill)

Manager - Gas Aquisition and Pricing PECO Energy Company 300 Front Street West Conshohocken, PA 19428-2723 Tel: (610) 832-6414 Fax: (610) 832-6499 Email: william.oppenheim@peco-energy.com

PAULSON, S. Laurence (Laurie)

Principal Hoffman-Paulson Associates 3314 Rosemary Lane Hyattsville, MD 20782-1032 Tel: (301) 779-6863 Fax: (301) 779-9263 Email: hoffson@aol.com

RANTALA, Stacey (Stacey)

National Energy Marketers Association 3333 K Street Suite 425 Washington, DC 20007-3500 Tel: (202) 333-3288 Fax: (202) 333-3266 Email: srantala@energymarkets.com

RINGENBACH, Thomas (Tom)*

Manager, Business Standards American Electric Power 700 Morrison Road Gahanna, OH 43230-6642 Tel: (614) 883-7542 Fax: (614) 883-7088 Email: tjringenbach@aep.com

ROBERT, Lisa (Lisa)

Energy Manager Defense Energy Support Center 8725 John J. Kingman Road Suite 4950 Fort Belvoir, VA 22060-6222 Tel: (703) 767-8546 Fax: (703) 767-8757 Email: lrobert@desc.dla.mil

ROBERTS, George (George)

Director & VP - Reg. Rls. & Strategy Schlumberger Communication Systems 5430 Metric Place Norcross, GA 30092-2550 Tel: (770) 368-3461 Fax: (770) 368-3490 Email: Roberts7@slb.com **ROSENTHAL, Robert (Robert)** Director, Bureau of Fixed Utly. Svcs. Pennsylvania Public Utilities Commission P.O. Box 3265 Harrisburg, PA 17120-Tel: (717) 783-5242 Fax: (717) 787-4750 Email: rrosental@puc.state.pa.us

ROTHMAN, Jeffrey P (Jeff)

Section Manager, Retail Access Consolidated Edison Company of New York, Inc. 4 Irving Place Room 509 New York, NY 10003-3502 Tel: (212) 780-8655 Fax: (212) 228-3436 Email: rothmanj@coned.com

SAPPENFIELD, Keith M II (Keith)

Regional Director, US Regulatory Affrs. PanCanadian Energy Services, Inc.` 1200 Smith Street Houston, TX 77002-4302 Tel: (713) 331-5182 Fax: (713) 331-5008 Email: ksappenfield@pcenergy.com

SCHNEIDER, Richard (Rich)

Transaction Manager, Retail Settlement Public Service Electric & Gas Company P.O. Box 570 80 Park Plaza, L1A Newark, NJ 07102-4106 Tel: (973) 430-6697 Fax: (973) 624-2891 Email: richard.schneider@pseg.com

SHAH, Nicole (Nicole)

Staff Reliant Energy Entex 1111 Louisiana Street 22nd Floor Houston, TX 77002-5230 Tel: (713) 207-8924 Fax: (713) 207-9595 Email: nshah@reliantenergy.com

SHANNING, Robert E. (Bob)*

Director, Gas Supply & Transportation Management Niagara Mohawk Power Corporation James A. O'Neill Building 1304 Buckley Road North Syracuse, NY 13212-4302 Tel: (315) 460-4024 Fax: (315) 460-4007 Email: shanningr@niagaramohawk.com

SIEMER, Marc S (Marc)

Manager, Trading Systems Cinergy Corp. 7200 Industrial Road Florence, KY 41042-2910 Tel: (859) 372-5796 Fax: (859) 372-5750 Email: msiemer@cinergy.com

SPECHT ADUSEI-POKU, Marilyn (Marilyn)*

Senior Attorney NiSource Energy Distribution Cos. 333 S. Erie Street Toledo, OH 43602-1609 Tel: (419) 252-8152 Fax: (419) 252-8161 Email: mspecht@nisource.com

SULLIVAN, Stephen (Steve)*

Regulatory Manager, Gas Supply Consolidated Edison Company of New York, Inc. 4 Irving Place Room 2415-S New York, NY 10003-3502 Tel: (212) 460-2733 Fax: (212) 420-8744 Email: sullivans@coned.com

NAESB Retail Quadrants Meeting Participants — January 17, 2002

*An asterisk indicates that the individual will participate via conference call.

TIERNAN, Tom

Managing Editor Platts 1200 G Street Washington DC Tel: (202) 383-2260 Fax: (202) 383-2125 Email: tom_tiernan@platts.com

TILLSON, Laverne (Laverne)

Public Utilities Specialist Defense Energy Support Center 8725 John J. Kingman Road Suite 4950 Fort Belvoir, VA 22060-6217 Tel: (703) 767-9402 Fax: (703) 767-8751 Email: Itillson@desc.dla.mil

UNRUH, Julie (Julie)

Staff Reliant Energy P.O. Box 4567 Houston, TX 77210-4567 Tel: (713) 207-5973 Fax: (713) 207-9595 Email: junruh@reliant.com

VAN PELT, Kimberly (Kim)

GISB Coordinator CMS Panhandle Eastern Pipe Line 5444 Westheimer Suuite 77056 Houston, TX 77256-0000 Tel: (713) 989-7354 Fax: (713) 989-1121 Email: kvanpelt@cmsenergy.com

WEAVER, Alonzo (Alonzo)*

Vice President, Operations Memphis Light, Gas & Water 220 South Main Street Memphis, TN 38104-0000 Tel: (901) 528-4481 Fax: (901) 528-4086 Email: aweaver@mlgw.org

WOJICK, David (Dave)*

Correspondent Electricity Daily P.O. Box 30203 Alexandria, VA 22310-8203 Tel: (540) 858-3503 Email: dwojick@shentel.net

YACKER, Marc (Marc)

Director of Govt. and Public Affrs. Electric Consumers Council 1333 H Street, NW West Tower - 8th Floor Washington, DC 20001-0000 Tel: (202) 682-1390 Fax: (202) 289-6370 Email: myacker@elcon.org

YOUNG, Randall (Randy)

Director, Regulatory Compliance Gulf South Pipeline Company, LP 20 Greenway Plaza Houston, TX 77046-2011 Tel: (713) 544-4715 Fax: (713) 544-4184 Email: randy.young@gulfsouthpl.com

ZAVODNICK, Steve M. (Steve)

Senior Engineer Baltimore Gas and Electric Co. 1699 Leadenhall Street Baltimore, MD 21230-4854 Tel: (410) 291-4908 Fax: (410) 291-5104 Email: steven.m.zavodnick@bge.com

NAESB Retail Quadrants Meeting Minutes Washington DC - January 17, 2002 10:00 AM – 5:00 PM EDT

Interested participant in the Retail Gas and Retail Electric Quadrants of the newly formed North American Energy Standards Board (NAESB) met today to review the proposals for operating procedures of each of the two Quadrants. The meeting was moderated by Jim Buccigross (Group 8760), and a four-member panel of the following individuals was available to answer questions: Bill Boswell (Dominion), Charlie Driggs (Conectiv), Mike Novak (National Fuel), and Keith Sappenfield (PanCanadian Energy). Rae McQuade (NAESB Executive Director) was also available to answer questions. Upon reviewing the antitrust guidelines, the meeting participants completed the following business:

- **LISTENED TO A PRESENTATION BY MR. BOSWELL** (NAESB Board Chair) on the NAESB organization, its structure, principles, and its governance provisions. Mr. Boswell also discussed the timelines and milestones involved in setting up NAESB as a fully-populated and functioning organization and mentioned the upcoming meetings on the wholesale electric quadrant to be held at FERC and DOE on January 24 and 29, respectively.
- **LISTENED TO A PRESENTATION BY MR. BUCCIGROSS** (Executive Committee Chair, NAESB Wholesale Gas Quadrant), which explained the difference between the role of the NAESB Board and that of the NAESB Quadrants' Executive Committees (EC). Mr. Buccigross also briefly explained the difference between the various governance and procedural documents related to NAESB, the Quadrants, and the Segments. Finally, Mr. Buccigross made the following points:
 - The meeting's objective today is to review the Retail Gas Quadrant Procedures proposal and the Retail Electric Quadrant Procedure proposal in order to 1) ensure consistency with the NAESB Bylaws; 2) strive for consistency and harmony between the procedures of the two Quadrants, but NOT make them identical; and 3) begin to identify—but NOT debate—the issues of import to the various stakeholders.
 - Any comments and recommendations suggested by meeting participants today are NOT binding on any party in the discussion until the UBP-RSB and GRGS Task Force each reach consensus on the issues within their group.
- **REVIEWED AND MADE RECOMMENDATIONS ON THE TWO QUADRANTS' PROPOSALS** by comparing the two proposals, beginning with the differences, using as a starting point a work paper that compared the two proposals. The stakeholders were able to identify about ten or more differences/or issues at the meeting today. (A summary on the differences identified, the recommendations and suggestions made, and the issues that were mentioned will be distributed to meeting participants, the GRGS membership and the UBP-RSB membership).
- **STARTED A DIALOGUE AMONG THE VARIOUS STAKEHOLDERS** on issues of pertinence to each, which familiarized other participants with the different perspectives and concerns.
- DISCUSSED THE PURPOSE OF THE JANUARY 29 RETAIL QUADRANTS MEETING AND AGREED ON NEXT STEPS as follows:
 - Agreed to create a joint Retail Electric & Retail Gas Quadrants task force. A number of people volunteered at the meeting. Charlie Driggs (UBP-RSB Governance Committee Chair) and Mariam Arnaout (for the AGA-GRGS Task Force) agreed to recruit more volunteers to achieve balanced representation within the task force. Participants were asked to notify Mr. Driggs or Ms. Arnaout of their interest.
 - The task of this joint subcommittee will be to revise the proposals of both Quadrants to 1) make them consistent where applicable; 2) simplify them where possible (i.e. remove language that is already addressed in the NAESB Bylaws); and 3) ensure that the language conforms with the provisions in the NAESB Bylaws. This "REQ-RGQ task force" would then present their recommendations (i.e. the revised drafts) for consideration by all stakeholders at the January 29 meeting.
 - To conduct its business, this task force will convene via conference call and e-mail to conduct its business.
 - The first conference call will be next <u>Tuesday, January 22 at 10 AM EDT</u>.
- ADJOURNED AT 4:45 PM.

Summary of Comments Provided During the January 17, 2002 Retail Quadrants Open Industry Meeting

The following summarizes the comments provided by meeting participants during the January 17, 2002 meeting on the Retail Electric Quadrant Procedures and the Retail Gas Quadrant Procedures. Each recommendation is followed a note that characterizes the sense of the room on each suggestion, as noted by the meeting facilitator, Jim Buccigross, during the meeting.

Comments and Suggestions:

- Delete language that is already contained in the NAESB Bylaws. The Quadrants are supposed to conform with the NAESB Bylaws, so
 it is not necessary to repeat the provisions. What's important is to ensure the procedures determined by the Quadrants are consistent
 with the NAESB principles and rules.
 GENERAL AGREEMENT.
- Suggestion to use one or the other of the terms "Model business practices" and "standards"—not both. ▷ <u>SUGGESTION NOT</u> <u>ACCEPTED. It was deemed necessary to use both.</u>
- Clarification: The explanatory notes and italicized language will be removed from the final version of the REQ Procedures.
- Make the definitions sections within the two Quadrants' documents convergent: i.e. same terms defined; same definitions. r
- Several of the definitions in the two documents are already defined in the NAESB Bylaws. Therefore keep the terms in the document but replace the actual definitions with a reference to the NAESB Bylaws.
- Include a separate section to the Retail Electric Quadrant (REQ) Procedures that defines each Segment, because Segment definitions are determined by each Quadrant and tie into eligibility for membership in a given Segment.
- The definitions of Segments ought to be identical in the two Quadrants' procedures. ▷ SUGGESTION NOT ACCEPTED.
- Definition of Quorum:
- AGREED to remove the definition of Quorum in the REQ Procedures as it is currently written because the definition as it relates to the Executive Committee (EC) contradicts the NAESB Bylaws, which require a minimum attendance threshold for a supermajority vote to be taken. majority of the EC Member and a minimum-per-Segment threshold in attendance for a supermajority vote be taken (which is the case in the business of passing standards--the minimum being a majority of the EC Members and a minimum per-Segment threshold). Clarification: This requirement does not apply to votes on procedural matters, which require a simple majority vote of those in attendance. The GISB precedent on the matter was given as an illustration: 13 out of the 25 EC Members *AND* 2 per Segment (out of the 5) were required to be in attendance for business to be conducted (i.e. passing standards). Clarification: If the sufficient number of notional votes are submitted prior to a meeting, then the vote carries. However, often during meetings EC Members make motions to modify the recommendation for a standard/business practice. In such a case, notational ballots are distributed to the absent EC Members following the meeting.
- AGREED that it may be necessary to have the same definition of Quorum for both Quadrants for the purpose of conducting joint-Quadrant meetings.
- There are many references to members in both documents and in the NAESB Bylaws. The NAESB Bylaws distinguish between Voting and Non-voting Members. Need to distinguish between Voting Members and Non-Members within Segments as well (not sure if tat was exactly the point.)

Note: The following table was extracted from the meeting work paper comparing the two proposals. Comments on the 13 comparison points are in red font.

SEEMINGLY DIFFERERENT APPROACH OR PERSPECTIVE

REQ Proposal	RGQ Proposal
1. Definition of Agent: Definition is specific to the Segment (Section 2.0, line 91)	 Definition of Agent is excluded because it's already defined in the NAESB Bylaws and as such applies to all Members of all Quadrants and Segments (Section 1.0, starts on line 30).
 Consumer advocacy groups are required to show proof of their eligibility status upon application, to be approved by the REQ (through a simple majority of the Membership). (Section 3.2.b ii, lines 191-206) 	 The RGQ proposal does not include this stipulation for consumer advocacy groups.
3. The approval criteria for trade associations differ in the REQ and the RGQ proposals. In the REQ proposal, the approval criteria include: a) specific documentation to establish the legal status of the group (specific legal documents or a dated letter of authorization from the Segment); b) defines the threshold of "underrepresented" as two Members in the Segment; and c) a <i>simple majority</i> affirmative vote of the Segment membership is necessary; and d) immediate disqualification upon exceeding the association's constituents exceeding the two-Members-per-Segment threshold unless requested by its constituency in writing. (Section 3.2.b, lines 183-206).	 3. The RGQ approval criteria for trade associations: a) A special application with a recommendation from its Segment; b) <i>"affirmative vote of% of the Quadrant EC</i>;" c) an annual review with an EC notational vote on "under-representation" status of the association's constituents. (Section 2.0, lines 55-60). C Two viewpoints: 1) Must have criteria and approval process; just not to decide on a consistent approach. Agreement: 2) Exception is ELCON: Eliminate criteria or eligibility requirements.
 Consumer advocacy groups are required to show proof of their eligibility status upon application, to be approved by the REQ (through a simple majority of the Membership). (Section 3.2.b ii, lines 191-206) 	4. No such requirements in RGQ proposal Same issue as #3.
5. While both Quadrant in general require individuals to declare Segment affiliation, the REQ approach more clear than . REQ says that the duly authorized representative of the Member in multiple Segments must be designated to only one Segment; which implies that the individuals cannot represent one Segment during one meeting and another during a different meeting. Section 8.0, Line 221-224	 5. RGQ: RGQ says must declare Segment affiliation prior to a meeting. A representative cannot represent more than one Segment in any given meeting. A company can join multiple Segments: however, an individual cannot represent more than one Segment within the same
Some companies are structured as separate entities operating within the different Segments; However, some companies do operate within the different Segments within the same quadrant of the energy industry.	Quadrant.
 Designated Alternates: REQ: Unsuccessful candidates for EC seats with the greatest number of votes automatically become Designated Alternates. 	 RGQ: All Designated Alternates are elected separately; no provision for EC candidates to automatically be selected as Alternates.
KeySpan & National Fuel believe you need to elect Alternates because they will have to commit to the duties required.	➡ This is a case where it could be left to the Quadrant.
 Election of Designated Alternates: REQ: Each will elect no more than 3/no less than 1 per EC Member with a maximum number of 12. 	 7. RGQ: Alternates discussed in Segment Procedures. 8 is the number. This is a Segment procedure to be dealt with there.
8. Number of Alternates: REQ: no more than 3 etc.	8. Discussed in Distributor Segment Procedures
 REQ: Election rules for Directors determined at the Quadrant level. No details. 	 Same as above: at Segment level. RGQ: Election rules very detailed. Same provision for EC. Section 4.0, Line 110
	Note the terms of EC and Board can be different in each Quadrant and may be staggered differently.

10. REQ: Directors and EC Members serve a two-year term ↔ No problem with having terms of two. This does not have to be identical for both Quadrants. NAESB Bylaws state that the term can be 1 to 3 years.	10. RGQ: three years. Same provision for EC. Section 3.1, Line 77-79 & Section 4.1.a, Line 115-117
11. REQ: Directors and EC Members serve a two-year term This is not an issue.	 RGQ: three years. Same provision for EC. Section 3.2, Line 87-89 & Section 5.1, Line 167-169
12. Terms of Initial Directors: REQ: If Segment Membership is at 5 but hasn't reached 8; then the terms of the 2 initial Directors from each Segment will be two-years. If the Segment Membership reaches the 8-member threshold, then the terms 2 of the 4 elected Directors from the Segment will serve 3 years (greater number of votes) and the other 2 will serve a term of two years. Same provision for EC	12. RGQ: No provisions on this.
The language needs to be clarified to show that ALL segments will have an equal number of representatives; therefore, all the Segments must attain the 8 Member-threshold to elect 4 per Segment within 60 days. This does not violate NAESB Bylaws to have an interim period where you have only 2 representatives per Segment and 2 vacant seats.	
➡ KeySpan: Does anyone object to having a different number of representatives per Segment in each Quadrant as long as there's balanced voting?	
➡ NASUCA: Would like to think it over before deciding.	
PPL Services: Believes that both Quadrants need to discuss this before deciding. NatFuel agrees.	
➡ NASUCA: The transition language ought to be the same for both Quadrants.	
 Interim Directors: REQ: This requires a new election to select additional Directors to be held within 60 days of a Segment reaching the 8 Member threshold. Same provision for EC. 	13. RGQ: No provision.
Many definitions regarding Membership (Voting versus nonvoting and so on) in NAESB Bylaws. They do not make it clear that consumer advocacy groups can be members of the End User Segment within NAESB. Must have provision that allows representatives of consumers be Members of NAESB. Language in REQ addresses NASUCA's concern, except for Membership dues. NAESB does not define End User; the individual Quadrants do. No prohibition in language of Bylaws.	Solution that satisfies NASUCA: Change or eliminate definition of End User in section 2 and revert to language in 3.1.b regarding Voting Membership.
➡ ELCON: Does not agree with treatment of trade associations. Also agrees with NASUCA on the dues. Disagrees with definition of trade associations as being underrepresented. ➡ Bylaws issues. ➡ Bylaws leaves it up to the Quadrants to determine status of trade associations (whether their members are un-represented or underrepresented). Requests REQ to eliminate 3.2.b definition. Will take under advisement. ➡ IMP: Nonmembers can attend all subcommittee and task force meetings and vote; the exception is at the EC and Board level.	Also it is up to the Quadrant to decide what "under- represented" or "un-represented "mean.

 Exo Link (Services – EDI etc.): 3.2.b What does "significant business interest " mean? Brings possibility of exclusion from membership by other competitive members. ELCON: Strongly suggests replacing "significant" (in RGQ) and "substantial" (in REQ) with "legitimate." 	 Voting to accept Memberships is a simple majority vote. It's up to the Segment Membership to vote affirmatively (simple majority) to deny membership. Also votes are recorded. NAESB principle: inclusiveness. This provision was included to ensure that participants did actually have an interest in the retail gas versus the retail electric industry or both.
Transition question. Answer: Submit application and pay dues considered member, unless otherwise determined.	
 NASUCA/AARP: Membership dues issue. \$5,000 is too high to request of consumer advocacy groups. Also cannot represent all state consumer agencies in one voice. Recommendation: That both Retail Quadrants make the provision to allow such organizations to pay a \$500 membership fee for municipals and local consumer agencies. 	 You may very active and involved in the development of the standards without paying dues. Cannot resolve this issue today.

TAB 3

JOINT RETAIL ELECTRIC & RETAIL GAS WORKING GROUP ACTIVITY SUMMARY

Joint Retail Electric & Retail Gas Working Group Participants List

As of 1/22/02

Eric Ackerman, Edison Electric Institute Barbara Alexander, NASUCA/AARP Rick Alston, Old Dominion Cooperative Jeff Anthony, Wisconsin Electric - Wisconsin Gas Mariam Arnaout, American Gas Association Bill Barkas, Dominion Retail **Bill Boswell, Dominion Resources** Bill Bourbonnais, Wisconsin Public Service Jim Buccigross, Group 8760 Andy Burnham, Consumers Energy Carol Busto, Dominion Virginia Power Ward Camp, US Power Solutions Yvette Camp, Southern Company Dolores Chezar, KeySpan Michele Doyle, Consolidated Edison Company of New York Charlie Driggs, Conectiv Craig Goodman, National Energy Marketers Association Karen Hill. American Gas Association Misty Ann Khan, Telerx Ruth Kiselewich, Baltimore Gas & Elecctric Rose Lennon, Washington Gas Mark Maassel, Nisource Jean Mason, Ameren Services Ann McGlynn, ComEd Jim Minneman, PPL Service Terry Moran, Public Service Electric & Gas Les Nishida, Wisconsin Public Service Mike Novak, National Fuel Bill Oppenheim, Peco Energy Company Stacey Rantala, National Energy Marketers Association Anne Richert, New York State Electric & Gas George Roberts, Schlumberger Sema North America Keith Sappenfield, PanCanadian Energy Richard Schneider, Public Service Electric & Gas Marilyn Specht Adusei-Poku, Nisource Energy Distribution Companies Julie Unruh, Reliant Alonzo Weaver, Memphis Gas Light &Water Marc Wesbster. New York State Electric & Gas Eric Wylen, New York State Electric & Gas Steve Zavodnick, Baltimore Gas & Electric

Joint Retail Electric and Retail Gas Working Group Conference Call – January 22, 2002, 10:00 AM EST

Proposed Agenda

Charlie Driggs, with Conectiv and Chair of UBP-RSB Governance Committee, will lead the call.

- Roll Call; Review Antitrust Guidelines; Review Caller Etiquette
- Agree on Name for this Joint Task Force and Review its Role

The role of this joint group will be to revise the proposals of both Quadrants to 1) make them consistent where applicable; 2) simplify them where possible (i.e. remove language that is already addressed in the NAESB Bylaws); and 3) ensure that the language conforms with the provisions in the NAESB Bylaws. This "REQ-RGQ task force" would then present their recommendations (i.e. the revised drafts) for consideration by all stakeholders at the January 29 meeting. The task force will conducts its work via conference calls and e-mail. It's role is NOT to make the two documents identical.

- Review the Format and Organization of the Retail Electric Quadrant and the Retail Gas Quadrant Procedure Documents (*Please reference Workpaper 1*):
 - Make the organization and formatting of the two consistent where applicable
 - Review proposed organization—aligned with the NAESB Bylaws—submitted by Les Nishida and Bill Bourbonnais (Wisconsin Public Service) (*Please reference Workpaper 2*)
- Review Possible Next Steps (Tasks, Calls, Meetings)
 - For this joint task force
 - Possible meeting dates in February for the Retail Quadrants if needed

Joint Retail Electric and Retail Gas Working Group January 22, 2002 – 10:00 AM EST Conference Call Minutes

Participants:

Eric Ackerman, Edison Electric Institute Barbara Alexander, NASUCA/AARP Rick Alston, Old Dominion Cooperative Mariam Arnaout, American Gas Association Bill Barkas, Dominion Retail Bill Bourbonnais, Wisconsin Public Service Jim Buccigross, Group 8760 Andy Burnham, Consumers Energy Carol Busto, Dominion Virginia Power Ward Camp, US Power Solutions Yvette Camp, Southern Company Michele Doyle, Consolidated Edison Company of New York Charlie Driggs, Conectiv Misty Ann Khan, Telerx Ruth Kiselewich, Baltimore Gas & Electric Rose Lennon, Washington Gas Jean Mason, Ameren Services Les Nishida, Wisconsin Public Service Mike Novak, National Fuel Gas Distribution Company Anne Richert, New York State Electric & Gas Richard Schneider, Public Service Electric and Gas Marilyn Specht Adusei-Poku, Nisource Energy Distribution Companies Marc Webster, New York State Electric & Gas David Wojick, Electricity Daily Eric Wylen, New York State Electric & Gas Steve Zavodnick, Baltimore Gas & Electric

This joint working group met via conference call to review the organization and formatting of the draft NAESB Retail Electric Quadrant (REQ) Procedures and the draft NAESB Retail Gas Quadrant (RGQ) Procedures documents. Charlie Driggs (Conectiv) led the call. Two documents were presented by Wisconsin Public Service as work papers. Upon reviewing the antitrust guidelines, the working group completed the following business:

• AGREED TO THE NAME AND LIMITED ROLE of this working group. Call participants agreed, for clarity purposes, to refer to this temporary working group as the Joint Retail Electric & Retail Gas Task Force. Participants also agreed that this joint working group would operate for a limited period to accomplish a limited purpose, which is to make the NAESB REQ Procedures and the NAESB RGQ Procedures consistent in organization and format, where possible.

• AGREED TO USE THE ORGANIZATION AND FORMAT OF THE NAESB BYLAWS to make the two documents more consistent. Call participants reviewed the work papers presented by Wisconsin Public Service and accepted the recommendation to follow the NAESB Bylaws format for both documents and to tag sections that do not apply to a given Quadrant with the term "RESERVED."

• IDENTIFIED MILESTONES FOR GETTING THE TWO NAESB RETAIL QUADRANTS UNDERWAY:

- Finalized: NAESB REQ Procedures and NAESB RGQ Procedures
- Finalized: NAESB Retail Electric Segment Procedures and NAESB Retail Gas Segment Procedures
- 30-day membership recruitment drive completed: For the NAESB REQ and for the NAESB RGQ
- REQ populated (i.e. Members join NAESB and pay Membership fee)
- RGQ populated (i.e. Members join NAESB and pay Membership fee)
- NAESB Retail Electric Segments representatives elected to the NAESB Board and to the REQ Executive Committee (EC)
- NAESB Retail Gas Segments representatives elected to the NAESB Board and to the RGQ EC
- NAESB Board meeting March 7th, 2002
- NAESB Board adopts Retail Electric Quadrant Procedures and Retail Gas Quadrant Procedures (including Segments procedures)
- NAESB Parliamentary Committee completes drafting the NAESB Operational Procedures
- NAESB Board adopts NAESB Operational Procedures

• AGREED THAT ALL PARTIES MUST PROCEED SWIFTLY—YET DELIBERATELY—TO ACCOMPLISH THE STEPS NECESSARY TO SET UP BOTH RETAIL OUADRANTS. Participants

acknowledged that it might NOT be feasible to complete all of the above steps in a thorough and satisfactory manner before March 7th. It was understood that in such case, the NAESB Board would schedule another meeting to approve the Exhibits to its Bylaws.

OUTLINED POSSIBLE NEXT STEPS for NAESB Retail Electric and Retail Gas Quadrants Outlined

Stakeholders as follows:

- 1. Revise organization of Retail Electric Quadrant Procedures (Volunteers: Eric Ackerman; Bill Bourbonnais; Carol Busto; Charlie Driggs; Ruth Kiselewich; and Les Nishida)
- 2. Revise organization of the Retail Gas Quadrant Procedures (Volunteers: Mariam Arnaout, AGA; Bill Bourbonnais; Les Nishida; and Steve Zavodnick)
- 3. Complete procedures for Segments of the Retail Electric and of the Retail Gas Quadrants, respectively (To be presented during the 1/29/02 Retail Quadrants Open Industry Meeting)
- 4. Recruitment efforts
 - List of contacts for each of the eight Segments (Volunteers: REQ Suppliers, _____; REQ Distributors, Eric Ackerman; REQ End Users, _____; REQ Services, Jim Minneman and Misty Ann Khan; RGQ Suppliers, _____; RGQ Distributors, Mariam Arnaout; RGQ End Users _____; RGQ Services, Jim Minneman and Misty Ann Khan)
 - Recruitment tool proposed by Mike Novak and Charlie Driggs: Conceptual white paper, or options paper, outlining activities that might be undertaken during the first year of the two NAESB Retail Quadrants. To be discussed further during the 1/29/02 industry meeting.
- 5. Monitor and present comments on the drafting of the NAESB Operational Procedures (Both Quadrants and all Segments)

• AGREED THAT MORE MEETINGS WILL BE NECESSARY IN FEBRUARY. Mid-February and late-February were suggested as possible times to hold meetings.

1	(This document is an attempt to place the text of the Retail Electric Quadrant Procedures
2	into the same structure as the NAESB Bylaws. None of the original text was changed or
3	deleted but the original text was cut and moved to better fit the structure of the NAESB
4	Bylaws. All section numbers correspond to the Article numbers that appear in the
5	Bylaws. Sub-section numbers do not in all cases correspond to the section numbers of
6	the Bylaws. Sections 5 and 6 of the original text was split up with portions going into the
7	policy section of this document and part into the EC section of this document.)
8 9	
9	
10	
11 12	
12	
14	
15	
16	NORTH AMERICAN ENERGY STANDARDS BOARD
17	BYLAWS ADDENDUM
18	
19	EXHIBIT 4
19 20 21	
22	RETAIL ELECTRIC QUADRANT PROCEDURES

23

52

64

65

66

67

68

74

24 Section 1.0 DEFINITIONS 25

- 26 1.1 "Distributor Segment Member" means a person (e.g., corporation or 27 partnership) engaged in the local distribution of electricity, or its representative 28 (e.g., employee, officer, director, agent, or partner) who is authorized to cast 29 that Segment Member's vote, that satisfies the requirements for Membership 30 set forth in Article 5 of the bylaws of NAESB, and Section 3 of these Procedures, 31 and further satisfies the qualifications for Distributor Segment Membership of 32 the Retail Electric Quadrant as set forth in the Distributor Segment Procedures. 33
- 34 "End User Segment Member" means a person (e.g., corporation, partnership, 1.2 35 proprietorship, or individual) engaged in the consumption of electricity 36 delivered by an Electric Distributor, or its representative (e.g., employee, 37 officer, director, agent, or partner) who is authorized to cast that Segment 38 Member's vote, that satisfies the requirements for Membership set forth in 39 Article 5 of the Bylaws of NAESB, and Section 3 of these Procedures, and 40 further satisfies the qualifications for End User Segment Membership of the 41 Retail Electric Quadrant as set forth in the End User Segment Procedures. 119 42
- 43 1.3 "NAESB Office" means the office of the Secretary of NAESB44
- 45 1.4 "Quorum" is defined, for the purposes of a Quadrant EC meeting, as a minimum of one EC member representative from each Segment within the Retail Electric Quadrant, and at least half of the currently serving Retail Electric Quadrant EC members overall, present either in person or via telecommunications. and as stated in Section 6.6 of the NAESB Bylaws in all other respects.
- 51 1.5 "REQ" means the Retail Electric Quadrant
- 53 1.6 "Segment" means one of the co-equal Membership Segments of the NAESB
 54 Retail Electric Quadrant representing the following four (4) segments of the
 55 retail electric industry: Suppliers, Distributors, End Users, and Services.
 56
- 57 1.7 "Segment Membership" means the Segment Members collectively.58
- 1.8 "Services Segment Member" means a person (e.g., corporation, partnership or proprietorship or individual) that provides services to participants in the retail
 61 electric industry or an entity not otherwise declared in another segment, or its
 62 representa tive (e.g., employee, officer, director, agent, or partner) who is
 63 authorized to cast that Segment Member's vote, that
 - a) satisfies the requirements for Membership set forth in Article 5 of the Bylaws of NAESB, and Section 3 of these Procedures, and
 - b) further satisfies the qualifications for Services Segment Membership of the Retail Electric Quadrant as set forth in the Services Segment Procedures.
- 1.9 "Supplier Segment Member" means a person (e.g., corporation, partnership, or
 proprietorship) engaged in the competitive sale of electricity to end-users, or its
 representative (e.g., employee, officer, director, agent, or partner) who is
 authorized to cast that Segment Member's vote, that
 a) satisfies the requirements for Membership set forth in Article 5 of the
 - a) satisfies the requirements for Membership set forth in Article 5 of the Bylaws of NAESB and Section 3 of these Procedures, and

- b) further satisfies the qualifications for Supplier Segment Membership of the Retail Electric Quadrant as set forth in the Supplier Segment Procedures.
- 1.10 All other capitalized terms, if not defined in this Section 2.0, shall have the same definitions as specified in the Bylaws or Certificate of Incorporation of NAESB.

82 Section 2.0 PURPOSES, SCOPE, ACTIVITIES AND POLICIES 83

84 2.1 Purpose 85

The purpose of the Retail Electric Quadrant of the North American Energy Standards Board (NAESB) is to propose, evaluate and adopt voluntary standards and model business practices to promote competitive, efficient and reliable service in the retail electric industry.

91 2.2 Scope 92

The Retail Electric Quadrant is to address issues and practices that are within the
 scope of NAESB and appropriate to electric usage at the individual consumer
 level—that is, usage by an individual, partnership, corporation, or other entity
 consuming electricity at one or more facilities served by an Electric Distributor.

98 Issues may also be addressed in conjunction with other Quadrants when such
99 issues are common to both electric distribution at the individual consumer level
100 and some combination of wholesale or retail activities in the electric or natural
101 gas industries.

102

75

76

77

86

87

88

89

90

103 Explanatory Note:

104 The Retail Electric Quadrant would normally address issues and practices which 105 are generally within the jurisdiction of state or local regulatory authorities. As 106 such, the findings, recommendations and proposed standards or model business 107 practices produced by this Quadrant of NAESB may be implemented in individual 108 jurisdictions only to the extent that they are consistent with applicable law and 109 regulation. Standards and model business practices that prove successful in those 110 states that adopt them might well be adopted by additional jurisdictions if 111 compatible with policies in their jurisdiction. 112

113 Furthermore, the Retail Electric Quadrant will operate in conformance with the 114 principles of NAESB as established in Section 2.2 (b) of the NAESB Bylaws, and in 115 compliance with ANSI guidelines as codified therein . In keeping with these 116 Principles, the Retail Electric Quadrant encourages membership from across the 117 spectrum of the Retail Electric industry, reflecting the principle of inclusiveness. 118 All Retail Electric Quadrant meetings, including those of its Members, Executive 119 Committee (EC), subcommittees and task forces, shall be open to any interested 120 member of the public and that all minutes thereof shall be available to the public 121 in keeping with the principle of openness. 122

123 2.3 Policy

- a) All meetings held in association with the NAESB organization or the REQ are open to any interested person.
- 126 127

124 125

130 business. 131 132 (i) Designation of the same individual as the duly authorized representative of 133 the Member in multiple Segments of the REQ (including Agent 134 representatives) is not permitted. 135 (ii) In the event that a duly authorized representative in one Segment also 136 represents the Member in another Segment in any other NAESB Quadrant, 137 and both affected Segments are engaged in a joint meeting requiring 138 Member voting, then the person serving as representative in both 139 Segments must choose the Segment and Quadrant in which they will vote 140 on any given issue and may only cast one vote on each voting issue. To 141 the extent that any other involved Quadrant's rule may differ on this 142 point, the representative must still honor this Quadrant's rule for any such 143 vote or designate a substitute representative for the entity's interests 144 during the joint meeting. 145 146 Note: The provisions of Section 2.3 b (ii) deal with an issue applicable to all 147 Quadrants and may be more appropriate for inclusion in either the Bylaws or 148 NAESB Operating Procedures. In addition, since nearly all votes by a Member in 149 NAESB will either be for election of officers or ratification, joint action with 150 another Quadrant requiring voting should be very rare, and may never actually 151 occur. Need to decide whether to keep or drop this. 152 153 c) Only REQ Voting Members have the right to ratify standards or model 154 business practices adopted by the Quadrant EC. Ratification voting will be in 155 accordance with the requirements of the NAESB Bylaws. 156 157 158 Section 5.0 MEMBERS 159 160 The definition of Membership applies to all Segments within the REQ. Members of 161 the REQ are persons who meet the following membership requirements in any 162 one of the four Segments of this Quadrant. Upon applying for Voting Membership 163 (as described in NAESB Bylaws Sections 1.1 and 5.1) in the REQ, each 164 prospective Member shall declare the Segment with which they are to be 165 identified. 166 167 5.1 Segment Membership Eligibility 168 169 a) Distributor Segment 170 Parties eligible for membership in this Segment are engaged in the local 171 distribution of electricity. 172 173 b) End User Segment 174 Parties eligible for membership in this Segment consume electricity or 175 represent consumers of electricity. 176 177 c) Suppliers Segment 178 Parties eligible for membership in this Segment are service providers and 179 marketers, engaged in the competitive sale of electricity to end-users 180

b) From time to time, there will be joint meetings of the REQ with other

Quadrants within NAESB, and Segments will meet jointly to transact Quadrant

128

129

181 182 183	 d) Services Segment Parties eligible for membership in this Segment provide services to participants in the retail electric industry, including equipment manufacturers,
184 185 186	equipment vendors, software providers, consultants, and other companies or individuals not otherwise eligible for membership in another Segment.
187 188	5.2 Voting Membership
189	Membership and voting rights in the REQ of NAESB shall be open to any Member
190 191	that meets the following requirements:
192	a) The Member has a significant business interest $_1$ in the retail electric market
193	(or is a representative or Agent of such person), as determined by a simple
194	majority of the segment membership (As used in this section, "significant business
195 196	interests" specifically includes the interests of statutorily appointed consumer advocates)
197	b) The Member may be a trade association or an advocacy group representing a
198	group of prospective members, provided that
199	
200	(i) the group represents an interest that is identified within a Segment, yet is
201	(a) not represented, or
202	(b) is underrepresentedin the judgement of a simple majority of the
203	Segment membership, and
204	(ii) The prospective representative's relationship to the group wishing to be
205	represented in this manner is documented in one of the following ways:
206	(a) Has statutorily or regulatory designated authority to represent the
207	interests of the subject group
208	(b) Is generally recognized representative of the subject group for other
209	transactions (e.g. provides group insurance, group buying of another
210	product, etc.)
211	(c) provides a letter of authorization from the group, including an
212	expiration date for the letter of authorization
213	(iii)no more than two entities within the group represented subsequently join
214	as full Members of the Segment, at which point the representative's
215	authorization for representation and voting rights will be immediately
216	voided unless reinstituted by letter for the remainder of the operating year
217	by the remaining members of the representative's constituency.
218	
219	Note: Per provisions of the NAESB Bylaws, the voting rights of any REQ Member
220	shall be suspended if its dues are delinquent, and attendance of that delinquent
221	Member at any meeting shall not be counted in determination of a Quorum being
222	present.
223	
224	
225	5.3 Multiple Memberships
226	
227	Memberships in multiple Segments of the REQ are permissible for any person
228	provided each Membership is filed and declared with NAESR, the person meets

228 provided each Membership is filed and declared with NAESB, the person meets 229 the membership requirements of each Segment joined, and Membership dues are 230 paid for each Segment. 231

a) Only one Membership per Segment is permissible for any person.

- 232 b) Multiple companies under common control within a corporate organization 233 that desire to become members must join individually. Members cannot 234 extend their membership to their parent company, affiliates, or subsidiaries. 235
- 236 5.4 Member Representation 237

238

239

240

241

248

249

251 252

253

254 255

258

259

260

261

262 263

264

265

266

267

268

269

270

271

272

273

274

275 276

277

278

Representatives chosen by any person in any Segment should have, but are not required to have, corporate responsibility for the business interests causing the person to seek membership in the Segment.

242 5.5 Removal of Members 243

244 Members who cease to have a significant business interest pertaining to the 245 descriptions contained in sections 3.1(a)-3.1(d) may be removed from 246 membership by either 247

a) a simple majority vote of their declared Segment membership, or

b) resolution of the REQ Executive Committee

250 Section 7.0 BOARD

The REQ shall elect representatives to the Board of Directors of NAESB and elect an Executive Committee (EC), which shall participate in the larger NAESB EC body as well as govern the activities of the REQ.

- 256 7.1 Number and Election of Directors 257
 - a) The REQ shall elect sixteen (16) NAESB Directors, subject to the provisions of Section 8 of these Procedures. All Directors from the REQ will be comprised of an equal number of representatives elected from each Segment of the Quadrant.
 - b) Each Segment of the Quadrant will elect four (4) Directors, subject to the provisions of Section 8 of these Procedures. Nominations for and election of all Directors will be in accordance with Segment Procedures.
 - c) In the event that a Director resigns or otherwise vacates their Board seat, the Quadrant will hold an election within 60 days to fill the vacant seat with a candidate from the same Segment as the original Director.
 - d) Directors shall be elected for two-year terms, with half of the terms expiring in alternating years.
 - e) Subject to the provisions of Section 8, two Directors will be elected from each Segment each year to fill expiring terms. Group A Directors will have their terms expire in odd numbered years, and Group B Director terms will expire in even numbered years, in conjunction with the end of the operating year of NAESB or as otherwise defined by the Board, Certificate of Incorporation or Bylaws, as amended.
 - f) Directors elected from the REQ may run for re-election without restriction on the number of terms held.
- 279 g) Directors elected from the REQ may not hold both a Board seat and a seat on 280 the Executive Committee at any point in time. If an Executive Committee 281 member is elected as a Director from the REQ, their EC seat is vacated when 282 the NAESB Board seats them as a Director.
- 283 h) No two Directors elected by the REQ may be employees of the same Member 284 holding Membership in multiple Segments within the Quadrant. This
- restriction does not prohibit election of two Directors from two affiliated
 companies within a holding company which each have individual Member
 status, or from two companies with a parent-subsidiary relationship, provided
 that the two Directors from companies with such a relationship represent
 Members of differing Segments.
- 290 [Example: In the LDC Segment, the Exelon Corporation might have interests 291 in multiple REQ Segments. If Exelon joins all segments as the Exelon, rather 292 than as its operating subsidiaries doing business in those Segments, Exelon 293 could only have one Board member elected from the REQ, period. If instead, 294 Exelon joined each Segment via its operating subsidiaries active in each 295 Segment's portion of the industry, one or more Board members representing 296 interests within the Exelon corporate structure could be elected from the REQ 297 Segments and serve concurrently. However, if mulitiple Exelon operating 298 subsidiaries joined one Segment (Commonwealth Edison and PECO Energy 299 joining the LDC Segment, for example), only one Board seat could be held by 300 representatives of that Segment. Intent is to prevent one company / member 301 from being able to "load" the Board in their favor, particularly in representing 302 the interests of a single Segment. The rule, as proposed, counts on the 303 diversity of market presence between Segments to tend to limit the number 304 of Board seats affiliated with a single holding company across Segments.] 305
- i) In preparation for any election of NAESB Directors (other than initial 306 Directors, as provided for in Section 8), a Nominating Committee consisting of 307 five EC members of the REQ comprised of one member from each Segment 308 plus the EC Vice-Chair, shall develop a slate of candidates from the Segment 309 Membership. Other nominations may be made at or prior to the close of the 310 REQ End User Segment nomination period by any Segment Member eligible to 311 serve on the Board of Directors by submitting their name(s) to the NAESB 312 Office in a form as specified in NAESB Operating Procedures if such 313 requirements exist. All nominations must be made and conveyed in writing to 314 the NAESB Secretary no less than 31 days prior to the election date.

316

317

318

319

320

321

322

323

324

325

326

327

328

329

330

331

332

333

334

335

336

- j) Election of Directors shall occur in the same month for all Segments of the REQ, and shall be coordinated by the Executive Committee. The EC will ensure that the requirements of this Section are satisfied by each Segment's slate of candidates prior to any actual voting by any Segment.
- k) In addition to being subject to removal from office by the NAESB Board of Directors, Directors may also be removed from office by the Quadrant membership at any time upon a 75% majority vote of the Quadrant membership and a simple majority (51%) of each of the Segments of the REQ. The vacant seat is to be refilled in accordance with all of the requirements of this Section 4.1.
- Elected Directors may not declare a Designated Alternate for their seat for any reason, and may not delegate their Board vote to another except by Proxy to another Board member.
- m) In the event that the Director changes affiliation to another Member within his/her industry Segment, the Director's term will continue until its natural expiration, provided that there is no other Director already representing the same Member, in which case the Director changing his or her affiliation will vacate their seat for election of a new Director. In the event that a Director is no longer affiliated with their electing industry Segment, the Director will vacate their seat for election of a new Director.
 - n) Meetings of Directors from the REQ will be in accordance with the NAESB Bylaws.

348

349

350

351

352

353

354

355

356

357

358

359

360

361

362

363

364

365

366

Section 10.0 EXECUTIVE COMMITTEE

- 340 10.1 Number and Election of Executive Committee Members341
- a) The REQ shall elect sixteen (16) NAESB Executive Committee members,
 subject to Section 8 of these Procedures, with each Segment electing four (4)
 of these EC Members, pursuant to procedures established by that Segment.
 All EC representation from the REQ will be comprised of an equal number of
 representatives elected from each Segment of the Quadrant.
 The REQ will elect four (4) EC members from each Segment, subject to the
 - b) The REQ will elect four (4) EC members from each Segment, subject to the provisions of Section 8 of these Procedures. Nominations for and election to an EC seat will be in accordance with Segment Procedures.
 - c) In the event that an EC member resigns or otherwise vacates their seat, the Quadrant will hold an election within 60 days to fill the vacant seat with a candidate from the same Segment as the original EC member.
 - d) EC members shall be elected for two-year terms, with half of the terms expiring in alternating years.
 - e) Subject to the provisions of Section 8, two EC members will be elected from each Segment each year to fill expiring terms. Group A EC members will have their terms expire in odd numbered years, and Group B EC member terms will expire in even number years, in conjunction with end of the operating year of NAESB or as otherwise defined by the Board of Directors, Certificate of Incorporation or Bylaws, as amended.
 - f) EC members from the REQ may run for re-election without restriction on the number of terms held.
 - g) EC members elected from the REQ may not hold both a Board seat and a seat on the Executive Committee at any point in time. If an Executive Committee member is elected to the Board from the REQ, their EC seat is vacated when the NAESB Board accepts them as a member of the Board.
- h) No two EC members elected by the REQ may be employees of the same
 Member holding Membership in multiple Segments within the Quadrant. This
 restriction does not prohibit election of two EC members from two affiliated
 companies within a holding company which each have individual Member
 status, or from two companies with a parent-subsidiary relationship, provided
 that the two EC members from companies with such a relationship represent
 Members of differing Segments.
- 374 i) In preparation for any election of EC Members (other than initial EC Members, 375 as provided for in Section 8), a Nominating Committee consisting of five EC 376 members of the REQ comprised of one member from each Segment plus the 377 EC Vice-Chair, shall develop a slate of candidates from the Segment 378 Membership. Other nominations may be made at or prior to the close of the 379 REQ End User Segment nomination period by any Segment Member eligible to 380 serve on the EC by submitting their name(s) to the NAESB Office in a form as 381 specified in NAESB Operating Procedures if such requirements exist. All 382 nominations must be made and conveyed in writing to the NAESB Secretary 383 no less than 31 days prior to the election date.
- j) Election of EC members shall occur in the same month for all Segments of the REQ, and shall be coordinated by the existing REQ Executive Committee. The EC will ensure that the requirements of this Section are satisfied by each Segment's slate of candidates prior to any actual voting by any Segment.
- k) EC members may be removed from office by the Quadrant membership at
 any time upon a 75% majority vote of the Quadrant membership and a

390 391 392 393 394 395	I)	simple majority (51%) of each of the Segments of the REQ. The vacant seat is to be refilled in accordance with all of the requirements of this Section 4.2. Elected EC members may declare representation at an EC meeting by a Designated Alternate named in accordance with Section 7 of these Procedures, for any reason, subject to the EC participation requirements contained in Section 10.4 of the NAESB Bylaws as applied to all Executive
396		Committee meetings.
397 398	m)	In the event that the EC member changes affiliation to another Member within his/her industry Segment, the EC member's term will continue until its
399		natural expiration, provided that there is no other EC member already
400		representing the EC member's new affiliation, in which case the EC member
401		changing affiliation will resign their seat for election of a new EC member. In
402		the event that an EC member is no longer affiliated with their electing
403		industry Segment, the EC member will vacate their seat and a new EC
404	,	member will be elected to fill the seat.
405	n)	The REQ EC shall
406 407		 (i) elevate the prior year Vice Chair of the Quadrant EC to Chair of the REQ EC at its first meeting in the new operating year and elect a new Vice
407		Chair. If the Vice Chair is vacant at the time of the first meeting of a new
409		operating year, both a Chair and Vice-Chair will be elected, and
410		(ii) designate an EC Secretary responsible for production of meeting minutes
411		per NAESB Operating Procedures. This may involve an individual assigned
412		this responsibility on a permanent basis, or an agreement on how to
413		designate a temporary Secretary for the purpose of recording minutes of
414 415		each meeting.
415	o)	An EC member shall cease to be a member of the EC upon (i) the resignation of REQ End User Segment Membership by the Member
417		whom they represent
418		(ii) the lapse of the Segment Membership dues of the Member whom they
419		represent,
420		(iii) the EC member's resignation, removal, or death
421		(iv) the first day of the NAESB operating year following an election for the seat
422		held by the EC member, wherein the member was not re-elected by the
423 424		Segment Membership as declared by the NAESB Secretary
424 425		 (v) receipt by the NAESB Secretary of a rescission letter indicating that the individual no longer represents the Segment Member
426	n)	A vacancy resulting from removal of an EC member shall be filled for the
427	P7	remainder of that term in accordance with the following requirements
428		(i) Any EC seats becoming vacant shall be filled by a special election subject
429		to the following provisions
430		(ii) If the vacancy occurs with less than 125 days remaining in the EC
431		member's term of office, the seat will remain vacant until filled by election
432		of a new EC member to a term commencing in the next operating year,
433 434		and a Designated Alternate will serve until expiration of the term.
434		(iii) All other vacancies will be filled by special election; such special elections are to be formally noticed to the Segment Membership by the Segment
436		Vice-Chair or Chair within 15 days of when the creation of the vacancy
437		becomes known and held within 60 days of the creation of the vacancy.
438		Candidates will be designated by the normal nomination process or
439		selected by the remaining EC members of the Segment from Segment
440		Membership. Candidates elected to fill the vacancy will serve the
441		remainder of the term, and may stand for re-election.

447

448

(iv)Until an election is held and a new EC member is elected, a Designated Alternate will serve in the open EC seat.

445 10.2 EC Meetings446

- a) Meetings of EC members from the REQ will be in accordance with the NAESB Bylaws.
- b) The Quadrant EC Chair and Vice-Chair will serve in the roles defined for the
 NAESB EC Chair and Vice-Chair positions in the NAESB Bylaws and chair all
 Quadrant EC meetings.
- 452 c) In the event that the EC of the REQ meets jointly with an EC of another
 453 NAESB Quadrant, the choice of Quadrant EC Chair presiding over the joint
 454 meeting will be determined by the precedence established in the order of
 455 rotation of Vice-Chairs as specified in the NAESB Bylaws.
- d) In the event that the provisions of 4.3(c) result in the REQ EC Chair or ViceChair presiding over a joint meeting, the actions taken to comply with Section
 4.2(m)(ii) will determine what person is responsible for taking and preparing
 meeting minutes.
- 460 e) Executive Committee Meetings shall be held at times and locations 461 determined by the Chair or Vice-Chair of the Quadrant Executive Committee. 462 Executive Committee Members may participate and vote by means of tele-463 conference or other electronic means unless in-person attendance is required 464 of all members by both the Chair and Vice-Chair of the Executive Committee, 465 and subject to the attendance requirements of the NAESB Bylaws. Attendance 466 records shall be reviewed by the Secretary of the EC during the course of 467 each Quadrant EC meeting for compliance with NAESB Bylaws for the term of 468 office served by each current Quadrant EC member. The Secretary shall 469 advise any members at risk for failure to comply of their status.
- 470 f) Failure to meet the in-person attendance requirements for reasons other than 471 hospitalization or disability requires issuing a notice of non-compliance to the 472 Quadrant membership and a vote for removal of the Executive Committee 473 seat within 30 days of failing to comply. Election of a new member, if 474 necessary, will proceed as provided for in Section 3. If the membership does 475 not vote for removal of the EC member, the Secretary or EC Chair may call 476 for another removal vote by the Quadrant membership following the next EC 477 meeting if the EC member's attendance record still does not satisfy the 478 Bylaws requirements. 479
- 480 10.3 Voting for Adoption481

482 Only Members of the REQ EC shall have the right to vote to adopt standards or
483 model business practices for the REQ. Quadrant voting procedures will be in
484 accordance with the NAESB Bylaws.

485 486

488

487 10.4 EC Subcommittees and Task Forces

489 Note: The provisions of this Section 4.4 may actually be more appropriately
490 located within NAESB's Operating Procedures. Until such time as there is Board
491 agreement to adopt them for all of NAESB, they remain here as REQ provisions.
492

493The EC of the REQ shall set up its own subcommittees and task forces to deal494with REQ-specific issues.

105	
495	
496	 a) The Executive Committee may establish voluntary standing subcommittees or
497	special purpose task forces to perform various functions required of the
498	organization.
499	(i) The Executive Committee will prepare a written statement of the purpose
500	of the subcommittee or task force and the tasks to be performed, name
501	the subcommittee or task force, state if balanced voting among the
502	Segments is applicable, and appoint a temporary chair.
503	(ii) The Temporary Chair may be a member of the Executive Committee or
504	any REQ Member willing to perform the required startup tasks and to
505	continue chairing the subcommittee / task force if elected by the members
506	its first meeting.
507	(iii) The Temporary Chair shall set up the first meeting of the subcommittee or
508	task force and prepare a meeting notice that:
509	(a) states the name and purpose of the subcommittee / task force,
510	(b) solicits participation in the subcommittee / task force,
510	
511	(c) announces the agenda for the first meeting and shall send the meeting
512 513	notice to all Members and non-members who have participated in
	Quadrant activities in the last twelve months. The notice shall be sent
514	at least two weeks prior to the meeting date. Shorter time periods for
515	notices of subsequent meetings will be permitted by a 75% vote of the
516	participants attending a duly scheduled meeting. All meeting notices
517	shall be posted on the NAESB website and transmitted in writing,
518	facsimile, or other electronic means to parties who have indicated an
519	interest in the duly scheduled meeting. The permanent subcommittee /
520	task force Chair may be any REQ member. Participation on
521	subcommittees or task forces is open to any interested party.
522	 b) Each EC subcommittee or task force will designate a meeting Secretary
523	responsible for recording and reporting voting results, keeping regular
524	minutes of its proceedings and providing copies of those minutes promptly to
525	the NAESB Executive Director's office in accordance to reporting criteria
526	established by the EC.
527	c) Each EC subcommittee or task force will report to the EC at no less than a
528	quarterly basis, on a schedule to be defined by the EC for as long as the
529	subcommittee or task force continues to exist
530	d) Each EC subcommittee / task force will strive for balanced representation and
531	balanced voting between Segments, and participation shall be open to all
532	NAESB Members and other interested parties, without regard to whether the
533	participant is a paid Member of NAESB in good standing.
534	
535	
536	Section 19.0 QUADRANT STARTUP & TRANSITION
537	
538	The requirements of the NAESB Bylaws for a minimum of 5 Members per
539	Segment and 40 Members per Quadrant do not provide for enough members to
540	satisfy the REQ requirements for election of Directors and EC members as stated
541	in Sections 3 and 4. Consequently, until a Segment of the Quadrant has at least
542	eight (8) Members, the number of Directors and EC Members elected by the
543	Segment membership will be halved, to 2 Directors and 2 EC members per
544	Segment and Segment balanced voting will remain intact by adjusting the weight
545	of all votes to achieve balance between Segments in the REQ. The following
546	provisions will apply and supercede the requirements of Sections 3 and 4 until
547	the REQ Segment achieves a minimum of eight (8) members.
-	

549 19.1 Initial Election of Directors 550

Acceptance of the REQ by the Board will involve seating of the REQ Directors at
the NAESB Board meeting at which an acceptance vote is cast. This requires that
there be REQ Directors who may be seated.

555 The founding membership of the REQ will elect Directors of their choosing by 556 means that are reasonably consistent with NAESB Bylaws and the requirements 557 of Section 4.1 of these Procedures, subject to the conditions addressed by 558 Section 8.2 this Section. Selection of candidates and their election will be by 559 procedures agreed to by consensus or voting methods adopted by the group. 560 Such elected representatives will be presented to the NAESB Board for 561 acceptance as representatives of the REQ, and acceptance by the Board will place 562 all the requirements and restrictions of these Procedures upon those individuals.

563 564

565 566

567

568

569

570

571

572

573

574

577

578

579

580

581

582

583

584

585

586

548

19.2 Term of Office of Initial Directors

- a) In the event that Directors are elected from a Segment with at least five (5), but less than eight (8) members, two Directors will be elected, each to serve a 3 year term, and both will be either Group A or Group B directors as determined by the expiration year of their original term.
 - b) In the event that a Segment has eight (8) or more members, four Directors will be elected. The two candidates having the most votes will serve three (3) year terms. The remaining two Directors elected by lesser vote counts will serve two year terms.
- 575 19.3 Interim Elections of Directors576

In the event that any Segment within the Quadrant achieves a minimum membership of eight Members after operating with less for a period of time after an election held under the requirements of Section 8.1, then

- a) a second election of Directors shall be held within 60 days of the addition of the eighth Member of the Segment in order to select two additional Directors from that Segment to serve alongside the initially elected Directors;
- b) the term of office of the second group elected at this time will expire one year prior to the term of the initially elected Directors, thereby allowing for two Group A and two Group B Directors from each Segment.
- 587 19.4 Initial Election of Executive Committee588

Acceptance of the REQ by the Board will involve seating of the REQ Directors at
the NAESB Board meeting at which an acceptance vote is cast. Initial operation of
the REQ will require that an Executive Committee be in place.

593 The founding membership of the REQ will elect an Executive Committee of their 594 choosing by means that are reasonably consistent with NAESB Bylaws and the 595 requirements of Section 4.2 of these Procedures, subject to the conditions 596 addressed by Sections 8.5 of this Section. Selection of candidates and their 597 election will be by procedures agreed to by consensus or voting methods adopted 598 by the group. Such elected representatives will be presented to the NAESB Board 599 for acceptance as representatives of the REQ, and acceptance by the Board will

- place all the requirements and restrictions of these Procedures upon those
 individuals.
- 603 19.5 Term of Office of Initial Executive Committee Members 604
- The initial group of EC members will serve a 2 year term of office as either Group
 A or Group B members, as determined by the expiration year of their original
 term of office.
- a) In the event that EC members are elected from a Segment with at least five
 (5), but less than eight (8) members, two EC members will be elected, each
 to serve a 3 year term, and both will be either Group A or Group B EC
 members as determined by the expiration year of their original term. The
 candidate with the greatest number of votes will be designated the EC Chair;
 in the event of a tie vote, a coin toss will determine the EC Chair and ViceChair positions.
- b) In the event that a Segment has eight (8) or more members, four EC
 members will be elected. The two candidates having the most votes will serve
 a three (3) year term. The remaining two EC members elected by lesser vote
 counts will each serve a two year term. A separate election within the EC will
 determine which member becomes the Chair and which becomes the ViceChair, by vote of the four EC members.
- 622 19.6 Interim Elections of EC Members

In the event that the Segment achieves a minimum membership of eight Members after operating with less for a period of time following an election held under the provisions of Section 8.5, then

- a) a second election of EC members shall be held within 60 days of the addition of the eighth Member of the Segment in order to select two additional EC members from the Segment to serve alongside the initially elected EC members;
- b) the term of office of the second group elected at this time will expire one year
 prior to the term of the initially elected EC members, thereby allowing for two
 Group A and two Group B EC members from each Segment.
 c) The previously elected Chair and Vice-Chair of the Quadrant EC will continue
 - c) The previously elected Chair and Vice-Chair of the Quadrant EC will continue in their official role until the end of the operating year. Chair rotation will proceed per the requirements of Section 4.2(m)(i).
- 638 19.7 Balanced Voting During The Transition Period

Any Segment which has elected Directors or EC members under the provisions of
this section will retain the full Segment votes entitled to it under the Balanced
Voting principle of the Bylaws, and its representatives' votes will count for either
one or two Segment votes as is appropriate to the situation.

- 645 Section 20.0 EC DESIGNATED ALTERNATES
- 647 20.1 EC Designated Alternate Authority648
- 649 Any person presenting themselves at an EC meeting as a Designated Alternate
- 650 will be accepted as a participant provided that
- 651

646

623 624

625

626

627

628

629

630

635

636

652 a) An EC Member from that Segment either indicates to the NAESB Office, the 653 EC Chair or Vice-Chair that they will be absent, or is in fact absent and 654 remains absent. 655 b) The name of the Designated Alternate is on a list of approved Designated Alternates selected by the appropriate Segment membership according to 656 657 these Procedures and on file with both the NAESB office and the EC. 658 659 20.2 Election of Designated Alternates 660 661 a) Each Segment will select no more than three (3), and no less than one 662 Designated Alternate EC representative in each election year, allowing for up 663 to twelve (12) Designated Alternates overall. b) Designated Alternates will be selected annually by the same election that 664 determines new EC members, with those unsuccessful EC member 665 666 candidate(s) receiving the greatest number of votes becoming a Designated 667 Alternate in the event that there already are or are expected to be less than 668 three Designated Alternates from the Segment in the coming year. c) A Designated Alternate will attend and vote at meetings of the EC of the REQ 669 670 or of the larger EC of NAESB, when one or more regular EC Members from the 671 Segment are unable to attend a meeting of the EC and when the absent 672 member or the Vice-Chair of the REQ EC requests their attendance as a 673 Designated Alternate. 674

1	(This document is an attempt to place the text of the Retail Gas Quadrant Procedures into
2	the same structure as the NAESB Bylaws. None of the original text was changed or
3	deleted but the original text was cut and moved to better fit the structure of the NAESB
4	Bylaws. All section numbers correspond to the Article numbers that appear in the
5	Bylaws. Sub-section numbers correspond to the section numbers of the Bylaws. A
6	Section 20 was added in order to include the Selection of Alternates into the Retail Gas
7	Quadrant Procedures.)
8	
0	
9	
10	
11	
12	NORTH AMERICAN ENERGY STANDARDS BOARD
12	NORTH AMERICAN ENERGY STANDARDS BOARD
13	BYLAWS ADDENDUM
14	EXHIBIT 3
15	RETAIL GAS QUADRANT PROCEDURES
13	RETAIL GAS QUADRANT I ROCEDURES

16	
17 18	Section 1.0 Definitions
10 19 20	1.1 "Board" means the Board of Directors of NAESB.
20 21 22	1.2 "Bylaws" means the corporate bylaws of NAESB as amended from time to time.
23 24 25	1.2 "Designated Alternate" is defined as a person named by a Segment in a list, submitted to the NAESB office, to serve in place of an EC Member who is unable to attend an EC meeting.
26 27	1.4 "Director" means an individual serving on the NAESB Board.
28 29 30	1.5 Unless otherwise noted, the "EC" means the Executive Committee of the Retail Gas Quadrant.
31 32	1.6 "NAESB" means the North American Energy Standards Board, Inc.
33 34 35 36	1.7 All other capitalized terms, if not defined in this Section 1.0, shall have the same definitions as specified in the bylaws or certificate of incorporation of NAESB.
37 38	Section 2.0 Purposes, Scope, Activities and Policies
39 40	2.1 Purposes, Scope, and Activities
40 41 42 43 44 45 46 47	The purpose of the Retail Gas Quadrant of NAESB is to propose and adopt voluntary model business practices or standards to promote more competitive, efficient and reliable service in the retail natural gas industry. The Retail Gas Quadrant is concerned with and tasked to handle natural gas related issues and practices that are within the scope of NAESB and typically addressed at the retail natural gas distribution level.
48 49 50 51 52 53 54 55 56 57 58 59 60	The Retail Gas Quadrant shall work closely with other NAESB Quadrants to mitigate inconsistencies where proposed standards and model business practices affect those other Quadrants. The Retail Gas Quadrant shall comply with the policies and procedures laid out in the bylaws and the certificate of incorporation of NAESB. As such, the Retail Gas Quadrant encourages a widely based membership and has identified Segments, which reflect this principle of inclusiveness. Also consistent with this approach, the Retail Gas Quadrant shall ensure that all meetings, including those of its Members, Executive Committee (EC), and Subcommittees and Task Forces, shall be open to all persons and that all minutes hereof shall be available to the public. Further, the Retail Gas Quadrant intends to operate in conformance with the principles of the umbrella organization, as established in Section 2.2 (b) of the NAESB Bylaws, and in compliance with ANSI guidelines.
61 62 63 64	Each Segment of the Retail Gas Quadrant shall elect representatives from the Quadrant to the NAESB Board of Directors (Board). Also the Retail Gas Quadrant shall have an Executive Committee (EC), which shall also participate in the larger NAESB EC body.

6566 2.2 Policies

67

83

85

92

94

All meetings held in association with the NAESB Retail Gas Quadrant shall be open to all interested persons. Only Retail Gas Quadrant EC Members shall have the right to vote to adopt model business practices or standards affecting only the Retail Gas Quadrant. Only Retail Gas Quadrant Voting Members have the right to ratify model business practices and standards that were adopted by the Quadrant EC and that affect only the Retail Gas Quadrant.

75 At NAESB joint Subcommittee or joint Task Force meetings, the same individual may represent different Segments from different Quadrants, as long as that individual 76 declares prior to the joint meeting the Quadrant and Segment for which he or she is 77 casting a vote. At Retail Gas Ouadrant Subcommittee or Task Force meetings, the 78 79 same individual may represent different Segments, as long as that individual declares prior to the meeting the Retail Gas Quadrant Segment for which he or she is casting a 80 vote. However, at the EC and Board level, an elected representative may hold not 81 more than one seat, representing only one Segment within one Quadrant. 82

- 84 2.3 Quadrants and Segments
- What follows is a description of the Retail Gas Quadrant organizational structure,
 including a listing of Retail Gas Quadrant Segments.
- Suppliers: Persons, other than distributors, engaged in the sale of natural gas and/or
 pipeline capacity, including marketers (retail, wholesale), aggregators, producers,
 asset managers and pipelines.
- 93 Distributors: Persons engaged in the regulated local distribution of natural gas.
- End Users: Persons that consume natural gas, including consumer advocacy
 organizations and retail customer groups.
- Service Providers: Persons that provide services to the participants in the retail natural
 gas industry, including equipment manufacturers, equipment and service vendors,
 software providers, energy consultants, and other companies not otherwise declared
 in any other Segment.
- Each Segment within the Retail Gas Quadrant shall have the flexibility to determine
 its own Segment rules and procedures, as long as they conform with NAESB Bylaws
 and Retail Gas Quadrant procedures and to elect or select its own representatives to
 the NAESB Board and the Retail Gas Quadrant EC.
- 107
- 108
- 109 Section 3.0 Offices (Reserved)110
- 111
- 112 Section 4.0 Seal (Reserved)
- 113

114 115	Section 5.0 Members
116 117	5.1 Voting Members
118	
119	The definition of Membership applies to all Segments within the Retail Gas
120	Quadrant. Members of the Retail Gas Quadrant are persons with significant interest in
121	the retail natural gas market and which meet the definitions of one of the four
122	Segments. Upon applying for Voting Membership (as described in NAESB Bylaws
123	Sections 1.1 and 5.1), each prospective Member shall declare the Segment with which they are to be identified. Voting Membership in the Retail Gas Quadrant of NAESB
124 125	shall be open to all persons that meet all of the following requirements:
125	- The person has a substantial interest in the retail natural gas market (or is a
120	representative or Agent of such person).
128	<u>r</u>
129	- The person meets the description of one of the co-equal industry Segments identified by the Retail
130 131	Gas Quadrant of NAESB.
131	- The person has declared its affiliation to one, and only one, of the Segments of the Retail Gas
133	Quadrant.
134 135	- Through special application, which includes a recommendation by its Segment and an affirmative
135	vote of% of the Quadrant EC, a trade association may join the Retail Gas Quadrant
137	Membership if it represents an interest that is identified within a Segment yet is not represented, or
138	is underrepresented, in the Quadrant Membership. The membership status of such trade
139 140	associations shall be considered on an annual basis and may be voted upon through notational vote by the Quadrant EC.
141	
142	5.2 Non-Voting Members – (Reserved)
143 144	5.3 Nontransferable – (Reserved)
145	
146	5.4 Resignation – (Reserved)
147 148	
140	Section 6.0 – Meetings of Members – (Reserved)
150	
151 152	Section 7.0 Board
152	Section 7.0 Doard
154	7.1 Board
155	
156	The Retail Gas Quadrant shall have representatives to the NAESB Board of
157	Directors, with each Segment electing or selecting an equal number of these
158	representatives, pursuant to the procedures specified by that Segment. The initial
159	Retail Gas Quadrant Directors on the NAESB Board shall be divided into two groups
160 161	within each Segment. Group A Directors shall serve for an initial term of three years, while Group B Directors shall serve for an initial term of two years. Each Segment
162	shall have an equal number of Group A Directors and an equal number of Group B
162	Directors. Upon the completion of the initial terms of both groups of Directors,
164	succeeding Directors shall thereafter be elected for terms of three years, consistent
165	with NAESB Bylaws.

166		
167	7.2 Qu	alifications of Directors
168	_	
169	То	be eligible to serve as representative on the NAESB Board of Directors:
170		- The person must be a Member of the Retail Gas Quadrant;
171		- The person must be an executive of the organization represented and have broad
172		understanding of the natural gas industry;
173 174		 The person must have a working knowledge of the NAESB process; The person must be willing to commit the time and resources necessary to fulfill their
174		- The person must be willing to commit the time and resources necessary to fulfill their obligations as a NAESB Director and to meet the minimum threshold of participation and
176		attendance established in the NAESB Bylaws [Section 9.7 (f)];
177		- The person must disclose their interest, or their employer's interest, in the natural gas industry
178		and the relationship with other entities with which the employer may be affiliated; and
179	Th	e individual serving on the NAESB Board shall not simultaneously serve on the
180		tail Gas Quadrant EC. [DISCUSSION POINT]
181		
182	7.3 Nu	umber and Election of Directors
183		
184	a.	The Retail Gas Quadrant of NAESB shall be represented on the NAESB Board of
185		Directors by sixteen (16) persons who shall be elected, from time to time, as
186		required by Article 7 of the NAESB Bylaws and in this Exhibit to those Bylaws.
187		
188	b.	In preparation for an election of NAESB Directors representing the Retail Gas
189		Quadrant, a slate of candidates shall be solicited and developed from interested
190		Quadrant Members. The NAESB office shall administer the elections, and
191		individual Segments shall specify a point of contact that shall provide the list of
192		nominees.
193		
194	с.	The NAESB office shall coordinate the election process for Retail Gas Quadrant
195		representatives to the NAESB Board during the second week of November of
196		each year. The elections, within the Segment, for Retail Gas Quadrant
197		representatives to the NAESB Board shall be subject to the following provisions:
198		- Any Member of the Retail Gas Quadrant who is current in the payment of its dues is eligible
199		to vote.
200		- Retail Gas Quadrant Members shall cast votes for candidates in a number not exceeding the
201		number of vacant directorships.
202 203		- No Quadrant Member shall cast more than one vote per vacant directorship.
203		- The candidates receiving the greatest numbers of votes shall be elected.
205	74 Te	rm of Office
205	/	
207	Af	ter the initial Directors terms are completed, subsequent elected Directors from the
208		tail Gas Quadrant shall serve a term of three (3) years on the NAESB Board.
209		
210		
211	7.5 Va	cancies
212		
213	A	NAESB Director shall cease to be a Director upon 1) the resignation of the Retail
214	Ga	s Quadrant Membership of the entity of which the Director is a partner, director,
215	off	ficer, or agent; 2) the lapse of the Quadrant Membership of the entity of which the

216	Director is a partner, director, officer, agent or employee, through delinquent NAESB
217	dues for the Quadrant Membership; or (3) the Director's resignation, removal, or
218	death. A Director vacancy shall be filled for the remainder of that term in accordance
219	with the procedures specified in Section 4.1.
220	
221 222	7.6 Removal of Directors
223	Each Segment shall have the authority to remove a Director for cause. Prior to voting
224	on such resolution to remove a Director for cause, the Segment shall give the Director
225	at least 30-day notice of the proposed action and an opportunity to respond. A simple
226	majority of the Segment Membership shall be required to remove a Director.
227	
228	7.7 Resignations
229	
230	A Director may resign his or her directorship by submitting a letter to the Secretary of
231	NAESB, stating that he or she is resigning and giving the effective date of the
232	resignation.
233	
234	7.8 Board Committees – (Reserved)
235 236	7.9 Advisory Council – (Reserved)
230	7.9 Advisory Council – (Reserved)
238	
239	Section 8.0 – Officers – (Reserved)
240	
241	
242	Section 9.0 – Meetings of Directors – (Reserved)
243 244	
244	Section 10.0 Executive Committee
246	
247	10.1 Duties and Responsibilities
248	
249	The Retail Gas Quadrant shall have an Executive Committee, consisting of
250	representatives from each Segment, with each Segment electing an equal number of
251	these EC Members, pursuant to procedures established in Section 5. The initial EC
252	Members of the Retail Gas Quadrant shall be divided into two groups within each
253	Segment. Group A EC Members shall serve for an initial term of three years, and
254	Group B EC Members shall serve for an initial term of two years. Each Segment of
255	the Retail Gas Quadrant shall have an equal number of Group A EC Members and an
256	equal number of Group B EC Members. Upon the completion of the initial terms of
257	both groups of EC Members, succeeding EC Members shall thereafter be elected for
258	terms of three years, consistent with NAESB Bylaws. No substitutes are permitted to
259	vote at Retail Gas Quadrant meetings; however, a Designated Alternate may vote at
260	EC meetings in place of an absent EC Member from a given Segment. A Designated
200	
261	
261 262	Alternate holds identical voting rights as the EC Member in whose place he or she
262	Alternate holds identical voting rights as the EC Member in whose place he or she serves, except for those matters on which the EC Member has already voted by
262 263	Alternate holds identical voting rights as the EC Member in whose place he or she
262	Alternate holds identical voting rights as the EC Member in whose place he or she serves, except for those matters on which the EC Member has already voted by

266		
267	a.	The Retail Gas Quadrant of NAESB shall be represented on the Quadrant's EC by
268		sixteen (16) persons who shall be elected, from time to time, as required by
269		Article 10 of the NAESB Bylaws and in this Exhibit to those Bylaws. After the
270		initial terms of EC Members are completed, Retail Gas Quadrant EC Members
271		shall serve a term of three (3) years. To be eligible to serve as a representative on
272		the Retail Gas Quadrant EC:
272	-	The person must have broad understanding of the natural gas industry and be at least a senior
273	-	functional manager of the organization represented;
275	-	The person must have a working knowledge of the NAESB process;
276	-	The person must be willing to commit the time and resources necessary to fulfill their obligations
277		as a Retail Gas Quadrant EC Member and to meet the minimum threshold of participation and
278		attendance established in the NAESB Bylaws [Section 9.7 (f)];
279	-	The person must disclose their interest, or their employer's interest, in the natural gas industry and
280		the relationship with other entities with which the employer may be affiliated; and
281	-	The individual serving on the Retail Gas Quadrant EC shall not simultaneously serve on the
282		NAESB Board. [DISCUSSION POINT]
283		
284	b.	In preparation for an election of Retail Gas Quadrant EC Members, a slate of
285		candidates shall be solicited and developed from interested Quadrant Members.
286		The NAESB office shall administer the elections, and individual Segments shall
287		specify a point of contact that shall provide the list of nominees.
287		speeny a point of contact that shall provide the list of holdinees.
280	c.	The NAESB office shall coordinate the election process for Retail Gas Quadrant
	ι.	1
290		EC representatives during the second week of November of each year. The
291		elections for representatives to the Retail Gas Quadrant EC shall be subject to the
292		following provisions:
293	-	Any Member of Retail Gas Quadrant who is current in the payment of its dues is eligible to vote.
294	-	Retail Gas Quadrant Members shall cast votes for candidates in a number not exceeding the
295		number of vacant EC seats.
296	-	No Quadrant Member shall cast more than one vote per vacant EC seat.
297	-	The candidates receiving the greatest numbers of votes shall be elected.
298	1	
299	d.	A Retail Gas Quadrant EC Member may resign from the EC by submitting a letter
300		to the Secretary of NAESB, stating that he or she is resigning and giving the
301		effective date of the resignation.
302		
303	e.	A Retail Gas Quadrant EC Member shall cease to be an EC Member upon 1) the
304		resignation of the Retail Gas Quadrant Membership of the entity of which the EC
305		Member is a partner, director, officer, or agent; 2) the lapse of the Quadrant
306		Membership of the entity of which the EC representative is a partner, director,
307		officer, agent or employee, through a delinquency in payment of NAESB dues for
308		the Quadrant Membership; or (3) the EC Member's resignation, removal, or death.
309		An EC Member vacancy shall be filled for the remainder of that term in
310		accordance with the procedures specified in Section 5.1.
311		
312	f.	Each Segment shall have the authority to remove an EC Member for cause. Prior
313		to voting on such resolution to remove an EC Member for cause, the Segment
314		shall give the EC Member at least 30-day notice of the proposed action and the
		6 2-e intenter attense e e ang nouee of the proposed action and the

315	opportunity to respond. A simple majority of the Segment Membership shall be
316	required to remove an EC Member.
317	
318	10.3 EC Organization – (Reserved)
319	
320	10.4 Meetings – (Reserved)
321	10.5 EC Subcommittees
322	10.5 EC Subcommutees
323	While there may be Subcommittees and Teely Econog established by the NAESD EC
324	While there may be Subcommittees and Task Forces established by the NAESB EC,
325	to be comprised of NAESB Members and other interested parties, the Retail Gas
326	Quadrant shall set up its own Subcommittees and Task Forces to deal with Retail Gas
327	Quadrant-specific issues. Each EC Subcommittee shall report to the EC and each
328	shall:
329 330	a short a shoir who shall be an EC Member and who shall serve until removed by the
331	a. elect a chair who shall be an EC Member and who shall serve until removed by the Subcommittee's membership;
332	b. carry out its work in accordance with procedures adopted by the NAESB EC for EC
333	Subcommittees;
334	c. provide notice of meetings and agendas;
335	d. practice balanced voting and record voting results; and
336	e. keep regular minutes of its proceedings and provide copies of these minutes promptly to the
337 338	NAESB office.
339	Any task force established by EC Subcommittees shall be open to all NAESB
340	Members and other interested parties.
340 341	Members and other interested parties.
342	
343	Section 11.0 Notice – (Reserved)
344	
345	
346 347	Section 12.0 Conflicts of Interest – (Reserved)
347	
349	Section 13.0 Limitation of Liability; Insurance – (Reserved)
350	
351	
352	Section 14.0 Indemnification – (Reserved)
353 354	
355	Section 15.0 Annual Report – (Reserved)
356	Section 15.0 Annual Report – (Reserved)
357	
358	Section 16.0 Transaction of Business – (Reserved)
359	
360	
361 362	Section 17.0 Corporate Records – (Reserved)
362 363	
364	Section 18.0 Amendments
365	
366	In order for Retail Gas Quadrant Procedures to be amended, upon petition of at least
367	five (5) Members of the Quadrant, the Vice Chair of the EC for the Quadrant shall
507	Work namer 2 Provided by Wisconsin Public Service
	Work nanger / Provided by Wisconsin Public Service

368	an	nounce a Quadrant meeting. Such announcement shall provide for at least 30-days
369	no	tice. In order to transact business at the Quadrant meeting, there shall be a quorum
370		nsisting of at least 1/3 of the Quadrant Membership. Following such meeting, the
371		oposed resolution adopted at the meeting shall be sent out for comment, and the
372	-	omments shall be distributed to Quadrant members in advance of a notational vote.
373		ny Member not choosing to vote shall be considered to have voted in favor of the
374	-	oposed change. In order for a proposed change to take effect, it must be approved
375	-	v at least 2/3 of Retail Gas Quadrant Members and 40% of each Segment's
376	Μ	embership.
377		
378	a	
379	Section	n 19.0 Transition Procedures – (Reserved)
380		
381 382	Section	n 20.0 Designated Alternates
383	Section	1 20.0 Designated Atternates
384	20.1 Se	election of Designated Alternates:
385	20.1 50	
386	a.	The Distributors Segment will select not more than ten (10) Designated Alternates
387		to attend and vote at meetings of the Executive Committee when one or more
388		regular members of the Executive Committee are unable to attend a meeting of
		the Executive Committee.
389		me Executive Committee.
390	1	
391	b.	In preparation for any election of Designated Alternates for the EC, other than the initial
392 393		Designated Alternates for the EC, representing the Distributor Segment, a Nominating Committee, which chall be comprised of five (5) Segment Members, shall select a cleat of condidetes from the
393 394		which shall be comprised of five (5) Segment Members, shall select a slate of candidates from the Segment Membership. Other nominations may be made at or before the Distributor Segment
395		meeting at which the election shall occur. Segment Members shall cast votes for candidates in a
396		number not exceeding the number of vacant positions for Designated Alternate for the EC. The
397		candidates receiving the greatest numbers of votes shall be elected Designated Alternates.
398		
399	с.	All persons selected by the Nominating Committee, or whose nomination is made in any other
400		manner, shall state in writing their willingness to accept the responsibility of serving as a
401		Designated Alternate prior to the submission of their names to the Segment Membership at the
402		election.
403	1	
404 405	d.	When designated by a member of the Executive Committee to attend a meeting of the EC, a Designated Alternate shall do so unless he or she is unable to attend, in which case, the Designated
405		Alternate shall so advise the member of the EC, and the member of the EC shall select another
407		Designated Alternate. When attending a meeting of the EC, the Designated Alternate shall have
408		the full voting rights of the regular member of the EC, except as may be limited by that regular
409		member's notice in writing to the Chair of the EC, the Secretary of REQ (or the person designated
410		to serve as the secretary of the EC), and the Designated Alternate.
411		
412	e.	The Chairperson of the Distributor Segment will advise REQ's Secretary in writing of the names,
413		addresses, telephone and facsimile numbers of the Designated Alternates for the Distributor
414		Segment.
415	20 2 F	
416	20.2 Re	esignation of Designated Alternates:
417 418	٨	Designated Alternated for the Executive Committee may region his or her position by submitting a
418 419		Designated Alternated for the Executive Committee may resign his or her position by submitting a ter to the Secretary of REQ with a copy to the Chairperson of the Distributor Segment stating that he
420		she is resigning and giving the effective date of the resignation.

Work paper 2 Provided by Wisconsin Public Service 1-22-02 Joint Retail Gas and Retail Electric Working Group Conference Call – page 9 of 11

421	
422	20.3 Removal:
423	
424	A Designated Alternate may be removed upon a majority vote of the Segment Members.
425	
426	20.4 Vacancy:
427	
428	A Designated Alternate shall cease to be a Designated Alternate upon (1) the resignation of his or her
429	Segment Membership or lapse of his or her Segment Membership for delinquency in payment of the
430	membership fee (if the Designated Alternate is an individual Segment Member); (2) the resignation or
431	lapse (through a delinquency in payment of the membership fee) of the Segment Membership of the
432	entity of which the Designated Alternate is a partner, director, officer, or employee; or (3) the
433	Designated Alternate's resignation, removal, or death. A vacancy shall be filled for the remainder of
434	that term in accordance with the procedures specified in Section 6.1 above.
435	

436	
437	
438	
439	RETAIL GAS QUADRANT SUPPLIERS SEGMENT PROCEDURES
440	

OUTLINE OF NAESB BYLAWS

2	Section 1.1	Definitions
3	А.	"Act"
4	В.	"Agent"
5	C.	"Balanced Voting"
6	D.	"Board"
7	E.	"Certificate"
8	F.	"Director"
9	G.	"EC"
10	H.	"EC Subcommittee"
11	I.	"Exhibit"
12	J.	"Majority"
13	K.	"Members"
14	L.	"Model Business Practice".
15	М.	"NAESB".
16	N.	"Operating Procedures"
17	О.	"Quadrant
18	Р.	"Reconsideration"
19	. Q.	"Segment"
20	R.	"Standard"
21	S.	"Triage Process"
22	Т.	"Voting Member"

ARTICLE 2 - PURPOSES, SCOPE, ACTIVITIES AND POLICIES

24 Section 2.1 Purposes, Scope and Activities

25	Section 2.2	Policies
26		Independence –
27		Openness
28		Voluntary
29		Balance of Interests –
30		Inclusivity –
31		Consensus-Based Decisions –
32		No Advocacy
33		Membership Driven –
34		Develop Practices, Not Policy –
35		Incorporate Best Practices –
36		Broad Applicability –
37		ANSI Accreditation –

38	Section 2.3	Quadrants and Segments
39		ARTICLE 3 - OFFICES
40	Section 3.1	Offices
41		ARTICLE 4 - SEAL
42	Section 4.1	Seal
43		ARTICLE 5 - MEMBERS
44	Section 5.1	Voting Members
45	Section 5.2	Non-Voting Members
46	Section 5.3	Nontransferable
47	Section 5.4	Resignation
48		ARTICLE 6 - MEETINGS OF MEMBERS
49	Section 6.1	Place of Meetings
50	Section 6.2	Annual Meeting
51	Section 6.3	Special Meetings of Members
52	Section 6.4	Determination of Members of Record
53	Section 6.5	Notice of Meetings of Members
54	Section 6.6	Quorum
55	Section 6.7	Adjournment
56	Section 6.8	Organization
57	Section 6.9	Voting on Particular Issues
58		
59		ARTICLE 7 - BOARD
60	Section 7.1	Board
61	Section 7.2	Oualifications of Directors

- 62 Section 7.3 Number and Election of Directors
- 63 Section 7.4 Term of Office
- 64 Section 7.5 Vacancies
- 65 Section 7.6 Removal of Directors
- 66 Section 7.7 Resignations
- 67 Section 7.8 Board Committees
- 68 Section 7.9 Advisory Council
- 69

ARTICLE 8 - OFFICERS

- 70 Section 8.1 Number
- 71 Section 8.2 Qualifications of Officers of NAESB
- 72 Section 8.3 Election and Term of Office
- 73 Section 8.4 Removal of Officers
- 74 Section 8.5 Resignations
- 75 Section 8.6 The Chair
- 76 Section 8.7 The Vice Chairs
- 77 Section 8.8 The Secretary
- 78 Section 8.9 The Treasurer

- 79 Section 8.10 The Executive Director
- 80

ARTICLE 9 - MEETINGS OF DIRECTORS

- 81 Section 9.1 Place of Meetings
- 82 Section 9.2 Organization
- 83 Section 9.3 Annual Meeting
- 84 Section 9.4 Regular Meetings
- 85 Section 9.5 Special Meetings of the Board
- 86 Section 9.6 Quorum
- 87 Section 9.7 Participation and Voting in Meetings
- 88 **ARTICLE 10 EXECUTIVE COMMITTEE**
- 89 Section 10.1 Duties and Responsibilities
- 90 Section 10.2 EC Members
- 91 Section 10.3 EC Organization
- 92 Section 10.4 Meetings
- 93 Section 10.5 EC Subcommittees
- 94

ARTICLE 11 - NOTICE

- 95 Section 11.1 Written Notice
- 96 Section 11.2 Waiver by Writing
- 97 Section 11.3 Waiver by Attendance
- 98 ARTICLE 12 CONFLICTS OF INTEREST
- 99 Section 12.1 Interested Directors and Officers
- 100**ARTICLE 13 LIMITATION OF LIABILITY; INSURANCE**
- 101 Section 13.1 Limitation of Liability
- 102

- 103 Section 13.2 Insurance
- 104

ARTICLE 14 - INDEMNIFICATION

- 105 Section 14.1 Representative Defined
- 106 Section 14.2 Third-Party Actions
- 107 Section 14.3 Derivative and Corporate Actions
- 108 Section 14.4 Procedure for Effecting Indemnification
- 109 Section 14.5 Advancing Expenses
- 110 Section 14.6 Supplementary Coverage
- 111 Section 14.7 Duration and Extent of Coverage
- 112 Section 14.8 Reliance and Modification
- 113

- **ARTICLE 15 ANNUAL REPORT**
- 114 Section 15.1 Annual Report
- 115ARTICLE 16 TRANSACTION OF BUSINESS
- 116 Section 16.1 Real Property
- 117 Section 16.2 Negotiable Instruments
- ARTICLE 17 CORPORATE RECORDS
- 119 Section 17.1 Corporate Records
 - **ARTICLE 18 AMENDMENTS**
- 121 Section 18.1 Amendments
- 122 ARTICLE 19 TRANSITION PROCEDURES
- 123 Section 19.1 Officers of the Board
- 124 Section 19.2 Officers of the EC
- 125 Section 19.3 Procedures to Constitute Board & EC Representation
- 126

127	EXHIBIT 1
128	WHOLESALE GAS QUADRANT PROCEDURES
129	EXHIBIT 2
130	WHOLESALE ELECTRIC QUADRANT PROCEDURES
131	EXHIBIT 3
132	RETAIL GAS QUADRANT PROCEDURES
133	EXHIBIT 4
134	RETAIL ELECTRIC QUADRANT PROCEDURES

TAB 4

PROPOSED NAESB RETAIL ELECTRIC QUADRANT PROCEDURES

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
SECT	TION 1 - DEFINITIONS	
Α.	"Distributor Segment Member" means a person (e.g., corporation or partnership) engaged in the local distribution of electricity, or its representative (e.g., employee, officer, director, agent, or partner) who is authorized to cast that Segment Member's vote, that satisfies the requirements for Membership set forth in Article 5 of the bylaws of NAESB, and Section 2.3 of these Procedures, and further satisfies the qualifications for Distributor Segment Membership of the Retail Electric Quadrant as set forth in the Distributor Segment Procedures.	Comments during January 17 th meeting suggested replacing the definitions listed here under 1.1, 1.2, 1.8 & 1.9 with the simplified definitions of the Segments provided in Ver. 2.1, Sec 3.1 (REQ Draft Procedures) derived from the UBP Governance Document, as now shown in Section 2.4(C) of this document.

- Β. "End User Segment Member" means a person (e.g., corporation, partnership, proprietorship, or individual) engaged in the consumption of electricity delivered by an Electric Distributor, or its representative (e.g., employee, officer, director, agent, or partner) who is authorized to cast that Segment Member's vote, that satisfies the requirements for Membership set forth in Article 5 of the Bylaws of NAESB, and Section 2.3 of these Procedures, and further satisfies the gualifications for End User Segment Membership of the Retail Electric Quadrant as set forth in the End User Segment Procedures.
- C. "NAESB Office" means the office of the Secretary of NAESB
- D. "REQ" means the Retail Electric Quadrant

Ε. "Segment" means one of the co-equal Membership Segments of the NAESB Retail Electric Quadrant representing the following four (4) segments of the retail electric industry: Suppliers, Distributors, End Users, and Services.

01/24/02

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
F.	"Segment Membership" means the Segment Members collectively.	
G.	 "Services Segment Member" means a person (e.g., corporation, partnership or proprietorship or individual) that provides services to participants in the retail electric industry or an entity not otherwise declared in another segment, or its representative (e.g., employee, officer, director, agent, or partner) who is authorized to cast that Segment Member's vote, that a) satisfies the requirements for Membership set forth in Article 5 of the Bylaws of NAESB, and Section 2.3 of these Procedures, and b) further satisfies the qualifications for Services Segment Membership of the Retail Electric Quadrant as set forth in the Services Segment Procedures. 	
H.	 "Supplier Segment Member" means a person (e.g., corporation, partnership, or proprietorship) engaged in the competitive sale of electricity to end-users, or its representative (e.g., employee, officer, director, agent, or partner) who is authorized to cast that Segment Member's vote, that a) satisfies the requirements for Membership set forth in Article 5 of the Bylaws of NAESB and Section 3 of these Procedures, and b) further satisfies the qualifications for Supplier Segment Membership of the Retail Electric Quadrant as set forth in the Supplier Segment Procedures. 	
1.	All other capitalized terms, if not defined in this Section 1.0, shall have the same definitions as specified in the Bylaws or Certificate of Incorporation of NAESB.	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	J	
SECT	TION 2 – PURPOSES, SCOPE, PRINCIPLES AND POLICIES	
2.1 -	Purpose & Scope	
A.	Purpose - The purpose of the Retail Electric Quadrant of the North American Energy Standards Board (NAESB) is to propose, evaluate and adopt voluntary standards and model business practices to promote competitive, efficient and reliable service in the retail electric industry.	
В.	 Scope – The Retail Electric Quadrant is to address issues and practices that are within the scope of NAESB and appropriate to electric usage at the individual consumer level—that is, usage by an individual, partnership, corporation, or other entity consuming electricity at one or more facilities served by an Electric Distributor. Issues may also be addressed in conjunction with other Quadrants when such issues are common to both electric distribution at the individual consumer level and some combination of wholesale or retail activities in the electric or natural gas industries. 	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	 Explanatory Note from Version 2.1: The Retail Electric Quadrant would normally address issues and practices which are generally within the jurisdiction of state or local regulatory authorities. As such, the findings, recommendations and proposed standards or model business practices produced by this Quadrant of NAESB may be implemented in individual jurisdictions only to the extent that they are consistent with applicable law and regulation. Standards and model business practices that prove successful in those states that adopt them might well be adopted by additional jurisdictions if compatible with policies in their jurisdiction. Furthermore, the Retail Electric Quadrant will operate in conformance with the principles of NAESB as established in Section 2.2 (b) of the NAESB Bylaws, and in compliance with ANSI guidelines as codified therein. In keeping with these Principles, the Retail Electric Quadrant encourages membership from across the spectrum of the Retail Electric industry, reflecting the principle of inclusiveness. All Retail Electric Quadrant meetings, including those of its Members, Executive Committee (EC), subcommittees and task forces, shall be open to any interested member of the public and that all minutes thereof shall be available to the public in keeping with the principle of openness. 	This Explanatory Note would be removed from the final REQ document. In comparison, the Jan-4-02 draft RGQ document addresses these points and others with language intended to remain in the document: "The Retail Gas Quadrant shall work closely with other NAESB Quadrants to mitigate inconsistencies where proposed standards and model business practices affect those other Quadrants. "The Retail Gas Quadrant shall comply with the policies and procedures laid out in the bylaws and the certificate of incorporation of NAESB. As such, the Retail Gas Quadrant encourages a widely based membership and has identified Segments, which reflect this principle of inclusiveness. "Also consistent with this approach, the Retail Gas Quadrant shall ensure that all meetings, including those of its Members, Executive Committee (EC), and Subcommittees and Task Forces, shall be open to all persons and that all minutes hereof shall be available to the public. "Further, the Retail Gas Quadrant intends to operate in conformance with the principles of the umbrella organization, as established in Section 2.2 (b) of the NAESB Bylaws, and in compliance with ANSI guidelines."
2.2 -	Policies	
A.	Meetings Of The Retail Electric Quadrant All meetings held in association with the NAESB organization or the REQ are open to any interested person.	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
B.	Joint Meetings From time to time, there will be joint meetings of the REQ with other Quadrants within NAESB, and Segments will meet jointly to transact Quadrant business.	
C.	 One Person, One Vote Requirement 1. Designation of the same individual as the duly authorized representative of the Member in multiple Segments of the REQ (including Agent representatives) is not permitted. 	RGQ draft approaches this point differently: "At Retail Gas Quadrant Subcommittee or Task Force meetings, the same individual may represent different Segments, as long as that individual Declares prior to the meeting the Retail Gas Quadrant Segment for which he or she is casting a vote."
	2. In the event that a duly authorized representative in one Segment also represents the Member in another Segment in any other NAESB Quadrant, and both affected Segments are engaged in a joint meeting requiring Member voting, then the person serving as representative in both Segments must choose the Segment and Quadrant in which they will vote on any given issue and may only cast one vote on each voting issue. To the extent that any other involved Quadrant's rule may differ on this point, the representative must still honor this Quadrant's rule for any such vote or designate a substitute representative for the entity's interests during the joint meeting.	Note: The provisions of this subsection deal with an issue applicable to all Quadrants and may be more appropriate for inclusion in either the Bylaws or NAESB Operating Procedures. In addition, since nearly all votes by a Member in NAESB will either be for election of officers or ratification, joint action with another Quadrant requiring Member voting should be very rare, and may never actually occur. Need to decide whether to keep, modify or drop this. Jan-4-02 version of RGQ draft approaches this point differently in two different places: "At NAESB joint Subcommittee or joint Task Force meetings, the same individual may represent different Segments from different Quadrants, as long as that individual declares prior to the joint meeting the Quadrant and Segment for which he or she is casting a vote." and "However, at the EC and Board level, an elected representative may hold not more than one seat, representing only one Segment within one Quadrant."

Draft V2.2 Retail Electric Quadrant Procedures -	Comment or Comparable Point in Gas Quadrant Draft
Restructured to Follow NAESB Bylaws	

D.	Ratification Only REQ Voting Members have the right to ratify standards or model business practices adopted by the Quadrant EC. Ratification voting will be in accordance with the requirements of the NAESB Bylaws.	RGQ Procedures express this differently: "Only Retail Gas Quadrant EC Members shall have the right to vote to adopt model business practices or standards affecting only the Retail Gas Quadrant."
Sect	ion 2.3 –Segment Organization & Membership Requirement	S
A.	 Membership Requirements The definition of Membership applies to all Segments within the REQ. Members of the REQ are persons who meet the following membership requirements in any one of the four Segments of this Quadrant. Upon applying for Voting Membership (as described in NAESB Bylaws Sections 1.1 and 5.1) in the REQ, each prospective Member shall declare the Segment with which they are to be identified. Distributor Segment Parties eligible for Membership in this Segment are engaged in the local distribution of electricity. End User Segment Parties eligible for Membership in this Segment consume electricity or represent consumers of electricity. Suppliers Segment Parties eligible for Membership in this Segment are service providers and marketers, engaged in the competitive sale of electricity to end-users Services Segment Parties eligible for Membership in this Segment provide services to participants in the retail electric industry, including equipment manufacturers, equipment vendors, software providers, consultants, and other companies or individuals not otherwise eligible for Membership in another 	The draft RGQ Procedures have similar requirements, and add other provisions that generally correspond to Bylaws Section 2.3: "Each Segment within the Retail Gas Quadrant shall have the flexibility to determine its own Segment rules and procedures, as long as they conform with the NAESB Bylaws and the Retail Gas Quadrant procedures, and to elect, or select, its own representatives to the NAESB Board and to the Retail Gas Quadrant EC." REQ should consider whether similar provisions need to be in the REQ Procedures in this section. A comment made during the Jan 17 th meeting suggested that the word "local" should be dropped from the description of Distributor Segment Membership. Agree / Disagree?

01/24/02

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	Segment.	
SEC	TIONS 3 & 4 are reserved for future use	
SEC.	TION 5 - MEMBERS	
Sect	tion 5.1 –Members	
Α.	Member Representation Representatives chosen by any person in any Segment should have, but are not required to have, corporate responsibility for the business interests causing the person to seek Membership in the Segment	Is this a satisfactory statement, or should it be more specific?
В.	 Removal of Members Members who cease to have a significant business interest pertaining to the descriptions contained in Section 2.3 may be removed from Membership by either 1. a simple majority vote of their declared Segment Membership, or 2. resolution of the REQ Executive Committee 	Comments at Jan 17 th meeting suggested change of wording to "Members <u>who do not have a legitimate</u> business interest "
C.	 Multiple Memberships Memberships in multiple Segments of the REQ are permissible for any person provided each Membership is filed and declared with NAESB, the person meets the Membership requirements of each Segment joined, and Membership dues are paid for each Segment. 1. Only one Membership per Segment is permissible for any person. 2. Multiple companies under common control within a corporate organization that desire to become Members 	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	must join individually. Members cannot extend their Membership to their parent company, affiliates, or subsidiaries.	
Sect	ion 5.2 – Voting Members	
A.	 Voting Membership Membership and voting rights in the REQ of NAESB shall be open to any Member that meets the following requirements: 1. The Member has a significant business interest¹ in the retail electric market (or is a representative or Agent of such person), as determined by a simple majority of the segment Membership. 2. The Member may be a trade association or an advocacy group representing a group of prospective members, provided that a) the group represents an interest that is identified within a Segment, yet is (i) not represented, or (ii) is underrepresented in the judgement of a simple majority of the Segment Membership, and b) the prospective representative's relationship to the group wishing to be represented in this manner is documented in one of the following ways: (i) has statutorily or regulatory designated authority to represent the interests of the subject group (ii) is generally recognized representative of the subject group for other transactions (e.g. provides group insurance, group buying of another product, etc.) (iii) provides a letter of authorization from the group, 	Regarding 5.2(A)(1), comment at Jan 17 th meeting proposed replacing "significant" with "legitimate". Regarding 5.2(A)(2), ELCON requested that these provisions be deleted in their entirety. Jan 4 th RGQ draft addresses Trade Associations differently: "Through special application, which includes a recommendation by its Segment and an affirmative vote of % of the Quadrant EC, a trade association may join the Retail Gas Quadrant Membership if it represents an interest that is identified within a Segment yet is not represented, or is underrepresented in the Quadrant Membership. The membership status of such trade associations shall be considered on an annual basis and may be voted upon through notational vote by the Quadrant EC."

¹ as used in this section, "significant business interests" specifically includes the interests of statutorily appointed consumer advocates

01/24/02

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft	
	authorization c) no more than two entities within the group represented subsequently join as full Members of the Segment, at which point the representative's authorization for representation and voting rights will be immediately voided unless reinstituted by letter for the remainder of the operating year by the remaining members of the representative's constituency.		
SEC	TION 6 – Reserved for future use		
SECTION 7 - BOARD Section 7.1 - Board A. Board Representation The REQ shall elect Directors to the Board. Draft RGQ Procedures go further, requiring: "each Segment electing or selecting an equal number of those			
		representatives, pursuant to the procedures specified by that Segment."	
Sect	ion 7.2 – Qualification of Directors		
Α.	No two Directors elected by the REQ may be employees of the same Member holding Membership in multiple Segments within the Quadrant. This restriction does not prohibit election of two Directors from two affiliated companies within a holding company which each have individual Member status, or from two companies with a parent-subsidiary relationship, provided that the two Directors from companies with such a relationship represent Members of differing Segments.	Draft RGQ requirements are: "To be eligible to serve as representative on the NAESB Board of Directors: - The person must be a Member of the Retail Gas Quadrant; - The person must be an executive of the organization represented and have broad understanding of the natural gas industry; - The person must have a working knowledge of the NAESB process; - The person must be willing to commit the time and	
В.	Directors elected from the REQ may not hold both a Board seat and a seat on the EC at any point in time. If an EC Member is	resources necessary to fulfill their obligations as a NAESB Director and to meet the minimum threshold of participation	
	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft	
------	---	---	
	elected as a Director from the REQ, their EC seat is vacated	and attendance established in the NAESB Bylaws [Section 9.7]	
	when the Board seats them as a Director.	(f)];	
	(Note: No other requirements have been proposed regarding qualifications to be a Director; Does the Quadrant membership wish to add provisions similar to those proposed by the RGQ?)	 The person must disclose their interest, or their employer's interest, in the natural gas industry and the relationship with other entities with which the employer may be affiliated. The individual serving on the NAESB Board shall not simultaneously serve on the Retail Gas Quadrant EC." 	
Sect	ion 7.3 – Number and Election of Directors		
A.	Number of Directors The REQ shall elect sixteen (16) NAESB Directors, subject to the provisions of Section 19 of these Procedures. Each Segment of the Quadrant will elect four (4) Directors, subject to the provisions of Section 19 of these Procedures	<i>Note: RGQ is considering having 24 Directors. Having differing numbers of Directors from each Quadrant is not a violation of the Bylaws.</i>	
В.	 Election of Directors Nominations for and election of all Directors will be in accordance with Segment Procedures, or in the absence of Segment Procedures, as follows: 1. In preparation for any election of NAESB Directors (other than initial Directors, as provided for in Section 19), a) a nominating committee consisting of five EC Members of the REQ comprised of one Member from each Segment plus the EC Vice-Chair, shall develop a slate of candidates from the Segment Membership. b) other nominations may be made at or prior to the close of the REQ End User Segment nomination period by any Segment Member eligible to serve on the Board of Directors by submitting their name(s) to the NAESB Office in a form as specified in NAESB Operating Procedures if such requirements exist. c) all nominations must be made and conveyed in writing to 	 Note: Jan-4-02 draft of RGQ procedures propose to have the NAESB Office administer elections, and mechanism for developing a slate of candidates is not specified. Other RGQ provisions which differ from those shown in this draft REQ document are: "The elections, within the Segment, for Retail Gas Quadrant Representatives to the NAESB Board shall be subject to the following provisions: Any Member of the Retail Gas Quadrant who is current in the payment of its dues is eligible to vote. Retail Gas Quadrant Members shall cast votes for candidates in a number not exceeding the number of vacant directorships. No Quadrant Member shall cast more than one vote per vacant directorship. The candidates receiving the greatest numbers of votes 	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	the NAESB Secretary no less than 31 days prior to the election date.	shall be elected."
C.	Timing of Elections Subject to the provisions of Section 19, election of Directors shall occur in the same month for all Segments of the REQ, and shall be coordinated by the Executive Committee. The EC will ensure that the requirements of this Section are satisfied by each Segment's slate of candidates prior to any actual voting by any Segment.	<i>Note: This may be an issue that should be addressed in NAESBops rather than in Quadrant documents</i>
Secti	on 7.4 – Term of Office	
Α.	 Term of Office Directors shall be elected for two-year terms, with half of the terms expiring in alternating years, subject to the provisions of Section 19. 1. Two Directors will be elected from each Segment each year to fill expiring terms. 2. Group A Directors will have their terms expire in odd numbered years; 3. Group B Director terms will expire in even numbered years;' 4. Term expiration will be in conjunction with the end of the operating year of NAESB or as otherwise defined by the Board, Certificate of Incorporation or Bylaws, as amended. 	RGQ is proposing a 3 year term of office
В.	Limit on Number of Terms of Office Directors elected from the REQ may run for re-election without restriction on the number of terms held.	

Draft V2.2 Retail Electric Quadrant Procedures -	Comment or Comparable Point in Gas Quadrant Draft
Restructured to Follow NAESB Bylaws	

C.	 Change of Affiliation In the event that the Director changes affiliation 1. to another Member within his/her industry Segment, the Director's term will continue until its natural expiration, provided that there is no other Director already representing the same Member, in which case the Director changing his or her affiliation will vacate their seat for election of a new Director; 2. and is no longer affiliated with their electing industry Segment, the Director will vacate their seat for election of a new Director. 	<i>Note: Per discussion on Jan 17th, these provisions are different from past practice of GISB, but not in conflict with Bylaws</i>
Secti	on 7.5 - Vacancies	
А.	In the event that a Director resigns or otherwise vacates their Board seat, the Quadrant will hold an election within 60 days to fill the vacant seat with a candidate from the same Segment as the original Director.	Jan-4-02 draft of RGQ Procedures consider other conditions which create a vacancy and which should be addressed in the REQ Procedures: "A NAESB Director shall cease to be a Director upon 1) the resignation of the Retail Gas Quadrant Membership of the entity of which the Director is a partner, director, officer, agent; 2) the lapse of the Quadrant Membership of the entity of which the Director is a partner, director, officer, agent or employee, through delinquent NAESB dues for the Quadrant Membership; or (3) the Director's resignation, removal, or death."
Secti	on 7.6 – Removal of Directors	
A.	In addition to being subject to removal from office by the NAESB Board of Directors, Directors may also be removed from office by the Quadrant Membership at any time upon a 75%	<i>RGQ approach: "Each Segment shall have the authority to remove a Director for cause. Prior to voting on such resolution to remove a</i>

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	majority vote of the Quadrant Membership and a simple majority (51%) of each of the Segments of the REQ. The vacant seat is to be refilled in accordance with the requirements of Section 7.5.	Director for cause, the Segment shall give the Director at least 30-day notice of the proposed action and an opportunity to respond. A simple majority of the Segment Membership shall be required to remove a Director."
SEC	TIONS 8 & 9 - Reserved for future use	
	TION 10 – EXECUTIVE COMMITTEE tion 10.1 – Duties and Responsibilities Within Quadrant EC	
Α.	 The REQ EC shall 1. elevate the prior year Vice Chair of the Quadrant EC to Chair of the REQ EC at its first meeting in the new operating year and elect a new Vice Chair. If the Vice Chair is vacant at the time of the first meeting of a new operating year, both a Chair and Vice-Chair will be elected, and 2. designate an EC Secretary responsible for production of meeting minutes per NAESB Operating Procedures. This may involve an individual assigned this responsibility on a permanent basis, or an agreement on how to designate a person or persons responsible for recording minutes of each meeting. 	A Designated Alternate holds identical voting rights as the EC Member in whose place he or she serves, except for those matters on which the EC Member has already voted by notational ballot prior to the beginning of the meeting.
Sect	tion 10.2 – Qualifications of EC Members	
A.	EC Members elected from the REQ may not hold both a Board seat and a seat on the Executive Committee at any point in time. If an EC Member is elected to the Board from the REQ, their EC seat is vacated when the Board seats them as a Director.	Jan-4-02 draft of RGQ Procedures provide additional requirements regarding qualification to serve on the EC, and these requirements may need to be addressed in the REQ Procedures: "To be eligible to serve as a representative on the Retail Gas Quadrant EC:

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
В.	Note: Should the above be amended to "EC Member representatives"? As written, can be interpreted that no Member may hold both an EC seat and a Board seat. No two EC Members elected by the REQ may be employees of the same Member holding Membership in multiple Segments within the Quadrant. This restriction does not prohibit election of two EC Members from two affiliated companies within a holding company which each have individual Member status, or from two companies with a parent-subsidiary relationship, provided that the two EC Members from companies with such a relationship represent Members of differing Segments.	 The person must have broad understanding of the natural gas industry and be at least a senior functional manager of the organization represented; The person must have a working knowledge of the NAESB process; The person must disclose their interest, or their employer's interest, in the natural gas industry and the relationship with other entities with which the employer may be affiliated; The person must be willing to commit the time and resources necessary to fulfill their obligations as a the Retail Gas Quadrant EC Member and to meet the minimum threshold of participation and attendance established in the NAESB Bylaws [Section 9.7 (f)]"
Secti	ion 10.3 – Number and Election of EC Members	
Α.	Number of EC Members The REQ shall elect sixteen (16) EC Members, subject to Section 19 of these Procedures, with each Segment electing four (4) of these EC Members, pursuant to procedures established by that Segment. All EC representation from the REQ will be comprised of an equal number of representatives elected from each Segment of the Quadrant.	Note: RGQ is considering having up 24 EC Members, per comments at Jan 17 th meeting
В.	 Election of EC Members Nominations for and election to an EC seat will be in accordance with Segment Procedures, and the following requirements 1. In preparation for any election of EC Members (other than initial EC Members, as provided for in Section 19, a) a nominating committee consisting of five EC Members of the REQ comprised of one Member from each Segment plus the EC Vice-Chair, shall develop a slate of candidates from the Segment Membership; 	Note: Jan-4-02 draft of RGQ procedures propose to have the NAESB Office administer elections, and mechanism for developing a slate of candidates is not specified. Other RGQ provisions which differ from those shown in this draft REQ document are: "The elections, within the Segment, for Retail Gas Quadrant Representatives to the NAESB Board shall be subject to the following provisions: - Any Member of the Retail Gas Quadrant who is current in the payment of its dues is eligible to vote.

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	 b) other nominations may be made at or prior to the close of the REQ End User Segment nomination period by any Segment Member eligible to serve on the EC by submitting their name(s) to the NAESB Office in a form as specified in NAESB Operating Procedures if such requirements exist; c) all nominations must be made and conveyed in writing to the NAESB Secretary no less than 31 days prior to the election date. 	 Retail Gas Quadrant Members shall cast votes for candidates in a number not exceeding the number of vacant EC seats. No Quadrant Member shall cast more than one vote per vacant EC seat. The candidates receiving the greatest numbers of votes shall be elected."
C.	Timing of Elections Subject to the provisions of Section 19, election of EC Members shall occur in the same month for all Segments of the REQ, and shall be coordinated by the existing REQ Executive Committee. The EC will ensure that the requirements of this Section are satisfied by each Segment's slate of candidates prior to any actual voting by any Segment.	Note: This may be an issue that should be addressed in NAESBops rather than in Quadrant documents RGQ draft procedures specify the timing of EC elections as the second week of November
D.	 Term of Office EC Members shall be elected for two-year terms, with half of the terms expiring in alternating years. 1. Subject to the provisions of Section 19, two EC Members will be elected from each Segment each year to fill expiring terms. 2. Group A EC Members will have their terms expire in odd numbered years, and 3. Group B EC Member terms will expire in even number years, 4. Terms of office will end in conjunction with end of the operating year of NAESB or as otherwise defined by the Board of Directors, Certificate of Incorporation or Bylaws, as amended 	Jan-4-02 draft RGQ Procedures incorporates a 3 year term of office
E.	Limit on Number of Terms of Office EC Members from the REQ may run for re-election without	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
F.	 restriction on the number of terms held. Change of Affiliation In the event that the EC Member changes affiliation to another Member within his/her industry Segment, the EC Member's term will continue until its natural expiration, provided that there is no other EC Member already representing the EC Member's new affiliation, in which case the EC Member changing affiliation will resign their seat for election of a new EC Member; is no longer affiliated with their electing industry Segment, the EC Member will vacate their seat and a new EC Member will be elected to fill the seat. 	
Secti	 on 10.4 - EC Meetings EC Meetings 1. REQ EC meetings shall be held at times and locations determined by the Chair or Vice-Chair of the REQ EC. EC Members may participate and vote by means of tele-conference or other electronic means unless in-person attendance is required of all EC Members by both the Chair and Vice-Chair of the EC, and subject to the attendance requirements of Article 10, Section 10.4(j) of the Bylaws. 2. Attendance records shall be reviewed by the person acting as Secretary of the EC during the course of each Quadrant EC meeting for compliance with NAESB Bylaws for the term of office served by each current Quadrant EC Member. The Secretary shall advise any EC Members of their status regarding risk for failure to comply with the Bylaws participation requirement. 	Note: These provisions may be better positioned in NAESBops. These subsections within Section 10.4 have been edited for greater compliance with the Bylaws.

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
В.	 Joint EC Meetings 1. In the event that the EC of the REQ meets jointly with an EC of another NAESB Quadrant, the choice of Quadrant EC Chair presiding over the joint meeting will be determined by the precedence established in the order of rotation of Vice-Chairs as specified in the NAESB Bylaws. 2. In the event that the REQ EC Chair or Vice-Chair is presiding over a joint meeting, recording and preparing meeting minutes will be in accordance with normal procedures adopted by the REQ EC for all REQ EC meetings. 	Note: These provisions may be better positioned in NAESBops. These subsections have been edited for greater compliance with the Bylaws.
Secti	on 10.5 – EC Subcommittees	
A.	 Establishing Subcommittees & Task Forces The EC of the REQ shall set up its own subcommittees and task forces to deal with REQ-specific issues. 1. The Executive Committee may establish voluntary standing subcommittees or special purpose task forces to perform various functions required of the organization. a. The Executive Committee will prepare a written statement of the purpose of the subcommittee or task force and the tasks to be performed, name the subcommittee or task force, state if balanced voting among the Segments is applicable, and appoint a temporary chair. b. The Temporary Chair may be a Member of the Executive Committee or any REQ Member willing to perform the required startup tasks and to continue chairing the subcommittee / task force if elected by the Members its first meeting. c. The Temporary Chair shall set up the first meeting of the subcommittee or task force and prepare a meeting 	Note: These provisions may be better positioned in NAESBops. The RGQ draft document approaches this section differently: "While there may be Subcommittees and Task Forces established by the NAESB EC, to be comprised of NAESB Members and other interested parties, the Retail Gas Quadrant shall set up its own Subcommittees and Task Forces to deal with Retail Gas Quadrant-specific issues. "Each EC Subcommittee shall report to the EC and each shall: a. elect a chair, who shall be an EC Member and who shall serve until removed by the Subcommittee's membership; b. carry out its work in accordance with procedures adopted by the NAESB EC for EC Subcommittees; c. provide notice of meetings and agendas; d. practice balanced voting and record voting results; e. and keep regular minutes of its proceedings and provide

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	 notice that: i) states the name and purpose of the subcommittee / task force, ii) solicits participation in the subcommittee / task force, iii) announces the agenda for the first meeting. and shall send the meeting notice to all Members and non-members who have participated in Quadrant activities in the last twelve months. The notice shall be sent at least two weeks prior to the meeting date. Shorter time periods for notices of subsequent meetings will be permitted by a 75% vote of the participants attending a duly scheduled meeting. All meeting notices shall be posted on the NAESB website and transmitted in writing, facsimile, or other electronic means to parties who have indicated an interest in the duly scheduled meeting. The permanent subcommittee / task force Chair may be any REQ Member. Participation on subcommittees or task forces is open to any interested party. 	copies of these minutes promptly to the NAESB office. "Any task force established by EC Subcommittees shall be open to all NAESB Members and other interested parties."
В.	Meeting Minutes Each EC subcommittee or task force will designate a meeting Secretary responsible for recording and reporting voting results, keeping regular minutes of its proceedings and providing copies of those minutes promptly to the NAESB Executive Director's office in accordance to reporting criteria established by the EC.	
C.	Reporting Each EC subcommittee or task force will report to the EC at no less than a quarterly basis, on a schedule to be defined by the EC for as long as the subcommittee or task force continues to exist.	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
D.	Balanced Representation Each EC subcommittee / task force will strive for balanced representation and balanced voting between Segments, and participation shall be open to all NAESB Members and other interested parties, without regard to whether the participant is a paid Member of NAESB in good standing.	
Sect	ion 10.6 - Removal of EC Members	
Α.	 Nonparticipation 1. Failure to meet the Bylaws in-person attendance requirements [Article 10, Section 10.4(j)] for reasons other than hospitalization or temporary disability requires a) issuing a notice of non-compliance to the Quadrant Membership and b) a vote for removal of the Executive Committee Member from their seat within 30 days of failing to comply. c) If the REQ Membership does not vote for removal of the EC Member, the Secretary or EC Chair may call for another removal vote by the Quadrant Membership following the next EC review of participation records if the EC Member's attendance record still does not satisfy the Bylaws requirements. 	The Jan-4-02 RGQ draft procedures consider the following causes for removal of an EC Member: "A Retail Gas Quadrant EC Member shall cease to be an EC Member upon 1) the resignation of the Retail Gas Quadrant Membership of the entity of which the EC Member is a partner, director, officer, or agent; 2) the lapse of the Quadrant Membership of the entity of which the EC representative is a partner, director, officer, agent or employee, through a delinquency in payment of NAESB dues for the Quadrant Membership; or (3) the EC Member's resignation, removal, or death."
В.	 Resignation or Ineligibility An EC Member shall cease to be a Member of the EC upon 1. The resignation of REQ End User Segment Membership by the Member whom they represent; 2. The lapse of the Segment Membership dues of the Member whom they represent; 3. The EC Member's resignation, removal, or death; 4. The first day of the NAESB operating year following an election for the seat held by the EC Member, wherein the EC Member was not re-elected by the Segment Membership as 	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
C.	 declared by the NAESB Secretary; 5. Receipt by the NAESB Secretary of a rescission letter indicating that the individual no longer represents the Segment Member. 6. Election of the EC Member to a Board seat, and seating of the EC Member as a Director. Other Cause EC Members may be removed from office by the Quadrant Membership at any time upon a 75% majority vote of the Quadrant Membership and a simple majority (51%) of each of the Segments of the REQ. The vacant seat is to be refilled in accordance with Section 10.7. 	Jan-4-02 draft RGQ provisions address the topic covered in part C of the REQ subsection differently: "Each Segment shall have the authority to remove an EC Member for cause. Prior to voting on such resolution to remove an EC Member for cause, the Segment shall give the EC Member at least 30-day notice of the proposed action and the opportunity to respond. A simple majority of the Segment Membership shall be required to remove an EC Member."
Secti	ion 10.7 - Vacancies	
	 In the event that an EC Member resigns or otherwise vacates their seat, the Quadrant will hold an election within 60 days to fill the vacant seat with a candidate from the same Segment as the original EC Member. 1. A vacancy resulting from removal of an EC Member shall be filled for the remainder of that term in accordance with the following requirements a. Any EC seats becoming vacant shall be filled by a special election subject to the following provisions b. If the vacancy occurs with less than 125 days remaining in the EC Member's term of office, the seat will remain vacant until filled by election of a new EC Member to a term commencing in the next operating year, and a Designated Alternate will serve until expiration of the term. 	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	 c. All other vacancies will be filled by special election; such special elections are to be formally noticed to the Segment Membership by the Segment Vice-Chair or Chair within 15 days of when the creation of the vacancy becomes known and held within 60 days of the creation of the vacancy. Candidates will be designated by the normal nomination process or selected by the remaining EC Members of the Segment from Segment Membership. Candidates elected to fill the vacancy will serve the remainder of the term, and may stand for re-election. d. Until an election is held and a new EC Member is elected, a Designated Alternate will serve in the open EC seat. 	
Secti	on 10.8 - Voting	
Α.	Only Members of the REQ EC shall have the right to vote to adopt standards or model business practices for the REQ. Quadrant voting procedures will be in accordance with NAESB Bylaws.	
Secti	on 10.9 - Designated Alternates	
Α.	 Any person presenting themselves at an EC meeting as a Designated Alternate will be accepted as a participant provided that 1. An EC Member from that Segment either indicates to the NAESB Office, EC Chair or Vice-Chair that they will be absent, or is in fact absent and remains absent, 2. The name of the Designated Alternate is on a list of approved Designated Alternates selected by the appropriate Segment Membership according to these Procedures or Segment Procedures, and on file with both the NAESB Office and the EC. 	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
В.	 Election of Designated Alternates 1. Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to twelve (12) Designated Alternates overall. 2. Designated Alternates will be selected annually by the same election that determines new EC Members, with those unsuccessful EC Member candidate(s) receiving the greatest number of votes becoming a Designated Alternate in the event that there already are or are expected to be less than three Designated Alternates from the Segment in the coming year. 3. A Designated Alternate will attend and vote at meetings of the EC of the REQ or of the larger EC of NAESB, when one or more regular EC Members from the Segment are unable to attend a meeting of the EC and when the absent EC Member or the Vice-Chair of the REQ EC requests their attendance as a Designated Alternate. 	Note: The proposed method of selecting Designated Alternates as found in the draft RGQ Procedures (Jan-4-02) provides for a separate election of Designated Alternates following the election of EC Members. As explained in the discussions on Jan 17 th , that proposed approach may be more successful in attracting people to serve as Designated Alternates and should be considered in place of the provisions shown here.
	TIONS 11 – 18 Reserved for future use	
	(Not addressed in prior drafts, and needed)	RGQ draft procedures include the following provisions for amendment of the Quadrant Procedures: "In order for Retail Gas Quadrant Procedures to be amended, upon petition of at least five (5) Members of the Quadrant, the Vice Chair of the EC for the Quadrant shall announce a Quadrant meeting. Such announcement shall provide for at least 30-days notice. In order to transact business at the

Draft V2.2 Retail Electric Quadrant Procedures -	Comment or Comparable Point in Gas Quadrant Draft
Restructured to Follow NAESB Bylaws	

SECTION 19 – TRANSITION PROCEDURES

	The requirements of the NAESB Bylaws for a minimum of 5 Members per Segment and 40 Members per Quadrant do not provide for enough Members to satisfy the REQ requirements for election of Directors and EC Members as stated in Sections 7 and 10 of these Procedures. Consequently, until a Segment of the Quadrant has at least eight (8) Members, the number of Directors and EC Members elected by the Segment Membership will be halved, to 2 Directors and 2 EC Members per Segment and Segment balanced voting will remain intact by adjusting the weight of all votes to achieve balance between Segments in the REQ. The following provisions in this Section 19 will apply and supercede the requirements of Sections 7 and 10 until the REQ Segment achieves a minimum of eight (8) Members.	
Sect	ion 19.1 – Initial Election of Directors	
А.	Acceptance of the REQ by the Board will involve seating of the REQ Directors at the NAESB Board meeting at which an acceptance vote is cast. This requires that there be REQ	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	Directors who may be seated. The founding membership of the REQ will elect Directors of their choosing by means that are reasonably consistent with NAESB Bylaws and the requirements of Section 7 of these Procedures, subject to the conditions addressed by this Section. Selection of candidates and their election will be by procedures agreed to by consensus or voting methods adopted by the group. Such elected representatives will be presented to the NAESB Board for acceptance as Directors of the REQ, and acceptance by the Board will place all the requirements and restrictions of the Bylaws, including these Procedures, upon those individuals.	
В.	 Term of Office of Initial Directors 1. In the event that Directors are elected from a Segment with at least five (5), but less than eight (8) Members, two Directors will be elected, each to serve a 3 year term, and both will be either Group A or Group B directors as determined by the expiration year of their original term. 2. In the event that a Segment has eight (8) or more Members, four Directors will be elected. The two candidates having the most votes will serve three (3) year terms. The remaining two Directors elected by lesser vote counts will serve two year terms. 	Per discussion on Jan 17 th , RGQ approach is to designate the number of Directors when Quadrant is fully operational and begin operations with unfilled Board seats, if necessary.
C.	 Interim Elections of Directors In the event that any Segment within the Quadrant achieves a minimum membership of eight Members after operating with less for a period of time after an election held under the requirements of Section 19.1(B), then 1. a second election of Directors shall be held within 60 days of the addition of the eighth Member of the Segment in order to select two additional Directors from that Segment to serve alongside the initially elected Directors; 	<i>Note:</i> This subsection should probably include an expiration for it's provisions of no more than 2 years beyond acceptance of the Quadrant into NAESB. Elections in the second year would result in the second group of Directors serving for a single year before standing for re-election.

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	 the term of office of the second group elected at this time will expire one year prior to the term of the initially elected Directors, thereby allowing for two Group A and two Group B Directors from each Segment. 	
Sect	ion 19.2 – Initial Election of Executive Committee	
Α.	Acceptance of the REQ by the Board will involve seating of the REQ Directors at the NAESB Board meeting at which an acceptance vote is cast. Initial operation of the REQ will require that an Executive Committee be in place. The founding membership of the REQ will elect an EC of their choosing by means that are reasonably consistent with NAESB Bylaws and the requirements of Section 10 of these Procedures, subject to the conditions addressed by subsection B of this Section 19.2. Selection of candidates and their election will be by procedures agreed to by consensus or voting methods adopted by the group. Such elected representatives will be presented to the NAESB Board for acceptance as representatives of the REQ, and acceptance by the Board will place all the requirements and restrictions of these Procedures upon those individuals.	
В.	 Term of Office of Initial EC Members The initial group of EC Members will serve a 2 year term of office as either Group A or Group B Members, as determined by the expiration year of their original term of office. 1. In the event that EC Members are elected from a Segment with at least five (5), but less than eight (8) Members, two EC Members will be elected, each to serve a 3 year term, and both will be either Group A or Group B EC Members as determined by the expiration year of their original term. The candidate with the greatest number of votes will be 	

	Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
	 designated the EC Chair; in the event of a tie vote, a coin toss will determine the EC Chair and Vice-Chair positions. 2. In the event that a Segment has eight (8) or more Members, four EC Members will be elected. The two candidates having the most votes will serve a three (3) year term. The remaining two EC Members elected by lesser vote counts will each serve a two year term. A separate election within the EC will determine which Member becomes the Chair and which becomes the Vice-Chair, by vote of the four EC Members 	
C.	 Interim Elections of EC Members In the event that the Segment achieves a minimum Membership of eight Members after operating with less for a period of time following an election held under the provisions of Section 19.2(B), then 1. A second election of EC Members shall be held within 60 days of the addition of the eighth Member of the Segment in order to select two additional EC Members from the Segment to serve alongside the initially elected EC Members; 2. The term of office of the second group elected at this time will expire one year prior to the term of the initially elected EC Members, thereby allowing for two Group A and two Group B EC Members from each Segment. 3. The previously elected Chair and Vice-Chair of the Quadrant EC will continue in their official role until the end of the operating year. Chair rotation will proceed per the requirements of Section 10.1 4.2(nm)(i). 	
Sect	ion 19.3 – Balanced Voting During the Transition Period	
	Any Segment which has elected Directors or EC Members	

 Draft V2.2 Retail Electric Quadrant Procedures - Restructured to Follow NAESB Bylaws	Comment or Comparable Point in Gas Quadrant Draft
under the provisions of this section will retain the full Segment	

votes entitled to it under the Balanced Voting principle of the Bylaws, and its representatives' votes will count for either one or two Segment votes as is appropriate to the situation.

			applicable section in			Affiliation 8	k	Additional Comment or Explanatory
Item	Ref V. 1	Ref V. 2.1	V 2.2	Comment Made on Version 1	From	Segment	Disposition In Version 2.1	Note
1	1.1	1.1	2.1 (A)	[having a Purpose section is good]	W. Boswell	Chairman, NAESB	no change	Latest draft of RGQ Procedures (1-4-02) under development by AGA-GRGS incorporates very similar wording
2	1.1	1.1	longer	phrase "voluntary standards and model business practices" should be revised (throughout the document) to refer only to "model business practices"	B. Alexander	NASUCA & AARP / End Users	no change	this issue has been debated during several Plenary meetings, with the resolution being to continue to offer "voluntary standards" which states may or may not choose to adopt in their jurisdiction
3	1.2	1.2	2.1(B) + explanatory note	first paragraph is good; second paragraph is redundant with NAESB Bylaws and unnecessary	W. Boswell	Chairman, NAESB		
4	1.2	1.2	2.2(B)	second paragraph does contain Quadrant specific provisions	J. Unruh	Reliant / Suppliers		
5	1.2	1.2		closing statement in second paragraph may raise a red flag with state agencies	B. Alexander	NASUCA & AARP / End Users	Comments 3-5 have been addressed by a) retention of the first sentence of the second paragraph, b) rewriting the closing sentence of the former second paragraph, and c) conversion of the bulk of the	
6	1.3	1.3	see item #7	This [Principles] section is fully redundant with the Bylaws and is not needed.	W. Boswell	Chairman, NAESB		
7	1.3	1.3		[Having this section here may be redundant, but is helpful to readers unfamiliar with the NAESB Bylaws and the history of this effort]	J. Unruh	Reliant / Suppliers	Comments 6 & 7 addressed by converting former Section 1.3 into an expansion of the explanatory note cited above.	Latest draft of RGQ Procedures (1-4-02) addresses these points in a generally similar explanatory paragraph
8	2.0	2.0	see item #9	[some definitions listed are redundant with the Bylaws and need not be here]	W. Boswell	Chairman, NAESB		
9	2.0	2.0		[flagged several redundant definitions as helpful in reading the document, recommending that they be kept for now]	J. Unruh	Reliant / Suppliers	comments 8 & 9 addressed by retaining several definitions in italics, indicating that these are already defined in the Bylaws and show in the current draft	

Item	Ref V. 1	Ref V. 2.1	applicable section in V 2.2	Comment Made on Version 1	From	Affiliation & Segment	Disposition In Version 2.1	Additional Comment or Explanatory Note
10	2.0	2.0	1.0(B)	change {"End User Segment Member" means an entity (e.g., corporation, partnership, proprietorship, or individual) engaged in the consumption of electricity} to {"End User Segment Member" means a person or legal entity that consumes	B. Alexander	NASUCA & AARP / End Users	changed to: "End User Segment Member" means a person that consumes electricity	NAESB has adopted the position that the legal definition of "person" includes individuals, corporations and other entities. See also comments 15, 16 & 17
11	2.0	2.0	1.0(C)	electricity 3 "NAESB Office" does not appear to be a correct definition	J. Unruh	Reliant / Suppliers	no change	Defined for the purposes of the document as written; need to resolve whether the NAESB Office, as defined will have the role stated within its new roles in NAESB
12	2.0	2.0	deleted	part a) of definition shown for Quorum is not needed	W. Boswell	Chairman, NAESB	Entire definition removed from V2.2 as Quorum is defined	
13	2.0	2.0	see Item #12	[purpose of Quorum definition part a) is unclear]	J. Unruh	Reliant / Suppliers	resolution of comments 12 & 13 is that part a) has been deleted and the definition is	
14	2.0	2.0	1.0(G) & (H)	amend definition of Services Segment Member, and possibly Supplier Segment Member, to include individuals	K. Sappenfield	PanCanadian Energy/ Suppliers	change made to Services Segment Member definition	Is this appropriate for the Supplier definition as well?
15	3.0	3.0	2.3(A)	change " Members of the REQ are entities which meet" to "Members of the REQ are individuals or legal entities which meet"	B. Alexander	NASUCA & AARP / End Users		
16	various	various	various	[reword sections that use "entities"]	K. Sappenfield	PanCanadian Energy/ Suppliers		
17	various	various	various	[substitute 'person' for 'entity' throughout the document]	B. Alexander	AARP / End Users		
18	3.0 & 3.1	3.0 & 3.1	various	NAESB uses "person" as the legal equivalent of individuals, corporations, and any other entity with legal standing.	W. Boswell	Chairman, NAESB	comments 15, 16, 17 & 18 resolved by using "person" or "persons" where applicable; this engendered many small changes in wording throughout the document: in some cases.	

Comments Regarding Draft Version 1 & Their Disposition In Versions 2.1 & 2.2 (Bylaws Format)

Item	Ref V. 1	Ref V. 2.1	applicable section in V 2.2	Comment Made on Version 1	From	Affiliation & Segment	Disposition In Version 2.1	Additional Comment or Explanator Note
item	Rei V. I	Rei V. 2.1	V 2.2	comment made on version 1	From	Segment	Disposition in version 2.1	Note
19	3.1	3.1	2.3(A)	delete the term "Entities" and replace it with "Parties" at the beginning of each description	K. Sappenfield	PanCanadian Energy/ Suppliers	descriptions rewritten for all four Segments to simplify the construction	
20	3.2	3.2	5.2(A)	add to requirement (a) the statement: "as determined by a simple majority of the Segment Membership"	J. Unruh	Reliant / Suppliers	added for consistency with requirement (b)	
21	3.2	3.2	5.2	replace "individual or legal entity", "person or entity" and "entity" in 3.2, 3.2(a) & (b) with "Member"	K. Sappenfield	PanCanadian Energy/ Suppliers	change made and lines revised to accommodate	
22	3.2	3.2	5.2(A)(2)- (b)(iii)	"provides a letter of authorization from the group"	W. Boswell	Chairman, NAESB	change made	further comment to the effect that NAESB agrees with the need to require periodic reconfirmation of right to represent the group
23	3.2	3.2	5.2(A)	in (b)(ii)(c) delete the phrase: "by the representative"	W. Boswell	Chairman, NAESB	change made	
24	3.2	3.2		proposed language providing for reduction of membership fee for residential end user segment Members to a maximum of \$500 per annum.	B. Alexander	NASUCA & AARP / End Users		issue currently being explored; unresolved
25	3.2, 3.3 & 3.4	3.2, 3.3 & 3.4	deleted	3.2(b)(iii), 3.3 and 3.4 are all redundant with the Bylaws	W. Boswell	Chairman, NAESB	pertinent parts of Sections 3.2 & 3.3 converted to explanatory notes: see also comment 33	
26	3.4	3.3	5.1(C)	(in reference to comment 26); Section 3.4 provides helpful clarifying details not specified in the Bylaws and therefore unique to the Quadrant; retain this section.	J. Unruh	Reliant / Suppliers	Section 3.4 kept for now as renumbered Section 3.3	
27	3.5	3.4	5.1(A)	What is the purpose of this section?	J. Unruh	Reliant / Suppliers		Purpose of current Section 3.4 is to declare the intent of the Quadrant regarding the desired qualifications of representatives; specifically, persons representing members should have first hand knowledge of the business issues, operating authority to speak for the member and authority to cast a vote in

meetings without consultation with the

member on minor issues.

Item	Ref V. 1	Ref V. 2.1	applicable section in V 2.2	Comment Made on Version 1	From	Affiliation & Segment	Disposition In Version 2.1	Additional Comment or Explanator Note
28	3.6	3.6	N/A	remove closing statement following item (b) as it is redundant with the Bylaws	W. Boswell	Chairman, NAESB	change made	
29	3.6	3.6		add (c) removal for failure to attend requisite number of meetings	J. Unruh	Reliant / Suppliers	not added, as Bylaws provide for removal of EC or Board members for failure to attend meetings, but there is no minimum number of meetings a	
30	4.0	4.0	various	substitute the defined NAESB term of 'Director' instead of 'Board members', 'Board representation' and 'member of the Board'.	K. Sappenfield	PanCanadian Energy/ Suppliers	changes made	
31	4.1(b)	4.1(b)	7.3(A)	clarify to state that each Segment will elect four directors	K. Sappenfield	PanCanadian Energy/ Suppliers	changes made	
32	4.1(f)	4.1(f)	7.4(B)	do not agree with unlimited terms of office	J. Unruh	Reliant / Suppliers		This provision needs to be confirmed o removed by agreement of the prospective Quadrant Membership
33	4.1(f)	4.1(f)		fix some typos and a word deletion suggested	J. Unruh	Reliant / Suppliers	changes made	
34	4.1(g)	4.1(g)	7.2(B)	add 'REQ' to clarify to that an REQ representative may not be both a Director and and EC member at the same time	K. Sappenfield		change made	
35	4.1(g)	4.1(g)	7.2(B)	replace the word "accepts" with "seats"	W. Boswell	Chairman, NAESB	change made	
36	4.1(g)	4.1(g)	7.2(B)	replace the word "them" with "him/her"	J. Unruh	Reliant / Suppliers	change made	
37	4.1(h)	4.1(h)		as written, violates Labor Law; rephrase to say "No two Directors elected by the REQ may be employees of the same Member entity"	W. Boswell	Chairman, NAESB	change made	
38	4.1(h)	4.1(h)	V2.1 has	expands on comment 37, to further change following language to: " same Member entity, where an entity holds Membership in multiple Segments", and adds the phrase "in both situtions" later in the paragraph; also requested a clarifying example	J. Unruh 4 of 8	Reliant / Suppliers	changes made; example added; may have been impacted by subsequent changes to eliminate "entity"	The provisions, as written, need to be confirmed or changed by agreement of the prospective Quadrant Membership

Item	Ref V. 1	Ref V. 2.1	applicable section in V 2.2	Comment Made on Version 1	From	Affiliation & Segment	& Disposition In Version 2.1	Additional Comment or Explanatory Note
39	4.1(i)	4.1(j)	7.3(B)	Why will the EC coordinate the election of Directors?	J. Unruh	Reliant / Suppliers	see also "comment" #40; as a result of comment #40, this subsection has become subsection 4.1(j)	Somebody within NAESB will have to coordinate various elections; having the Vice-Chair of the Quadrant EC responsible for coordinating all Quadrant level elections (as provided for within the draft) defines a specific role for the Vice-Chair and removes some vagueness on how elections would be accomplished. This provision does not exist in the current NAESB operating procedures or Bylaws. Adoption of this approach over some other approach needs to be discussed and agreed upon by prospective Quadrant Membership.
40		4.1(i) + others	7.3(B) & (C); and 19.1	comment regarding draft End User Segment Procedures indicating that provisions regarding elections actually belong in the Quadrant document	W. Boswell	Chairman, NAESB	inserted a new subsection 4.1(i here and new subsections 4.2(i), 4.2(o), 4.2(p) & new sections 8.1 and 8.2.) The provisions, as written, need to be confirmed or changed by agreement of the prospective Quadrant Membership
41	4.1(j)	4.1(k)	7.6(A)	need to confirm the chosen percentages for majority voting	W. Boswell	Chairman, NAESB		The provisions, as written, need to be confirmed or changed by agreement of the prospective Quadrant Membership
42	4.1(k)	4.1(I)	deleted	redundant with Bylaws	W. Boswell	Chairman, NAESB	actually in conflict with Bylaws	
43	4.1(k)	4.1(l)	deleted	regarding comment 43; second sentence is Quadrant specific and should be kept	J. Unruh	Reliant / Suppliers	first sentence deleted, subsequent parts kept as subsection 4.1(I)	deletion of the rest of the section may still be appropriate; subject to further discussion
44	4.1(l)	4.1(m)	7.4(C)	change "their" to "his/her"	J. Unruh	Reliant / Suppliers	change made	uscussion
45	4.2	4.2	10	[changes were suggested to conform the structure of this section to that of Section 4.1]	J. Unruh	Reliant / Suppliers	changes made	
46	4.2	4.2	10.3(E)	do not agree with unlimited terms of office	J. Unruh	Reliant / Suppliers		This provision needs to be confirmed or removed by agreement of the prospective Quadrant Membership
47	4.2	4.2	10	change "their" to "his/her" in several locations within the section	J. Unruh	Reliant / Suppliers	changes made	

Item	Ref V. 1	Ref V. 2.1	applicable section in V 2.2	Comment Made on Version 1	From	Affiliation & Segment	Disposition In Version 2.1	Additional Comment or Explanatory Note
48	4.2(g)	4.2(g)		section prohibits Members from holding a seat on both Board and EC from REQ Segment simultaneously; AGA-GRGS draft RGQ Procedures prohibit only a representative from doing both.	M. Scheel	Dynegy / Suppliers		This provision needs to be confirmed or modified by agreement of the prospective Quadrant Membership
49	4.2(h)	4.2(h)	10.2(B)	same as comment #38	W. Boswell	Chairman, NAESB	change made	
50	4.2(h)	4.2(h)	10.2(B); V2.1 example deleted	same as comment #39	J. Unruh	Reliant / Suppliers	changes made; example added	The provisions, as written, need to be confirmed or changed by agreement of the prospective Quadrant Membership
51	4.2(i)	4.2(j)	19.2(A)	[there is no existing EC when the Quadrant is first formed to carry out the provisions of this subsection]	J. Unruh	Reliant / Suppliers	added language to Section 8 to provide a proposed means of dealing with the issue	The provisions, as written, need to be confirmed or changed by agreement of the prospective Quadrant Membership
52	4.2(j)	4.2(k)	10.6(C)	same as comment #41	W. Boswell	Chairman, NAESB		
53	4.2(k)			redundant with Bylaws	W. Boswell	Chairman, NAESB	subsection was deleted	
54	4.2(m)	4.2(m)	10.3(F)	change "their" to "his/her"	J. Unruh	Reliant / Suppliers	changes made	
55	4.2(n)(iii)	4.2(n)(ii)	10.1(A)(2); further clarified	how does the Secretary provision work given GISB's past practice?	J. Unruh	Reliant / Suppliers	subsection revised to better reflect existing practice within NAESB	provisions still need to be discussed and agreed upon by prospective Quadrant Membership
56	4.3(b)	4.2(n)(i)	10.1(A)(1)	first sentence is redundant with Bylaws, second belongs in Section 4.2	W. Boswell	Chairman, NAESB	first sentence deleted; second provision now resides in	
57	4.3(c)	4.3(b)	deleted; redundant	should apply to REQ EC Chair and Vice-Chair	J. Unruh	Reliant / Suppliers	correction made	
58	4.3(d)	4.3(c)	10.4(B)(1)	this proposed procedure resolves a problem that had been recognized by the Board	W. Boswell	Chairman, NAESB	no change required	
59	4.3(d)	4.3(d)	10.4(B)(2)	how is the meeting Secretary determined under this situation?	J. Unruh	Reliant / Suppliers	added a new subsection 4.3(d) to address this	
60	4.3(e)	4.3(e)	10.4(A)(1)	replace "in person" with "held"	W. Boswell	Chairman, NAESB	change made	past GISB practice has included numerous EC meetings held by means of telecommunications
61	4.3(e)	4.3(e)	10.3(A)(1)	wording suggested to simplify the paragraph	J. Unruh	Reliant / Suppliers	change made	

Item	Ref V. 1	Ref V. 2.1	applicable section in V 2.2	Comment Made on Version 1	From	Affiliation & Segment	Disposition In Version 2.1	Additional Comment or Explanatory Note
62	4.3(f)	4.3(f)	10.6(A)(1)-(c)	Why is there a vote? Should be automatic removal	J. Unruh	Reliant / Suppliers		The provisions, as written, need to be confirmed or changed by agreement of the prospective Quadrant Membership
63	4.4	4.4	10.5	Section 4.4 belongs in the "NAESB Operating Procedures (NAESBops)" rather than in the Quadrant document, fulfilling a recognized need for such procedures	W. Boswell	Chairman, NAESB	V2.2 includes note that provisions may belong in NAESBops rather than REQ document	
64	4.4	4.4	10.5	regarding comment 64; this section should remain in the document as it is quadrant specific since there are no NAESBops at this time	J. Unruh	Reliant / Suppliers	left in the document at the present time, with an added note that this section may be more appropriate in NAESBops	Needs to be resolved in discussions with NAESB
65	4.4(a)(i)	4.4(a)(i)	10.5(A)(1)- (a)	add the provision that the EC will also designate whether balanced voting will be applicable in establishing a specific working group	K. Sappenfield	PanCanadian Energy/ Suppliers		
66	4.4(a)(iii)-(3)	4.4(a)(iii)- (3)	10.5(A)(1)- (c)(iii)	[revise to permit shorter notice periods for meetings once the working group/task force is established]	K. Sappenfield	PanCanadian Energy/ Suppliers	changes made	
67	[formerly 4.5]			this section covers topics that are only appropriate to the Bylaws for all of NAESB, as the Triage Committee only functions on that level.	W. Boswell	Reliant / Suppliers	section deleted	
68	[formerly 5.1]			this subsection is neither needed nor correct, as Section 5.3 covers the issues	W. Boswell	Reliant / Suppliers	section deleted	
69	6.1(b)	6.1(b)	2.2(C)	this subsection belongs in NAESBops or the Bylaws as it is an issue for all Quadrants	W. Boswell	Chairman, NAESB	left in the document at the present time, with an added note that this section may be more appropriate in NAESBops	Needs to be resolved in discussions with NAESB
70	6.1(b)	6.1(b)	2.2(C)	regarding comment 69; this section should remain in the document as it is quadrant specific since there are no NAESBops at this time and the Bylaws do not reflect the issue	J. Unruh	Reliant / Suppliers	note abbrobrate in NAESBobs left in the document at the present time, with an added note that this section may be more appropriate in the Bylaws or NAESBobs	Needs to be resolved in discussions with NAESB

Item	Ref V. 1	Ref V. 2.1	applicable section in V 2.2	Comment Made on Version 1	From	Affiliation & Segment	Disposition In Version 2.1	Additional Comment or Explanatory Note
71	6.1(b)	6.1(b)	2.2(C)	rewrite to clarify limitations / requirements	K. Sappenfield	PanCanadian Energy/ Suppliers	left in the document at the present time, with an added note that this section may be	not rewritten; other changes bring into question whether this section is needed at all; see note in the text
72	6.1(c)			this provision is not necessary under current GISB EC practice	J. Unruh	Reliant / Suppliers	more appropriate in the Bylaws section deleted	
73	7.1			as written, the section is inconsistent with unwritten GISB practice that will carry over into NAESB	W. Boswell	Chairman, NAESB	section deleted	Agreed in discussing the comment that NAESB needs to formally document the voting practices for a Ratification ballot in all Quadrants, so this belongs either in NAESBops or the Bylaws
74	7.1			as written, the rules are too complex to be workable	J. Unruh	Reliant / Suppliers	(section deleted)	
75	7.1			provisions for designating alternates needs to be in the Segment Procedures	M. Scheel	Dynegy / Suppliers	(section deleted)	
76	7.1			rewrite to address two different comments	K. Sappenfield	PanCanadian Energy/ Suppliers	(section deleted)	
77	7.2	7.1	10.9(A)	add provision to allow EC member to notify NAESB Office of absence	K. Sappenfield		change made	
78	8.0	8.0	19	provisions of this section need further clarification; a minor wording change also suggested	W. Boswell	Chairman, NAESB		
79	8.0	8.0	19	provisions of this section need further clarification, particularly regarding balanced voting	M. Scheel	Dynegy / Suppliers		
80	8.0	8.0	19	revise language as shown in redline comments throughout this section	K. Sappenfield	PanCanadian Energy/ Suppliers		
81	8.0	8.0	19	provisions of this section need further clarification, particularly regarding balanced voting	J. Unruh	Reliant / Suppliers	(section revised extensively to respond to comments 78-81)	the provisions of this section do need further clarification and must be agreed to by prospective Quadrant Membership

1 To the UBP-RSB membership, and other interested parties

2

The Governance Committee of UBP-RSB is issuing this document to you to assist you in
preparing for the Retail Quadrants meeting scheduled for January 17th in Washington DC.

5

6 The document that follows is a second draft of proposed Retail Electric Quadrant 7 Procedures, and should be reviewed along with a companion document (spreadsheet) 8 listing the comments received on Version 1 and their disposition within this document. The 9 proposed procedures within this document are not final, as several sections still contain 10 "placeholder" provisions that need discussion and approval by prospective members of the 11 Retail Electric Quadrant.

12

There are several locations in this document where it is suggested that the provisions written are redundant with the Bylaws. The GC is retaining these sections in italics, and noted as being there for explanatory purposes, until such time as the Quadrant Procedures are formally adopted by NAESB. At that point, the document would be stripped of explanatory sections as it will become an integral part of the NAESB Bylaws.

18

19 Please feel free to provide comments to your Governance Committee representatives. The 20 more comments we receive, the better we can reflect your issues, concerns and needs, and 21 the better the Quadrant Procedures will tend to be.

22

23 Thank you for your time, and your interest.

- 24
- 25 Charles L. Driggs
- 26 Chair, Governance Committee
- 27
- 28
- 29

30	
31	
32	
33	
34	
35	
36	
37	
38	
39	NORTH AMERICAN ENERGY STANDARDS BOARD
40	BYLAWS ADDENDUM
41	
42	EXHIBIT 4
43	
44	
45	RETAIL ELECTRIC QUADRANT PROCEDURES

46 47

1.0 THE RETAIL ELECTRIC QUADRANT

- 1.1 Purpose 48 49 The purpose of the Retail Electric Quadrant of the North American Energy 50 Standards Board (NAESB) is to propose, evaluate and adopt voluntary standards 51 and model business practices to promote competitive, efficient and reliable service 52 in the retail electric industry. 53 54 1.2 Scope 55 The Retail Electric Quadrant is to address issues and practices that are within the 56 scope of NAESB and appropriate to electric usage at the individual consumer level—that is, usage by an individual, partnership, corporation, or other entity 57 58 consuming electricity at one or more facilities served by an Electric Distributor. 59
- Issues may also be addressed in conjunction with other Quadrants when such
 issues are common to both electric distribution at the individual consumer level
 and some combination of wholesale or retail activities in the electric or natural gas
 industries.
- 64

65 Explanatory Note:

66 The Retail Electric Quadrant would normally address issues and practices which 67 are generally within the jurisdiction of state or local regulatory authorities. As 68 such, the findings, recommendations and proposed standards or model business 69 practices produced by this Quadrant of NAESB may be implemented in individual 70 jurisdictions only to the extent that they are consistent with applicable law and 71 regulation. Standards and model business practices that prove successful in those states that adopt them might well be adopted by additional jurisdictions if
compatible with policies in their jurisdiction.

74

75 Furthermore, the Retail Electric Quadrant will operate in conformance with the 76 principles of NAESB as established in Section 2.2 (b) of the NAESB Bylaws, and in 77 compliance with ANSI guidelines as codified therein. In keeping with these 78 Principles, the Retail Electric Quadrant encourages membership from across the 79 spectrum of the Retail Electric industry, reflecting the principle of inclusiveness. All Retail Electric Quadrant meetings, including those of its Members, Executive 80 81 Committee (EC), subcommittees and task forces, shall be open to any interested 82 member of the public and that all minutes thereof shall be available to the public in keeping with the principle of openness. 83

84

85 2.0 **DEFINITIONS**:

86 When final and adopted, this document becomes an Exhibit to the NAESB Bylaws,

87 and the Bylaws define a number of terms used throughout the Bylaws and the

88 Exhibits. Some definitions listed in this section at this time are shown only for

89 ease of reading this document in the absence of the NAESB Bylaws; definitions

90 *included for this purpose are presented in italics.*

91 "Agent" means an individual, partnership, firm, corporation, or other entity

92 representing the interests of a member of the End User Segment, but who, itself,

93 is not necessarily a member of the Segment.

94 "Board" means the Board of Directors of NAESB

95 "Bylaws" means the corporate bylaws of NAESB as amended from time to time.

- 96 "Designated Alternate" is defined as a person, specifically named by a Segment in
 97 a list submitted to the NAESB office, designated to serve in place of an EC Member
 98 who is unable to attend an EC meeting.
- 99 "Director" means an individual serving on the Board of Directors either from the
 100 Retail Electric Quadrant or any other Quadrant.
- 101 "Distributor Segment Member" means a person (e.g., corporation or partnership)
- 102 engaged in the local distribution of electricity, or its representative (e.g.,
- 103 employee, officer, director, agent, or partner) who is authorized to cast that
- 104 Segment Member's vote, that satisfies the requirements for Membership set forth
- 105 in Article 5 of the bylaws of NAESB, and Section 3 of these Procedures, and further
- 106 satisfies the qualifications for Distributor Segment Membership of the Retail
- 107 Electric Quadrant as set forth in the Distributor Segment Procedures.
- 108 "EC" means the Executive Committee of NAESB or a subset of the Executive
- 109 Committee of NAESB consisting of members from the Retail Electric Quadrant, or
- 110 of the Retail Electric Quadrant and one or two other Quadrants in combination, as
- 111 appropriate to the context.
- "End User Segment Member" means a person (e.g., corporation, partnership,
- proprietorship, or individual) engaged in the consumption of electricity delivered
- by an Electric Distributor, or its representative (e.g., employee, officer, director,
- agent, or partner) who is authorized to cast that Segment Member's vote, that
- satisfies the requirements for Membership set forth in Article 5 of the Bylaws of
- 117 NAESB, and Section 3 of these Procedures, and further satisfies the qualifications
- 118 for End User Segment Membership of the Retail Electric Quadrant as set forth in
- 119 the End User Segment Procedures.
- 120 "NAESB" means the North American Energy Standards Board, Inc.
- 121 "NAESB Office" means the office of the Secretary of NAESB

122	"Quorum" is defined, for the purposes of a Quadrant EC meeting, as a minimum of
123	one EC member representative from each Segment within the Retail Electric
124	Quadrant, and at least half of the currently serving Retail Electric Quadrant EC
125	members overall, present either in person or via telecommunications. and as
126	stated in Section 6.6 of the NAESB Bylaws in all other respects.
127	"REQ" means the Retail Electric Quadrant
128	"Segment" means one of the co-equal Membership Segments of the NAESB Retail
129	Electric Quadrant representing the following four (4) segments of the retail electric
130	industry: Suppliers, Distributors, End Users, and Services.
131	"Segment Membership" means the Segment Members collectively.
132	"Services Segment Member" means a person (e.g., corporation, partnership or
133	proprietorship or individual) that provides services to participants in the retail
134	electric industry or an entity not otherwise declared in another segment, or its
135	representative (e.g., employee, officer, director, agent, or partner) who is authorized
136	to cast that Segment Member's vote, that
137	a) satisfies the requirements for Membership set forth in Article 5 of the
138	Bylaws of NAESB, and Section 3 of these Procedures, and
139	b) further satisfies the qualifications for Services Segment Membership of the
140	Retail Electric Quadrant as set forth in the Services Segment Procedures.
141	"Supplier Segment Member" means a person (e.g., corporation, partnership, or
142	proprietorship) engaged in the competitive sale of electricity to end-users, or its
143	representative (e.g., employee, officer, director, agent, or partner) who is
144	authorized to cast that Segment Member's vote, that
145	a) satisfies the requirements for Membership set forth in Article 5 of the
146	Bylaws of NAESB and Section 3 of these Procedures, and

further satisfies the qualifications for Supplier Segment Membership of the

148 Retail Electric Quadrant as set forth in the Supplier Segment Procedures. 149 150 All other capitalized terms, if not defined in this Section 2.0, shall have the same 151 definitions as specified in the Bylaws or Certificate of Incorporation of NAESB. 152 153 3.0 MEMBERSHIP REQUIREMENTS 154 The definition of Membership applies to all Segments within the REQ. Members of 155 the REQ are persons who meet the following membership requirements in any one 156 of the four Segments of this Quadrant. Upon applying for Voting Membership (as 157 described in NAESB Bylaws Sections 1.1 and 5.1) in the REQ, each prospective Member shall declare the Segment with which they are to be identified. 158 159 160 3.1 Segment Membership Eligibility 161 a) Distributor Segment 162 Parties eligible for membership in this Segment are engaged in the local 163 distribution of electricity. 164 b) End User Segment 165 Parties eligible for membership in this Segment consume electricity or represent consumers of electricity. 166 167 c) Suppliers Segment 168 Parties eligible for membership in this Segment are service providers and marketers, engaged in the competitive sale of electricity to end-users 169 170 c) Services Segment 171 Parties eligible for membership in this Segment provide services to 172 participants in the retail electric industry, including equipment

147

b)

173			manu	ifacture	rs, equipment vendors, software providers, consultants, and
174			other	compa	nies or individuals not otherwise eligible for membership in
175			anoth	ier Segr	nent.
176					
177	3.2	Votir	ng Merr	nbershi	p
178		Mem	bership	and vo [.]	ting rights in the REQ of NAESB shall be open to any Member
179		that r	meets tl	he follo	wing requirements:
180		a)	The M	lember	has a significant business interest ¹ in the retail electric market
181			(or is	a repre	esentative or Agent of such person), as determined by a simple
182			major	rity of t	he segment membership
183		b)	The M	lember	may be a trade association or an advocacy group representing
184			a gro	up of pr	ospective members, provided that
185			(i)	the gr	roup represents an interest that is identified within a Segment,
186				yet is	
187				(1)	not represented, or
188				(2)	is underrepresented
189				in the	judgement of a simple majority of the Segment membership,
190				and	
191			(ii)	The p	rospective representative's relationship to the group wishing
192				to be	represented in this manner is documented in one of the
193				follow	ving ways:
194				(a)	Has statutorily or regulatory designated authority to
195					represent the interests of the subject group

 $^{^{1}}$ as used in this section, "significant business interests" specifically includes the interests of statutorily appointed consumer advocates

- 196 (b) Is generally recognized representative of the subject group
 197 for other transactions (e.g. provides group insurance, group
 198 buying of another product, etc.)
- 199(c) provides a letter of authorization from the group, including200an expiration date for the letter of authorization201(iii)no more than two entities within the group represented202subsequently join as full Members of the Segment, at which point
- 203the representative's authorization for representation and voting204rights will be immediately voided unless reinstituted by letter for205the remainder of the operating year by the remaining members of206the representative's constituency.
- 207
- Note: Per provisions of the NAESB Bylaws, the voting rights of any REQ Member
 shall be suspended if its dues are delinquent, and attendance of that delinquent
 Member at any meeting shall not be counted in determination of a Quorum being
 present.
- 212

213 3.3 Multiple Memberships

Memberships in multiple Segments of the REQ are permissible for any person provided each Membership is filed and declared with NAESB, the person meets the membership requirements of each Segment joined, and Membership dues are paid for each Segment.

- a) Only one Membership per Segment is permissible for any person.
- b) Multiple companies under common control within a corporate organization
 that desire to become members must join individually. Members cannot

221			extend their membership to their parent company, affiliates, or
222			subsidiaries.
223			
224	3.4	Memb	per Representation
225		Repre	sentatives chosen by any person in any Segment should have, but are not
226		requir	ed to have, corporate responsibility for the business interests causing the
227		persor	n to seek membership in the Segment.
228			
229	3.5	Remo	oval of Members
230		Memb	ers who cease to have a significant business interest pertaining to the
231		descri	ptions contained in sections 3.1(a)-3.1(d) may be removed from
232		memb	pership by either
233		a)	a simple majority vote of their declared Segment membership, or
234		b)	resolution of the REQ Executive Committee
235			
236	4.0	ORGA	NIZATION OF THE RETAIL ELECTRIC QUADRANT:
237		The R	EQ shall elect representatives to the Board of Directors of NAESB and elect
238		an Exe	ecutive Committee (EC), which shall participate in the larger NAESB EC body
239		as wel	Il as govern the activities of the REQ.
240			
241	4.1	Direc	tors
242		a)	The REQ shall elect sixteen (16) NAESB Directors, subject to the provisions
243			of Section 8 of these Procedures. All Directors from the REQ will be
244			comprised of an equal number of representatives elected from each
245			Segment of the Quadrant.
b) Each Segment of the Quadrant will elect four (4) Directors, subject to the
provisions of Section 8 of these Procedures. Nominations for and election
of all Directors will be in accordance with Segment Procedures.

- c) In the event that a Director resigns or otherwise vacates their Board seat,
 the Quadrant will hold an election within 60 days to fill the vacant seat
 with a candidate from the same Segment as the original Director.
- 252 d) Directors shall be elected for two-year terms, with half of the terms
 253 expiring in alternating years.
- e) Subject to the provisions of Section 8, two Directors will be elected from each Segment each year to fill expiring terms. Group A Directors will have their terms expire in odd numbered years, and Group B Director terms will expire in even numbered years, in conjunction with the end of the operating year of NAESB or as otherwise defined by the Board, Certificate of Incorporation or Bylaws, as amended.
- 260 f) Directors elected from the REQ may run for re-election without restriction
 261 on the number of terms held.
- g) Directors elected from the REQ may not hold both a Board seat and a seat
 on the Executive Committee at any point in time. If an Executive
 Committee member is elected as a Director from the REQ, their EC seat is
 vacated when the NAESB Board seats them as a Director.
- h) No two Directors elected by the REQ may be employees of the same
 Member holding Membership in multiple Segments within the Quadrant.
 This restriction does not prohibit election of two Directors from two
 affiliated companies within a holding company which each have individual
- 270 Member status, or from two companies with a parent-subsidiary

271 relationship, provided that the two Directors from companies with such a272 relationship represent Members of differing Segments.

273 [Example: In the LDC Segment, the Exelon Corporation might have 274 interests in multiple REQ Segments. If Exelon joins all segments as the Exelon, rather than as its operating subsidiaries doing business in those 275 276 Segments, Exelon could only have one Board member elected from the 277 REQ, period. If instead, Exelon joined each Segment via its operating 278 subsidiaries active in each Segment's portion of the industry, one or more 279 Board members representing interests within the Exelon corporate 280 structure could be elected from the REQ Segments and serve concurrently. 281 However, if mulitiple Exelon operating subsidiaries joined one Segment (Commonwealth Edison and PECO Energy joining the LDC Segment, for 282 283 example), only one Board seat could be held by representatives of that 284 Segment. Intent is to prevent one company / member from being able to 285 "load" the Board in their favor, particularly in representing the interests of 286 a single Segment. The rule, as proposed, counts on the diversity of market 287 presence between Segments to tend to limit the number of Board seats 288 affiliated with a single holding company across Segments.] 289 i) In preparation for any election of NAESB Directors (other than initial Directors, as provided for in Section 8), a Nominating Committee 290 291 consisting of five EC members of the REQ comprised of one member from 292 each Segment plus the EC Vice-Chair, shall develop a slate of candidates 293 from the Segment Membership. Other nominations may be made at or 294 prior to the close of the REQ End User Segment nomination period by any 295 Segment Member eligible to serve on the Board of Directors by submitting 296 their name(s) to the NAESB Office in a form as specified in NAESB

297 Operating Procedures if such requirements exist. All nominations must be 298 made and conveyed in writing to the NAESB Secretary no less than 31 days 299 prior to the election date.

- Election of Directors shall occur in the same month for all Segments of the 300 j) 301 REQ, and shall be coordinated by the Executive Committee. The EC will 302 ensure that the requirements of this Section are satisfied by each 303 Segment's slate of candidates prior to any actual voting by any Segment. 304 k) In addition to being subject to removal from office by the NAESB Board of 305 Directors, Directors may also be removed from office by the Quadrant 306 membership at any time upon a 75% majority vote of the Quadrant 307 membership and a simple majority (51%) of each of the Segments of the REQ. The vacant seat is to be refilled in accordance with all of the 308 309 requirements of this Section 4.1.
- 310 I) Elected Directors may not declare a Designated Alternate for their seat for
 311 any reason, and may not delegate their Board vote to another except by
 312 Proxy to another Board member.
- m) In the event that the Director changes affiliation to another Member within
 his/her industry Segment, the Director's term will continue until its natural
 expiration, provided that there is no other Director already representing
 the same Member, in which case the Director changing his or her affiliation
 will vacate their seat for election of a new Director. In the event that a
 Director is no longer affiliated with their electing industry Segment, the
 Director will vacate their seat for election of a new Director.
- 320 n) Meetings of Directors from the REQ will be in accordance with the NAESB321 Bylaws.

322

323 4.2 Executive Committee

- a) The REQ shall elect sixteen (16) NAESB Executive Committee members,
 subject to Section 8 of these Procedures, with each Segment electing four
 (4) of these EC Members, pursuant to procedures established by that
 Segment. All EC representation from the REQ will be comprised of an
 equal number of representatives elected from each Segment of the
 Quadrant.
- b) The REQ will elect four (4) EC members from each Segment, subject to the
 provisions of Section 8 of these Procedures. Nominations for and election
 to an EC seat will be in accordance with Segment Procedures.
- c) In the event that an EC member resigns or otherwise vacates their seat,
 the Quadrant will hold an election within 60 days to fill the vacant seat
 with a candidate from the same Segment as the original EC member.
- 336 d) EC members shall be elected for two-year terms, with half of the terms
 337 expiring in alternating years.
- subject to the provisions of Section 8, two EC members will be elected
 from each Segment each year to fill expiring terms. Group A EC members
 will have their terms expire in odd numbered years, and Group B EC
 member terms will expire in even number years, in conjunction with end of
 the operating year of NAESB or as otherwise defined by the Board of
 Directors, Certificate of Incorporation or Bylaws, as amended.
- f) EC members from the REQ may run for re-election without restriction on
 the number of terms held.
- g) EC members elected from the REQ may not hold both a Board seat and a
 seat on the Executive Committee at any point in time. If an Executive

348 Committee member is elected to the Board from the REQ, their EC seat is 349 vacated when the NAESB Board accepts them as a member of the Board. h) No two EC members elected by the REQ may be employees of the same 350 351 Member holding Membership in multiple Segments within the Quadrant. This restriction does not prohibit election of two EC members from two 352 353 affiliated companies within a holding company which each have individual 354 Member status, or from two companies with a parent-subsidiary 355 relationship, provided that the two EC members from companies with such a relationship represent Members of differing Segments. 356 357 i) In preparation for any election of EC Members (other than initial EC 358 Members, as provided for in Section 8), a Nominating Committee consisting of five EC members of the REQ comprised of one member from 359 360 each Segment plus the EC Vice-Chair, shall develop a slate of candidates 361 from the Segment Membership. Other nominations may be made at or 362 prior to the close of the REQ End User Segment nomination period by any 363 Segment Member eligible to serve on the EC by submitting their name(s) 364 to the NAESB Office in a form as specified in NAESB Operating Procedures 365 if such requirements exist. All nominations must be made and conveyed in 366 writing to the NAESB Secretary no less than 31 days prior to the election 367 date. Election of EC members shall occur in the same month for all Segments of 368 j) 369 the REQ, and shall be coordinated by the existing REQ Executive 370 Committee. The EC will ensure that the requirements of this Section are 371 satisfied by each Segment's slate of candidates prior to any actual voting

by any Segment.

k) EC members may be removed from office by the Quadrant membership at
any time upon a 75% majority vote of the Quadrant membership and a
simple majority (51%) of each of the Segments of the REQ. The vacant
seat is to be refilled in accordance with all of the requirements of this
Section 4.2.

I) Elected EC members may declare representation at an EC meeting by a
Designated Alternate named in accordance with Section 7 of these
Procedures, for any reason, subject to the EC participation requirements
contained in Section 10.4 of the NAESB Bylaws as applied to all Executive
Committee meetings.

383 m) In the event that the EC member changes affiliation to another Member within his/her industry Segment, the EC member's term will continue until 384 385 its natural expiration, provided that there is no other EC member already 386 representing the EC member's new affiliation, in which case the EC 387 member changing affiliation will resign their seat for election of a new EC 388 member. In the event that an EC member is no longer affiliated with their electing industry Segment, the EC member will vacate their seat and a new 389 390 EC member will be elected to fill the seat.

391 n) The REQ EC shall

(i) elevate the prior year Vice Chair of the Quadrant EC to Chair of
the REQ EC at its first meeting in the new operating year and
elect a new Vice Chair. If the Vice Chair is vacant at the time of
the first meeting of a new operating year, both a Chair and ViceChair will be elected, and

397 (ii) designate an EC Secretary responsible for production of meeting
 398 minutes per NAESB Operating Procedures. This may involve an

399			individual assigned this responsibility on a permanent basis, or
400			an agreement on how to designate a temporary Secretary for
401			the purpose of recording minutes of each meeting.
402	o)	An EC	C member shall cease to be a member of the EC upon
403		(i)	the resignation of REQ End User Segment Membership by the
404			Member whom they represent
405		(ii)	the lapse of the Segment Membership dues of the Member whom
406			they represent,
407		(iii)	the EC member's resignation, removal, or death
408		(iv)	the first day of the NAESB operating year following an election for
409			the seat held by the EC member, wherein the member was not re-
410			elected by the Segment Membership as declared by the NAESB
411			Secretary
412		(v)	receipt by the NAESB Secretary of a rescission letter indicating that
413			the individual no longer represents the Segment Member
414	p)	A vac	ancy resulting from removal of an EC member shall be filled for the
415		rema	inder of that term in accordance with the following requirements
416		(i)	Any EC seats becoming vacant shall be filled by a special election
417			subject to the following provisions
418		(ii)	If the vacancy occurs with less than 125 days remaining in the EC
419			member's term of office, the seat will remain vacant until filled by
420			election of a new EC member to a term commencing in the next
421			operating year, and a Designated Alternate will serve until
422			expiration of the term.
423		(iii)	All other vacancies will be filled by special election; such special
424			elections are to be formally noticed to the Segment Membership by

425				the Segment Vice-Chair or Chair within 15 days of when the
426				creation of the vacancy becomes known and held within 60 days of
427				the creation of the vacancy. Candidates will be designated by the
428				normal nomination process or selected by the remaining EC
429				members of the Segment from Segment Membership. Candidates
430				elected to fill the vacancy will serve the remainder of the term, and
431				may stand for re-election.
432			(iv)	Until an election is held and a new EC member is elected, a
433				Designated Alternate will serve in the open EC seat.
434				
435	4.3	EC M	eeting	S
436		a)	Meeti	ngs of EC members from the REQ will be in accordance with the
437			NAES	B Bylaws.
438		b)	The C	Quadrant EC Chair and Vice-Chair will serve in the roles defined for the
439			NAES	B EC Chair and Vice-Chair positions in the NAESB Bylaws and chair all
140				
440			Quad	rant EC meetings.
440 441		c)		rant EC meetings. e event that the EC of the REQ meets jointly with an EC of another
		c)	In the	
441		c)	In the NAES	e event that the EC of the REQ meets jointly with an EC of another
441 442		c)	In the NAES meet	e event that the EC of the REQ meets jointly with an EC of another B Quadrant, the choice of Quadrant EC Chair presiding over the joint
441 442 443		c) d)	In the NAES meet rotati	e event that the EC of the REQ meets jointly with an EC of another B Quadrant, the choice of Quadrant EC Chair presiding over the joint ing will be determined by the precedence established in the order of
441 442 443 444			In the NAES meet rotati In the	e event that the EC of the REQ meets jointly with an EC of another B Quadrant, the choice of Quadrant EC Chair presiding over the joint ing will be determined by the precedence established in the order of on of Vice-Chairs as specified in the NAESB Bylaws.
 441 442 443 444 445 			In the NAES meet rotati In the Vice-	e event that the EC of the REQ meets jointly with an EC of another B Quadrant, the choice of Quadrant EC Chair presiding over the joint ing will be determined by the precedence established in the order of on of Vice-Chairs as specified in the NAESB Bylaws.
 441 442 443 444 445 446 			In the NAES meet rotati In the Vice- Sectio	e event that the EC of the REQ meets jointly with an EC of another B Quadrant, the choice of Quadrant EC Chair presiding over the joint ing will be determined by the precedence established in the order of on of Vice-Chairs as specified in the NAESB Bylaws. e event that the provisions of 4.3(c) result in the REQ EC Chair or Chair presiding over a joint meeting, the actions taken to comply with
 441 442 443 444 445 446 447 			In the NAES meeti rotati In the Vice- Section prepa	e event that the EC of the REQ meets jointly with an EC of another B Quadrant, the choice of Quadrant EC Chair presiding over the joint ing will be determined by the precedence established in the order of on of Vice-Chairs as specified in the NAESB Bylaws. e event that the provisions of 4.3(c) result in the REQ EC Chair or Chair presiding over a joint meeting, the actions taken to comply with on 4.2(m)(ii) will determine what person is responsible for taking and

451 Committee. Executive Committee Members may participate and vote by 452 means of tele-conference or other electronic means unless in-person 453 attendance is required of all members by both the Chair and Vice-Chair of 454 the Executive Committee, and subject to the attendance requirements of 455 the NAESB Bylaws. Attendance records shall be reviewed by the Secretary 456 of the EC during the course of each Quadrant EC meeting for compliance 457 with NAESB Bylaws for the term of office served by each current Quadrant 458 EC member. The Secretary shall advise any members at risk for failure to 459 comply of their status.

- 460 f) Failure to meet the in-person attendance requirements for reasons other 461 than hospitalization or disability requires issuing a notice of non-462 compliance to the Quadrant membership and a vote for removal of the 463 Executive Committee seat within 30 days of failing to comply. Election of a 464 new member, if necessary, will proceed as provided for in Section 3. If the 465 membership does not vote for removal of the EC member, the Secretary or 466 EC Chair may call for another removal vote by the Quadrant membership 467 following the next EC meeting if the EC member's attendance record still 468 does not satisfy the Bylaws requirements.
- 469

470 4.4 EC Subcommittees and Task Forces

471 Note: The provisions of this Section 4.4 may actually be more appropriately
472 Iocated within NAESB's Operating Procedures. Until such time as there is Board
473 agreement to adopt them for all of NAESB, they remain here as REQ provisions.

474

The EC of the REQ shall set up its own subcommittees and task forces to deal withREQ-specific issues.

- 477 a) The Executive Committee may establish voluntary standing subcommittees
 478 or special purpose task forces to perform various functions required of the
 479 organization.
- 480 (i) The Executive Committee will prepare a written statement of the
 481 purpose of the subcommittee or task force and the tasks to be
 482 performed, name the subcommittee or task force, state if balanced
 483 voting among the Segments is applicable, and appoint a temporary
 484 chair.
- 485 (ii) The Temporary Chair may be a member of the Executive Committee
 486 or any REQ Member willing to perform the required startup tasks
 487 and to continue chairing the subcommittee / task force if elected by
 488 the members its first meeting.
- 489 (iii) The Temporary Chair shall set up the first meeting of the490 subcommittee or task force and prepare a meeting notice that:
- 491 1) states the name and purpose of the subcommittee / task
 492 force,
- 493 2) solicits participation in the subcommittee / task force,
- 494 3) announces the agenda for the first meeting.

495 and shall send the meeting notice to all Members and non-members who have participated in Quadrant activities in the last twelve 496 497 months. The notice shall be sent at least two weeks prior to the 498 meeting date. Shorter time periods for notices of subsequent 499 meetings will be permitted by a 75% vote of the participants 500 attending a duly scheduled meeting. All meeting notices shall be 501 posted on the NAESB website and transmitted in writing, facsimile, 502 or other electronic means to parties who have indicated an interest

503			in the duly scheduled meeting. The permanent subcommittee /
504			task force Chair may be any REQ member. Participation on
505			subcommittees or task forces is open to any interested party.
506		b)	Each EC subcommittee or task force will designate a meeting Secretary
507			responsible for recording and reporting voting results, keeping regular
508			minutes of its proceedings and providing copies of those minutes promptly
509			to the NAESB Executive Director's office in accordance to reporting criteria
510			established by the EC.
511		c)	Each EC subcommittee or task force will report to the EC at no less than a
512			quarterly basis, on a schedule to be defined by the EC for as long as the
513			subcommittee or task force continues to exist
514		d)	Each EC subcommittee / task force will strive for balanced representation
515			and balanced voting between Segments, and participation shall be open to
516			all NAESB Members and other interested parties, without regard to whether
517			the participant is a paid Member of NAESB in good standing.
518			
519	5.0	MEET	TINGS OF THE RETAIL ELECTRIC QUADRANT:
520		All m	eetings held in association with the NAESB organization or the REQ are open
521		to an	y interested person.
522			
523	5.1	Votir	ng for Adoption
524		Only	Members of the REQ EC shall have the right to vote to adopt standards or
525		mode	el business practices for the REQ. Quadrant voting procedures will be in
526		accor	dance with the NAESB Bylaws.
527			
528	5.2	Ratif	ication

529		Only	REQ Voting Members have the right to ratify standards or model business
530		pract	ices adopted by the Quadrant EC. Ratification voting will be in accordance
531		with	the requirements of the NAESB Bylaws.
532			
533	6.0	JOIN	IT MEETINGS
534		From	time to time, there will be joint meetings of the REQ with other Quadrants
535		withi	n NAESB, and Segments will meet jointly to transact Quadrant business.
536			
537	6.1	One	Person, One Vote Requirement
538		a)	Designation of the same individual as the duly authorized representative of
539			the Member in multiple Segments of the REQ (including Agent
540			representatives) is not permitted.
541		b)	In the event that a duly authorized representative in one Segment also
542			represents the Member in another Segment in any other NAESB Quadrant,
543			and both affected Segments are engaged in a joint meeting requiring
544			Member voting, then the person serving as representative in both
545			Segments must choose the Segment and Quadrant in which they will vote
546			on any given issue and may only cast one vote on each voting issue. To
547			the extent that any other involved Quadrant's rule may differ on this point,
548			the representative must still honor this Quadrant's rule for any such vote or
549			designate a substitute representative for the entity's interests during the
550			joint meeting.
551			Note: The provisions of Section 6.1(b) deal with an issue applicable to all
552			Quadrants and may be more appropriate for inclusion in either the Bylaws
553			or NAESB Operating Procedures. In addition, since nearly all votes by a
554			Member in NAESB will either be for election of officers or ratification, joint

555			action with another Quadrant requiring voting should be very rare, and
556			may never actually occur. Need to decide whether to keep or drop this.
557			
558	7.0	EC D	ESIGNATED ALTERNATES
559	7.1	EC D	esignated Alternate Authority
560		Any p	person presenting themselves at an EC meeting as a Designated Alternate will
561		be ac	cepted as a participant provided that
562		a)	An EC Member from that Segment either indicates to the NAESB Office, the
563			EC Chair or Vice-Chair that they will be absent, or is in fact absent and
564			remains absent,
565		b)	The name of the Designated Alternate is on a list of approved Designated
566			Alternates selected by the appropriate Segment membership according to
567			these Procedures and on file with both the NAESB Office and the EC.
568			
568 569	7.2	Elect	ion of Designated Alternates
	7.2	Elect a)	ion of Designated Alternates Each Segment will select no more than three (3), and no less than one
569	7.2		
569 570	7.2		Each Segment will select no more than three (3), and no less than one
569 570 571	7.2		Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for
569 570 571 572	7.2	a)	Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to twelve (12) Designated Alternates overall.
569 570 571 572 573	7.2	a)	Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to twelve (12) Designated Alternates overall. Designated Alternates will be selected annually by the same election that
 569 570 571 572 573 574 	7.2	a)	Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to twelve (12) Designated Alternates overall. Designated Alternates will be selected annually by the same election that determines new EC members, with those unsuccessful EC member
 569 570 571 572 573 574 575 	7.2	a)	Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to twelve (12) Designated Alternates overall. Designated Alternates will be selected annually by the same election that determines new EC members, with those unsuccessful EC member candidate(s) receiving the greatest number of votes becoming a
 569 570 571 572 573 574 575 576 	7.2	a)	Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to twelve (12) Designated Alternates overall. Designated Alternates will be selected annually by the same election that determines new EC members, with those unsuccessful EC member candidate(s) receiving the greatest number of votes becoming a Designated Alternate in the event that there already are or are expected to
569 570 571 572 573 574 575 576 577	7.2	a)	Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to twelve (12) Designated Alternates overall. Designated Alternates will be selected annually by the same election that determines new EC members, with those unsuccessful EC member candidate(s) receiving the greatest number of votes becoming a Designated Alternate in the event that there already are or are expected to be less than three Designated Alternates from the Segment in the coming
 569 570 571 572 573 574 575 576 577 578 	7.2	a) b)	Each Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to twelve (12) Designated Alternates overall. Designated Alternates will be selected annually by the same election that determines new EC members, with those unsuccessful EC member candidate(s) receiving the greatest number of votes becoming a Designated Alternate in the event that there already are or are expected to be less than three Designated Alternates from the Segment in the coming year.

- from the Segment are unable to attend a meeting of the EC and when the
 absent member or the Vice-Chair of the REQ EC requests their attendance
 as a Designated Alternate.
- 584

585 8.0 QUADRANT STARTUP & TRANSITION

586 The requirements of the NAESB Bylaws for a minimum of 5 Members per Segment 587 and 40 Members per Quadrant do not provide for enough members to satisfy the 588 REQ requirements for election of Directors and EC members as stated in Sections 3 and 4. Consequently, until a Segment of the Quadrant has at least eight (8) 589 590 Members, the number of Directors and EC Members elected by the Segment 591 membership will be halved, to 2 Directors and 2 EC members per Segment and 592 Segment balanced voting will remain intact by adjusting the weight of all votes to 593 achieve balance between Segments in the REQ. The following provisions will 594 apply and supercede the requirements of Sections 3 and 4 until the REQ Segment 595 achieves a minimum of eight (8) members.

596

597 8.1 Initial Election of Directors

598 Acceptance of the REQ by the Board will involve seating of the REQ Directors at 599 the NAESB Board meeting at which an acceptance vote is cast. This requires that 600 there be REQ Directors who may be seated.

601

The founding membership of the REQ will elect Directors of their choosing by means that are reasonably consistent with NAESB Bylaws and the requirements of Section 4.1 of these Procedures, subject to the conditions addressed by Section 8.2 this Section. Selection of candidates and their election will be by procedures agreed to by consensus or voting methods adopted by the group. Such elected 608 representatives of the REQ, and acceptance by the Board will place all the requirements and restrictions of these Procedures upon those individuals. 609 610 Term of Office of Initial Directors 611 8.2 612 a) In the event that Directors are elected from a Segment with at least five 613 (5), but less than eight (8) members, two Directors will be elected, each to 614 serve a 3 year term, and both will be either Group A or Group B directors 615 as determined by the expiration year of their original term. 616 b) In the event that a Segment has eight (8) or more members, four Directors 617 will be elected. The two candidates having the most votes will serve three (3) year terms. The remaining two Directors elected by lesser vote counts 618 619 will serve two year terms. 620 8.3 Interim Elections of Directors 621 622 In the event that any Segment within the Quadrant achieves a minimum 623 membership of eight Members after operating with less for a period of time after 624 an election held under the requirements of Section 8.1, then 625 a) a second election of Directors shall be held within 60 days of the addition of the eighth Member of the Segment in order to select two additional 626 627 Directors from that Segment to serve alongside the initially elected 628 Directors; the term of office of the second group elected at this time will expire one 629 b) 630 year prior to the term of the initially elected Directors, thereby allowing for two Group A and two Group B Directors from each Segment. 631 632

representatives will be presented to the NAESB Board for acceptance as

607

633

634 8.4 Initial Election of Executive Committee

- Acceptance of the REQ by the Board will involve seating of the REQ Directors at
 the NAESB Board meeting at which an acceptance vote is cast. Initial operation of
 the REQ will require that an Executive Committee be in place.
- 638

639 The founding membership of the REQ will elect an Executive Committee of their 640 choosing by means that are reasonably consistent with NAESB Bylaws and the requirements of Section 4.2 of these Procedures, subject to the conditions 641 642 addressed by Sections 8.5 of this Section. Selection of candidates and their 643 election will be by procedures agreed to by consensus or voting methods adopted by the group. Such elected representatives will be presented to the NAESB Board 644 645 for acceptance as representatives of the REQ, and acceptance by the Board will 646 place all the requirements and restrictions of these Procedures upon those 647 individuals.

648

649 8.5 Term of Office of Initial Executive Committee Members

The initial group of EC members will serve a 2 year term of office as either Group
A or Group B members, as determined by the expiration year of their original term
of office.

a) In the event that EC members are elected from a Segment with at least
five (5), but less than eight (8) members, two EC members will be elected,
each to serve a 3 year term, and both will be either Group A or Group B EC
members as determined by the expiration year of their original term. The
candidate with the greatest number of votes will be designated the EC

- 658 Chair; in the event of a tie vote, a coin toss will determine the EC Chair659 and Vice-Chair positions.
- b) In the event that a Segment has eight (8) or more members, four EC
 members will be elected. The two candidates having the most votes will
 serve a three (3) year term. The remaining two EC members elected by
 lesser vote counts will each serve a two year term. A separate election
 within the EC will determine which member becomes the Chair and which
 becomes the Vice-Chair, by vote of the four EC members.
- 666
- 667

8.6 Interim Elections of EC Members

In the event that the Segment achieves a minimum membership of eight Members
after operating with less for a period of time following an election held under the
provisions of Section 8.5, then

- a) a second election of EC members shall be held within 60 days of the
 addition of the eighth Member of the Segment in order to select two
 additional EC members from the Segment to serve alongside the initially
 elected EC members;
- b) the term of office of the second group elected at this time will expire one
 year prior to the term of the initially elected EC members, thereby allowing
 for two Group A and two Group B EC members from each Segment.
- c) The previously elected Chair and Vice-Chair of the Quadrant EC will
 continue in their official role until the end of the operating year. Chair
 rotation will proceed per the requirements of Section 4.2(m)(i).
- 681

682 8.7 Balanced Voting During The Transition Period

- 683 Any Segment which has elected Directors or EC members under the provisions of
- this section will retain the full Segment votes entitled to it under the Balanced
- 685 Voting principle of the Bylaws, and its representatives' votes will count for either
- one or two Segment votes as is appropriate to the situation.
- 687

TAB 5

PROPOSED NAESB RETAIL GAS QUADRANT PROCEDURES

1	THIS DOCUMENT IS A DRAFT PROPOSAL FOR DISCUSSION PURPOSES, THE FINAL FORM OF
2	WHICH SHOULD MATERIALIZE ONLY AFTER ALL STAKEHOLDERS AGREE ON THE
3	PROCEDURES WITHIN.
4	
5	
6	
7	NORTH AMERICAN ENERGY STANDARDS BOARD
8	BYLAWS ADDENDUM
9	EXHIBIT 3
10	RETAIL GAS QUADRANT PROCEDURES

11	
12	1 - DEFINITIONS
13	Section 1.1 Definitions Included in the NAESB Bylaws
14	Capitalized terms, when used in this Exhibit to the NAESB Bylaws, shall have the meanings set
15	forth in Section 1.1 of the NAESB Bylaws.
16	Section 1.2 Definitions for the Purposes of this Exhibit
17	The following terms have not been defined in Section 1.1. of the NAESB Bylaws and when used
18	in this Exhibit, shall have the meanings set forth below:
19	A. "RGQ Designated Alternate" is defined as a person named by a Segment of the Retail Gas
20	Quadrant Segment, submitted to the NAESB office, to serve in place of a RGQ EC Member
21	who is unable to attend an EC meeting.
22	B. "RGQ EC" means the Executive Committee of the Retail Gas Quadrant of NAESB.
23	D. "RGQ EC Subcommittee" means a subcommittee established by the Executive Committee of
24	the Retail Gas Quadrant of NAESB.
25	C. "RGQ Members" means Voting Members of the Retail Gas Quadrant of NAESB that satisfy
26	the requirements of membership set forth in Section 5.1 of this Exhibit.
27	D. "RGQ Segment" means one of the four co-equal Segments of the Retail Gas Quadrant of
28	NAESB.
29	
30	2 – PURPOSES, SCOPE, ACTIVITIES, AND POLICIES
31	Section 2.1 Purposes, Scope and Activities
32	The purpose of the Retail Gas Quadrant (RGQ) of NAESB is to propose and adopt voluntary
33	model business practices or standards to promote more competitive, efficient and reliable service in the
34	retail natural gas industry. The Retail Gas Quadrant is concerned with and tasked to handle natural gas
35	related issues and practices that are within the scope of NAESB and typically addressed at the retail

natural gas distribution level. The Retail Gas Quadrant shall work closely with other NAESB Quadrants
 to mitigate inconsistencies where proposed standards and model business practices affect those other
 Quadrants.

39 Section 2.2 Policies

The Retail Gas Quadrant shall comply with the policies and procedures laid out in the bylaws 40 and the certificate of incorporation of NAESB. Further, the Retail Gas Quadrant intends to operate in 41 conformance with the principles of the umbrella organization, as established in Section 2.2 (b) of the 42 NAESB Bylaws and in compliance with ANSI guidelines. As such, the Retail Gas Quadrant encourages 43 a widely based membership and has identified Segments, which reflect this principle of inclusiveness. 44 45 Also consistent with this approach, the Retail Gas Quadrant shall ensure that all meetings, including those of its Members, Executive Committee (EC), and Subcommittees and Task Forces, shall be open to 46 all persons and that all minutes thereof shall be available to the public. However, at the EC and Board 47 48 level, an elected representative may hold not more than one seat, representing only one RGQ Segment 49 within the RGQ.

50

51 Section 2.3 RGQ Segments

What follows is a description of the Retail Gas Quadrant organizational structure, including a
listing of Retail Gas Quadrant Segments.

	Suppliers:	Persons, other than distributors, engaged in the sale of natural gas and/or pipeline capacity, including marketers (retail, wholesale), aggregators, producers, asset managers and pipelines.
	Distributors:	Persons engaged in the regulated or publicly owned local distribution of natural gas.
	End Users:	Persons that consume natural gas, including consumer advocacy organizations and retail customer groups.
	Service Providers:	Persons that provide services to the participants in the retail natural gas industry, including equipment manufacturers, equipment and service vendors, software providers, energy consultants, and other companies not otherwise declared in any other Segment.
54	Each RGQ Segment sha	Il have the flexibility to determine its own Segment rules and procedures,
55	as long as they conform with NA	AESB Bylaws and the Retail Gas Quadrant procedures in this Exhibit and
56	to elect or select its own represent	ntatives to the NAESB Board and to the RGQ EC.
57		
58		3 - RESERVED
59		
60		4 - RESERVED
61		
		5 DCO MEMDEDS
62		5 – RGQ MEMBERS
63	Section 5. 1 Voting Members	
64	The definition of Memb	pership applies to all RGQ Segments. RGQ Members are persons with
65	legitimate business interest in th	ne retail natural gas market and which meet the definitions of one of the
66	four RGQ Segments. Upon app	lying for Voting Membership (as described in NAESB Bylaws Sections
67	1.1 and 5.1), each prospective F	RGQ Member shall declare the RGQ Segment with which they are to be
68	identified. Voting Membership	in the Retail Gas Quadrant of NAESB shall be open to all persons that
69	meet <u>all of</u> the following require	ments:

70	- The person has a legitimate business interest in the retail natural gas market (or is a
71	representative or Agent of such person).
72	- The person meets the description of one of the co-equal industry Segments identified by the
73	Retail Gas Quadrant of NAESB.
74	- The person has declared its affiliation to one, and only one, of the RGQ Segments.
75	- Through special application, which includes a recommendation by its RGQ Segment and an
76	affirmative vote of at least 50% of the RGQ EC, a trade association may join the RGQ
77	Membership if it represents an interest that is identified within an RGQ Segment yet is not
78	represented, or is underrepresented, in the RGQ Membership. The membership status of
79	such trade associations shall be considered on an annual basis and may be voted upon
80	through notational vote by the RGQ EC.
81	
82	6 – MEETINGS OF RGQ MEMBERS
83	All meetings held in association with the NAESB Retail Gas Quadrant shall be open to all
84	interested persons. Only RGQ EC Members shall have the right to vote to adopt model business
85	practices or standards affecting only the Retail Gas Quadrant. Only Voting Members of the Retail Gas
86	Quadrant have the right to ratify model business practices and standards that were adopted by the RGQ
87	EC and that affect only the Retail Gas Quadrant.
88	
89	7 – BOARD
90	Section 7.1 Board
91	The Retail Gas Quadrant shall have representatives to the NAESB Board of Directors, with each
92	RGQ Segment electing or selecting an equal number of these representatives, pursuant to the procedures
93	specified by that RGQ Segment.

94 Section 7.2 Qualifications of Directors

- 95 To be eligible to serve as representative on the NAESB Board of Directors:
- 96 The person must be a Member of the Retail Gas Quadrant;
- 97 The person must be an executive of the organization represented and have broad
 98 understanding of the natural gas industry;
- 99 The person must have a working knowledge of the NAESB process;
- The person must be willing to commit the time and resources necessary to fulfill their
 obligations as a NAESB Director and to meet the minimum threshold of participation and
 attendance established in the NAESB Bylaws [Section 9.7 (f)]; and
- The person must disclose their interest, or their employer's interest, in the natural gas
 industry and the relationship with other entities with which the employer may be affiliated.

105 Section 7.3 Number and Election of Directors

- (a) The Retail Gas Quadrant of NAESB shall be represented on the NAESB Board of Directors
 by twenty-four (24) persons who shall be elected, from time to time, as required by Article 7
 of the NAESB Bylaws and in this Exhibit to those Bylaws.
- (b) In preparation for an election of NAESB Directors representing the Retail Gas Quadrant, a
 slate of candidates shall be solicited and developed from interested RGQ Members, and
 individual RGQ Segments shall specify a point of contact that shall provide the list of
 nominees to the NAESB office.
- (c) The NAESB office shall coordinate the election process for Retail Gas Quadrant
 representatives to the NAESB Board during the second week of November of each year. The
 elections within the RGQ Segment Gas for representatives to the NAESB Board shall be
 subject to the following provisions:
- Any RGQ Member who is current in the payment of its dues is eligible to vote.
- RGQ Members shall cast votes for candidates in a number not exceeding the number of

119 vacant directorships.

- No RGQ Member shall cast more than one vote per vacant directorship.

- 121 The candidates receiving the greatest numbers of votes shall be elected.
- 122 Section 7.4 Term of Office

The initial Retail Gas Quadrant Directors on the NAESB Board shall be divided into two groups within each RGQ Segment. Group A Directors shall serve for an initial term of three years, while Group B Directors shall serve for an initial term of two years. Each RGQ Segment shall have an equal number of Group A Directors and an equal number of Group B Directors. Upon the completion of the initial terms of both groups of Directors, succeeding Directors shall thereafter be elected for terms of three (3) years, consistent with NAESB Bylaws.

129 Section 7.5 Vacancies

A NAESB Director shall cease to be a Director upon 1) the resignation of the RGQ Membership of the entity of which the Director is a partner, director, officer, or agent; 2) the lapse of the RGQ Membership of the entity of which the Director is a partner, director, officer, agent or employee, through delinquent NAESB dues for the RGQ Membership; or (3) the Director's resignation, removal, or death. A Director vacancy shall be filled for the remainder of that term in accordance with the procedures specified in Section 7.3 of this Exhibit.

136 Section 7.6 Removal of Directors

Each RGQ Segment shall have the authority to remove a Director for cause. Prior to voting on such resolution to remove a Director for cause, the RGQ Segment shall give the Director at least 30-day notice of the proposed action and an opportunity to respond. A simple majority of the RGQ Segment Membership shall be required to remove a Director.

141 Section 7.7 Resignations

A Director may resign his or her directorship by submitting a letter to the Secretary of NAESB,
stating that he or she is resigning and giving the effective date of the resignation.

144	
145	8 – RESERVED
146	
147	9 - RESERVED
148	10 - EXECUTIVE COMMITTEE
149	Section 10.1 Duties and Responsibilities
150	The Retail Gas Quadrant shall have an Executive Committee (EC), which shall also participate in
151	the larger NAESB EC body, consisting of representatives from each RGQ Segment, with each RGQ
152	Segment electing an equal number of these RGQ EC Members, pursuant to procedures established in
153	Section 10.2 of this Exhibit. No substitutes are permitted to vote at Retail Gas Quadrant meetings;
154	however, an RGQ Designated Alternate may vote at EC meetings in place of an absent RGQ EC Member
155	from a given RGQ Segment. An RGQ Designated Alternate holds identical voting rights as the RGQ EC
156	Member in whose place he or she serves, except for those matters on which the RGQ EC Member has
157	already voted by notational ballot prior to the beginning of the meeting.
158	Section 10.2 RGQ EC Members
159	(a) <u>Number and Election of RGQ EC Members</u> : The Retail Gas Quadrant of NAESB shall be
160	represented on the RGQ's EC by twenty-four (24) persons who shall be elected, from time to
161	time, as required by Article 10 of the NAESB Bylaws and in this Exhibit to those Bylaws. In
162	preparation for an election of RGQ EC Members, a slate of candidates shall be solicited and
163	developed from interested RGQ Members, and individual RGQ Segments shall specify a
164	point of contact that shall provide the list of nominees. The NAESB office shall coordinate
165	the election process for RGQ EC Members during the second week of November of each
166	year. The elections of RGQ EC Members shall be subject to the following provisions:
167	- Any RGQ Member who is current in the payment of its dues is eligible to vote.

- RGQ Members shall cast votes for candidates in a number not exceeding the number of
 vacant EC seats.
- No RGQ Member shall cast more than one vote per vacant EC seat.
- The candidates receiving the greatest numbers of votes shall be elected.
- (b) <u>Term of Office</u>: The initial RGQ EC Members shall be divided into two groups within each
 Segment. Group A RGQ EC Members shall serve for an initial term of three years, and
 Group B RGQ EC Members shall serve for an initial term of two years. Each RGQ Segment
 shall have an equal number of Group A RGQ EC Members and an equal number of Group B
 RGQ EC Members. Upon the completion of the initial terms of both groups of RGQ EC
 Members, succeeding RGQ EC Members shall thereafter be elected for terms of three years,
- 178 consistent with NAESB Bylaws.
- 179 (c) <u>Qualifications of RGQ EC Members</u>: To be eligible to serve as a RGQ EC Member:
- The person must have broad understanding of the natural gas industry and be at least a
 senior functional manager of the organization represented;
- The person must have a working knowledge of the NAESB process;
- The person must be willing to commit the time and resources necessary to fulfill their
 obligations as a RGQ EC Member and to meet the minimum threshold of participation
 and attendance established in the NAESB Bylaws [Section 9.7 (f)]; and
- The person must disclose their interest, or their employer's interest, in the natural gas
 industry and the relationship with other entities with which the employer may be
 affiliated.
- (d) <u>Vacancies</u>: Vacancies in the RGQ EC may occur upon 1) the resignation of the RGQ
 Membership of the entity of which the RGQ EC Member is a partner, director, officer, or
 agent; 2) the lapse of the RGQ Membership of the entity of which the RGQ EC Member is a
 partner, director, officer, agent or employee, through a delinquency in payment of NAESB

- dues for the RGQ Membership; or (3) the RGQ EC Member's resignation, removal, or death.
 An RGQ EC Member vacancy shall be filled for the remainder of that term in accordance
 with the procedures specified in Section 10 of this Exhibit.
- (e) <u>Removal of RGQ EC Members</u>: Each RGQ Segment shall have the authority to remove an RGQ EC Member for cause. Prior to voting on such resolution to remove an RGQ EC
 Member for cause, the RGQ Segment shall give the RGQ EC Member at least 30-day notice of the proposed action and the opportunity to respond. A simple majority of the RGQ Segment Membership shall be required to remove an RGQ EC Member.
- (f) <u>Resignations</u>: A RGQ EC Member may resign from the RGQ EC by submitting a letter to
 the Secretary of NAESB, stating that he or she is resigning and giving the effective date of
 the resignation.
- 204 Section 10.3 Reserved
- 205 Section 10.4 Reserved

206 Section 10.5 RGQ EC Subcommittees

While there may be Subcommittees and Task Forces established by the NAESB EC, to be comprised of NAESB Members and other interested parties, the Retail Gas Quadrant shall set up its own Subcommittees and Task Forces to deal with Retail Gas Quadrant-specific issues. Each RGQ EC Subcommittee shall report to the RGQ EC and each shall:

- (a) elect a chair who shall be an RGQ EC Member and who shall serve until removed by the
 RGQ Subcommittee's membership;
- (b) carry out its work in accordance with procedures adopted by the NAESB EC for EC
 Subcommittees;
- 215 (c) provide notice of meetings and agendas;
- 216 (d) practice balanced voting and record voting results; and

217

218

(e) keep regular minutes of its proceedings and provide copies of these minutes promptly to the NAESB office.

Any task force established by RGQ EC Subcommittees shall be open to all NAESB Members and other interested parties. At NAESB joint Subcommittee or joint Task Force meetings, the same individual may represent different Segments from different NAESB Quadrants, as long as that individual declares prior to the joint meeting the NAESB Quadrant and Segment for which he or she is casting a vote. At RGQ Subcommittee or Task Force meetings, the same individual may represent different RGQ Segments, as long as that individual declares prior to the meeting the RGQ Segment for which he or she is casting a vote.

226 Section 10.6 RGQ Designated EC Alternates

Each RGQ Segment shall elect a number RGQ Designated EC Alternates to ensure that the business of the EC is conducted uninterruptedly. The role and duties of a RGQ Designated EC Alternate are as follows:

- (a) When requested by an RGQ EC Member to attend an EC meeting, an RGQ Designated EC
 Alternate shall do so unless he or she is unable to attend, in which case the RGQ Designated
 EC Alternate shall so advise the RGQ EC Member, and the RGQ EC Member shall select
 another RGQ Designated EC Alternate.
- (b) When attending an EC meeting, the RGQ Designated EC Alternate shall have the full voting
 rights of the RGQ EC Member, except as may be limited by that RGQ EC Member's notice
 in writing to the Chair of the EC, the Secretary of NAESB, and the RGQ Designated EC
 Alternate.
- (c) An RGQ Designated EC Alternate is required to attend at least 50% of all meetings for
 which they are requested to attend with a least one week's notice; however, the attendance
 requirement would not exceed 50% of all EC meetings scheduled annually.
- 241 The same eligibility criteria that are required of RGQ EC Members under Section 10.2 (c) of the

Retail Gas Quadrant Procedures also apply to elected RGQ Designated EC Alternates. Following the election of RGQ EC Members, each RGQ Segment will elect not more than eight (8) RGQ Designated EC Alternates to attend and vote at RGQ or NAESB EC meetings, when one, or more, regular RGQ EC Member is unable to attend a particular EC meeting.

In preparation for an election of RGQ Designated EC Alternates, a slate of candidates shall be solicited and developed from interested RGQ Members, and individual RGQ Segments shall specify a point of contact that shall provide the list of nominees to the NAESB office. The elected RGQ Designated EC Alternates for each RGQ Segment shall be announced to the NAESB Office as soon as possible after the election is complete. This written communication shall include the names, addresses, telephone numbers, facsimile numbers, and e-mail addresses of RGQ Designated EC Alternates for each RGQ Segment.

253 Once an RGQ EC Member obtains an agreement from an RGQ Designated EC Alternate to 254 substitute for them at an EC meeting, this designation shall be communicated in writing to the NAESB 255 office at least one day prior to the EC meeting.

- 256
- 257
 258
 259
 260
 261
 13 RESERVED
 262
 263
 14 RESERVED
 - 264

265	15 – RESERVED
266	
267	16 – RESERVED
268	
269	17 – RESERVED
270	
271	18 - AMENDMENTS
272	Section 18.1 Amendments
273	In order for Retail Gas Quadrant Procedures to be amended, upon petition of at least five (5)
274	RGQ Members, the Vice Chair of the EC for the RGQ shall announce an RGQ meeting. Such
275	announcement shall provide for at least 30-days notice. In order to transact business at the RGQ meeting,
276	there shall be a quorum consisting of at least 1/3 of the RGQ Membership. Following such meeting, the
277	proposed resolution adopted at the meeting shall be sent out for comment, and the comments shall be
278	distributed to all RGQ Members in advance of a notational vote. Any RGQ Member not choosing to vote
279	shall be considered to have voted in favor of the proposed change. In order for a proposed change to take
280	effect, it must be approved by at least 2/3 of RGQ Members and 40% of each RGQ Segment's
281	Membership.
282	

283

19 - RESERVED

284 285 **RETAIL GAS QUADRANT SUPPLIERS SEGMENT PROCEDURES** 286 [To be provided by the RGQ Suppliers Segment]

287 **RETAIL GAS QUADRANT DISTRIBUTORS SEGMENT PROCEDURES**

[To be provided by the RGQ Distributors Segment]

288

289 **RETAIL GAS QUADRANT END USER SEGMENTS PROCEDURES**

290

[To be provided by the RGQ End Users Segment]

291 **RETAIL GAS QUADRANT SERVICE PROVIDERS SEGMENT PROCEDURES**

292 [To be provided by the RGQ Service Providers Segment]
Summary of Revisions to NAESB RGQ Procedures 1-25-02

- 1. Followed the NAESB Bylaws organization. Accordingly various sections and paragraphs have been moved within the document.
- 2. Tagged sections taken from the NAESB Bylaws that have no corollary in the RGQ with the term "RESERVED"
- *3. Updated the section numbers in references within the document to other sections.*
- 4. Deleted all terms in Section 1.1, since most are defined in the NAESB Bylaws and added a reference to the Section in the Bylaws.
- 5. For those terms not defined in the Bylaws, added Section 1.2 to include those terms specific to Exhibit 3. They are "GRG Designated EC Alternate," "RGQ EC," "RGQ EC Subcommittee," "RGQ Members," and "RGQ Segment."
- 6. *Replaced terms in the document to reflect those new terms listed under #5.*
- 7. *Reviewed document to ensure that all terms that have been defined in the Bylaws or in this Exhibit are capitalized consistently throughout the document—when appropriate to the context.*
- 8. Moved Section 6.0 ("Retail Gas Quadrant Industry Segments") to Section 2.3
- 9. Replaced the term "substantial interest" with "legitimate business interest."
- 10. In Section 5.1 (line 76), inserted "at least 50%" for the minimum threshold of votes required to approve a trade association as an RGQ Member.
- 11. Corrected the couple of instances where the term "representative(s) to the EC" was used instead of REQ EC Member(s). Retained the usage of the term "representatives to the NAESB Board."
- 12. Inserted a new section (Section 10.6 Designated EC Alternates), which was moved from the RGQ Distributor Segment document to the Retail Gas Quadrant document.
- 13. Changed the number of representatives from the Retail Gas Quadrant on its EC and on the NAESB Board from 16 to 24.

Draft RGQ Procedures 01-04-02	Wisconsin Public Service Work Paper 2 (01-22-02)	DRAFT RGQ Procedures 01-25-02
Introduction lines 14-18	Section 2.1 Purposes Scope and Activities, lines 41-56	Section 2.1 Purposes Scope, and Activities, lines
		47-51
Introduction lines 19-20	2.1, 48-50	2.1, 51-52
Introduction lines 20-28	2.1, 48-59	2.2, 54-61, reorganized sentences
1.0 Definitions	1.0 Definitions	1.1 Definitions Included in the NAESB Bylaws
1.2, 1.4, 1.5, 1.6 (definitions)	1.2, 1.4, 1,5, 1.6	1.1 D, F, C, M
1.2 & 1.3 (definitions)	1.2 & 1.2	1.2 Definitions for the Purposes of this Exhibit
		(A & B)
		1.1 H, K, O [new definitions as applicable to
		RGQ]
1.7 (definition)	1.7	Deleted [No longer necessary]
2.0 Retail Gas Quadrant Membership	5.1 Voting Members, 119-140	5.1 Voting Members, 93-111
lines 43-60		
3.0 Organization of the Retail Gas Quadrant, lines 62-	2.1, 61-64	Deleted [Language repeated elsewhere]
64		
3.0, lines 64-66	2.3 Quadrants and Segments, 86-87	2.3 Quadrants and Segments, 80-81
3.0, lines 66-79	2.3, 103-106	2.3, 82-85
3.1 Retail Gas Quadrant Representatives to the NAESB	7.1 Board, 156-157	7.1 Board, 132-134
Board of Directors, 71-73		
3.1, 73-79	7.1 158-165	7.4 Term of Office, 169-174
3.2 Retail Gas Quadrant Executive Committee, 81-83	10.1 Duties and Responsibilities, 249-251	Deleted [Language repeated elsewhere]
3.2, 83-89	10.1, 252-258	10.2 EC Members: (b) Term of Office, 240-246
3.2, 89-94	10.1, 258-263	10.1 Duties and Responsibilities, 219-223
3.3 Retail Gas Quadrant EC Subcommittee and Task	10.5 EC Subcommittees, 324-340	10.5 EC Subcommittees, 278-291
Forces, 96-109		
4.1 Election of Directors: (a), 113-115	7.3 Number and Election of Directors: (a) 184-186	7.3 Number and Election of Directors: (a) 150- 152
4.1 (a), 115-117	7.4 Term of Office, 207-208	Deleted [Language repeated elsewhere]
4.1 (a), 117-131	7.2 Qualifications of Directors, 169-180	7.2 Qualifications of Directors, 136-148
4.1 (b), 131-134	7.3 (b), 188-192	7.3 (b) 154-157 [Edited]
4.1 (c), 135-144	7.3 (c), 194-203	7.3 (c), 158-167 [Edited & parts combined with
		7.3 (b) to avoid repetitive language]

4.2 Resignation of Directors	7.7 Resignations, 230-232	7.7 Resignations, 188-189
4.3 Vacancies Other Than Expirations, 149-155	7.5 Vacancies, 213-219	7.5 Vacancies, 176-181
4.4 Removal of Directors, 157-160	7.6 Removal of Directors, 223-226	7.6 Removal of Directors, 183-186
5.1 Election of Retail Gas Quadrant EC Members (a),	10.2 EC Members (a), 267-269	10.2 EC Members: (a) Number and Election of
165-167		EC members, 225-228
5.1. (a), 167-169	10.2 (a), 269-271	Deleted [Language repeated elsewhere]
5.1 (a), 169-181	10.2 (a), 271-182	10.2 (c) Qualifications of EC Members, 247-259
5.2 (b), 182-185	10.2 (b), 284-287	10.2 (a) 228-230
5.2 (c), 186-195	10.2 (c), 289-297	10.2 (a) 231-239 [Edited to avoid repetitive
		language]
5.2 Resignation of Retail Gas Quadrant EC Members,	10.2 (d), 299-301	10.2 (f)Resignations, 272-274
197-199		
5.3 Vacancies Other than Expirations, 201-207	10.2 (e), 303-310	10.2 (d) Vacancies, 260-266 [First sentence
		edited for clarity.]
5.4 Removal of Retail Gas Quadrant EC Members,	10.2 (f), 312-316	10.2 (e), 267-271
209-213		
6.0 Retail Gas Quadrant Industry Segments, Table, line	2.3 Quadrants and Segments, 89-101	2.3 Quadrants and Segments, Table, line 81
214		
7.0 Meetings of the Retail Gas Quadrant, 216-217	2.2 Policies, 68-69	6 – Meetings of Members [edited a second
		clause to the sentence referencing policies in
		Section 2.2]
7.0, 217-218	2.2, 69-71	2.2 Policies: (a) Voting to Adopt Standards, 62-
		63
7.0, 218-220	2.2, 71-73	2.2 Policies: (b) Ratification of Standards, 64-66
8.0 Joint Meetings of the Retail Gas and Other	2.2, 75-78	2.2 Policies: (c) Quadrant and Segment
Quadrants, 222-224		Representation at Joint Subcommittee Meetings,
		67-70
8.0, 224-227	2.2, 78-81	2.2 Policies: (d) Segment Representation at
		Retail Gas Quadrant Subcommittee Meetings,
		71-74
8.0, 227-228	2.2, 81-82	2.2 Policies: (e) Representation on the NAESB
		Board and the Retail Gas Quadrant Executive
		Committee, 75-77
9.0 Amendments to Retail Gas Quadrant Procedures,	18.0 Amendments, 366-376	18.1 Amendments, 326-334
230-239		

Distributor Segment Procedures Document:				
3.1 Qualifications of Designated EC Alternates, 29-30	Not included in this document	20.2 Qualifications of Designated Alternates, 355-356		
3.2 Election of Designated Alternates (a), 32-35	20.1 Selection of Designated Alternates [The section is different than the provisions in the Distributor Segment document. e.g. "no more than 10 Designated Alternates"], 386-389	20.3 (a) Designated EC Alternates [Inserted a general introductory sentence regarding the need for all segments to have Designated Alternates], 358-361		
3.2 (b), 36-39	20.1 (b) 391-396 [This section is different than what the original in the RGQ Distributor Segment document; I believe this section was extracted from the REQ Procedures]	20.3 Number and Election (b), 362-365		
3.2 (c), 40-43	20.1 (e), lines 412-414 [This section is different than what the original in the RGQ Distributor Segment document; I believe this section was extracted from the REQ Procedures]	20.3 (c), 366-369		
3.2 (d), 44-47	20.1 (c), 399-402 [This section is different than what the original in the RGQ Distributor Segment document; I believe this section was extracted from the REQ Procedures]	20.3 (d), 370-372		
3.3 Duties of Designated Alternates (a), 49-52	20.1 (d), 404-407	20.1 Designated EC Alternates (a), 345-347		
3.3 (b), 53-56	20.1 (d), 407-410 [This section is different and extracted from the REQ document; the RGQ does not have a Quadrant Chair]	20.1 (b), 348-350		
3.3 (b), 56-58	Not included in this document	20.1 (c), 351-353		
4.0 Amendment of Segment Procedures, 61-62	Not included in this document			
	Not included in this document			
	20.2 Resignation of Designated Alternates [Added to REQ Procedures]			
	20.3 Removal [Added to REQ Procedures]			
	20.4 Vacancy [Added to REQ Procedures]			

1 2	THIS DOCUMENT IS A DRAFT PROPOSAL FOR DISCUSSION PURPOSES, THE FINAL FORM OF WHICH SHOULD MATERIALIZE ONLY AFTER ALL STAKEHOLDERS AGREE ON
3	THE PROCEDURES WITHIN.
4	
5	
6	
7	
8	
9	NORTH AMERICAN ENERGY STANDARDS BOARD
10	BYLAWS ADDENDUM
11	EXHIBIT 3
12	RETAIL GAS QUADRANT PROCEDURES

13 THE RETAIL GAS QUADRANT

The purpose of the Retail Gas Quadrant of NAESB is to propose and adopt voluntary model business practices or standards to promote more competitive, efficient and reliable service in the retail natural gas industry. The Retail Gas Quadrant is concerned with and tasked to handle natural gas related issues and practices that are within the scope of NAESB and typically addressed at the retail natural gas distribution level.

19 The Retail Gas Quadrant shall work closely with other NAESB Quadrants to mitigate inconsistencies where proposed standards and model business practices affect those other Quadrants. The Retail Gas 20 Quadrant shall comply with the policies and procedures laid out in the bylaws and the certificate of 21 incorporation of NAESB. As such, the Retail Gas Quadrant encourages a widely based membership and 22 has identified Segments, which reflect this principle of inclusiveness. Also consistent with this approach, 23 the Retail Gas Quadrant shall ensure that all meetings, including those of its Members, Executive 24 25 Committee (EC), and Subcommittees and Task Forces, shall be open to all persons and that all minutes hereof shall be available to the public. Further, the Retail Gas Quadrant intends to operate in 26 conformance with the principles of the umbrella organization, as established in Section 2.2 (b) of the 27 28 NAESB Bylaws, and in compliance with ANSI guidelines.

29

30 1.0 DEFINITIONS:

31 1.1 "Board" means the Board of Directors of NAESB.

32 1.2 "Bylaws" means the corporate bylaws of NAESB as amended from time to time.

- 1.3 "Designated Alternate" is defined as a person named by a Segment in a list, submitted to the
 NAESB office, to serve in place of an EC Member who is unable to attend an EC meeting.
- 35 1.4 "Director" means an individual serving on the NAESB Board.
- 36 1.5 Unless otherwise noted, the "EC" means the Executive Committee of the Retail Gas
 37 Quadrant.

Proposed NAESB Retail Gas Quadrant Procedures – PRELIMINARY DRAFT 1-4-02

38 1.6 "NAESB" means the North American Energy Standards Board, Inc.

- 39 1.7 All other capitalized terms, if not defined in this Section 1.0, shall have the same definitions
 40 as specified in the bylaws or certificate of incorporation of NAESB.
- 41

42 2.0 RETAIL GAS QUADRANT MEMBERSHIP REQUIREMENTS:

The definition of Membership applies to all Segments within the Retail Gas Quadrant. Members of the Retail Gas Quadrant are persons with significant interest in the retail natural gas market and which meet the definitions of one of the four Segments. Upon applying for Voting Membership (as described in NAESB Bylaws Sections 1.1 and 5.1), each prospective Member shall declare the Segment with which they are to be identified. Voting Membership in the Retail Gas Quadrant of NAESB shall be open to all persons that meet <u>all of</u> the following requirements:

- The person has a substantial interest in the retail natural gas market (or is a representative or
 Agent of such person).
- The person meets the description of one of the co-equal industry Segments identified by the
 Retail Gas Quadrant of NAESB.
- The person has declared its affiliation to one, and only one, of the Segments of the Retail
 Gas Quadrant.
- 55 Through special application, which includes a recommendation by its Segment and an 56 affirmative vote of _____% of the Quadrant EC, a trade association may join the Retail Gas 57 Quadrant Membership if it represents an interest that is identified within a Segment yet is not 58 represented, or is underrepresented, in the Quadrant Membership. The membership status of 59 such trade associations shall be considered on an annual basis and may be voted upon 50 through notational vote by the Quadrant EC.

Proposed NAESB Retail Gas Quadrant Procedures - PRELIMINARY DRAFT 1-4-02

61 3.0 ORGANIZATION OF THE RETAIL GAS QUADRANT:

Each Segment of the Retail Gas Quadrant shall elect representatives from the Quadrant to the 62 NAESB Board of Directors (Board). Also the Retail Gas Quadrant shall have an Executive 63 Committee (EC), which shall also participate in the larger NAESB EC body. What follows is a 64 description of the Retail Gas Quadrant organizational structure, including a listing of Retail Gas 65 66 Quadrant Segments. Each Segment within the Retail Gas Quadrant shall have the flexibility to determine its own Segment rules and procedures, as long as they conform with NAESB Bylaws 67 and Retail Gas Quadrant procedures and to elect or select its own representatives to the NAESB 68 Board and the Retail Gas Quadrant EC. 69

70 3.1 <u>Retail Gas Quadrant Representatives to the NAESB Board of Directors:</u>

71 The Retail Gas Quadrant shall have representatives to the NAESB Board of Directors, with 72 each Segment electing or selecting an equal number of these representatives, pursuant to the 73 procedures specified by that Segment. The initial Retail Gas Quadrant Directors on the 74 NAESB Board shall be divided into two groups within each Segment. Group A Directors shall serve for an initial term of three years, while Group B Directors shall serve for an initial 75 76 term of two years. Each Segment shall have an equal number of Group A Directors and an equal number of Group B Directors. Upon the completion of the initial terms of both groups 77 of Directors, succeeding Directors shall thereafter be elected for terms of three years, 78 consistent with NAESB Bylaws. 79

80 3.2 <u>Retail Gas Quadrant Executive Committee</u>:

The Retail Gas Quadrant shall have an Executive Committee, consisting of representatives from each Segment, with each Segment electing an equal number of these EC Members, pursuant to procedures established in Section 5. The initial EC Members of the Retail Gas Quadrant shall be divided into two groups within each Segment. Group A EC Members shall

Exhibit 3 - Prepared for the Retail Gas Quadrant's Consideration - Page 15

Proposed NAESB Retail Gas Quadrant Procedures - PRELIMINARY DRAFT 1-4-02

85	serve for an initial term of three years, and Group B EC Members shall serve for an initial
86	term of two years. Each Segment of the Retail Gas Quadrant shall have an equal number of
87	Group A EC Members and an equal number of Group B EC Members. Upon the completion
88	of the initial terms of both groups of EC Members, succeeding EC Members shall thereafter
89	be elected for terms of three years, consistent with NAESB Bylaws. No substitutes are
90	permitted to vote at Retail Gas Quadrant meetings; however, a Designated Alternate may
91	vote at EC meetings in place of an absent EC Member from a given Segment. A Designated
92	Alternate holds identical voting rights as the EC Member in whose place he or she serves,
93	except for those matters on which the EC Member has already voted by notational ballot
94	prior to the beginning of the meeting.
95	3.3 Retail Gas Quadrant EC Subcommittees and Task Forces:
96	While there may be Subcommittees and Task Forces established by the NAESB EC, to be
97	comprised of NAESB Members and other interested parties, the Retail Gas Quadrant shall
98	set up its own Subcommittees and Task Forces to deal with Retail Gas Quadrant-specific
99	issues. Each EC Subcommittee shall report to the EC and each shall:
100	a. elect a chair who shall be an EC Member and who shall serve until removed by the
101	Subcommittee's membership;
102	b. carry out its work in accordance with procedures adopted by the NAESB EC for EC
103	Subcommittees;
104	c. provide notice of meetings and agendas;
105	d. practice balanced voting and record voting results; and
106	e. keep regular minutes of its proceedings and provide copies of these minutes promptly to
107	the NAESB office.
108	Any task force established by EC Subcommittees shall be open to all NAESB Members and
109	other interested parties.

Exhibit 3 - Prepared for the Retail Gas Quadrant's Consideration - Page 15

4.0 ELECTION, RESIGNATION, AND REMOVAL OF NAESB DIRECTORS FROM THE RETAIL GAS QUADRANT:

112 4.1 <u>Election of Directors</u>:

- 113a. The Retail Gas Quadrant of NAESB shall be represented on the NAESB Board of114Directors by sixteen (16) persons who shall be elected, from time to time, as required by115Article 7 of the NAESB Bylaws and in this Exhibit to those Bylaws. After the initial116Directors terms are completed, subsequent elected Directors from the Retail Gas117Quadrant shall serve a term of three (3) years on the NAESB Board. To be eligible to118serve as representative on the NAESB Board of Directors:
- The person must be a Member of the Retail Gas Quadrant;
- The person must be an executive of the organization represented and have broad
 understanding of the natural gas industry;
- The person must have a working knowledge of the NAESB process;
- The person must be willing to commit the time and resources necessary to fulfill
 their obligations as a NAESB Director and to meet the minimum threshold of
 participation and attendance established in the NAESB Bylaws [Section 9.7 (f)];
- The person must disclose their interest, or their employer's interest, in the natural gas
 industry and the relationship with other entities with which the employer may be
 affiliated; and
- The individual serving on the NAESB Board shall not simultaneously serve on the
 Retail Gas Quadrant EC. [DISCUSSION POINT]
- b. In preparation for an election of NAESB Directors representing the Retail Gas Quadrant,
 a slate of candidates shall be solicited and developed from interested Quadrant Members.
 The NAESB office shall administer the elections, and individual Segments shall specify
 a point of contact that shall provide the list of nominees.

Proposed NAESB Retail Gas Quadrant Procedures - PRELIMINARY DRAFT 1-4-02

- c. The NAESB office shall coordinate the election process for Retail Gas Ouadrant 135 representatives to the NAESB Board during the second week of November of each year. 136 137 The elections, within the Segment, for Retail Gas Quadrant representatives to the NAESB Board shall be subject to the following provisions: 138 Any Member of the Retail Gas Quadrant who is current in the payment of its dues is 139 eligible to vote. 140 Retail Gas Quadrant Members shall cast votes for candidates in a number not 141 exceeding the number of vacant directorships. 142 No Quadrant Member shall cast more than one vote per vacant directorship. 143 144 The candidates receiving the greatest numbers of votes shall be elected. 4.2 Resignation of Directors: 145 A Director may resign his or her directorship by submitting a letter to the Secretary of 146 147 NAESB, stating that he or she is resigning and giving the effective date of the resignation. 148 4.3 Vacancies Other than Expiration: A NAESB Director shall cease to be a Director upon 1) the resignation of the Retail Gas 149 Quadrant Membership of the entity of which the Director is a partner, director, officer, or 150 agent; 2) the lapse of the Quadrant Membership of the entity of which the Director is a 151 partner, director, officer, agent or employee, through delinquent NAESB dues for the 152
- Quadrant Membership; or (3) the Director's resignation, removal, or death. A Director vacancy shall be filled for the remainder of that term in accordance with the procedures specified in Section 4.1.
- 156 4.4 Removal of Directors:

Each Segment shall have the authority to remove a Director for cause. Prior to voting on such resolution to remove a Director for cause, the Segment shall give the Director at least 30-day notice of the proposed action and an opportunity to respond. A simple majority of

Exhibit 3 - Prepared for the Retail Gas Quadrant's Consideration - Páge 15

160 the Segment Membership shall be required to remove a Director. 161 ELECTION, RESIGNATION, AND REMOVAL OF RETAIL GAS QUADRANT 162 5.0 **EXECUTIVE COMMITTEE MEMBERS:** 163 5.1 Election of Retail Gas Quadrant EC Members: 164 The Retail Gas Quadrant of NAESB shall be represented on the Quadrant's EC by 165 a. sixteen (16) persons who shall be elected, from time to time, as required by Article 10 of 166 the NAESB Bylaws and in this Exhibit to those Bylaws. After the initial terms of EC 167 Members are completed, Retail Gas Quadrant EC Members shall serve a term of three 168 169 (3) years. To be eligible to serve as a representative on the Retail Gas Quadrant EC: The person must have broad understanding of the natural gas industry and be at least 170 a senior functional manager of the organization represented; 171 172 The person must have a working knowledge of the NAESB process; The person must be willing to commit the time and resources necessary to fulfill 173 _ their obligations as a Retail Gas Quadrant EC Member and to meet the minimum 174 threshold of participation and attendance established in the NAESB Bylaws [Section 175 176 9.7 (f)]; The person must disclose their interest, or their employer's interest, in the natural gas 177 industry and the relationship with other entities with which the employer may be 178 179 affiliated; and The individual serving on the Retail Gas Quadrant EC shall not simultaneously serve 180 on the NAESB Board. [DISCUSSION POINT] 181 b. In preparation for an election of Retail Gas Quadrant EC Members, a slate of candidates 182 183 shall be solicited and developed from interested Quadrant Members. The NAESB office 184 shall administer the elections, and individual Segments shall specify a point of contact

Proposed NAESB Retail Gas Quadrant Procedures - PRELIMINARY DRAFT 1-4-02

c. The NAESB office shall coordinate the election process for Retail Gas Quadrant EC

representatives during the second week of November of each year. The elections for

representatives to the Retail Gas Quadrant EC shall be subject to the following

that shall provide the list of nominees.

185

186

187

188

189	provisions:
190	- Any Member of Retail Gas Quadrant who is current in the payment of its dues is
191	eligible to vote.
192	- Retail Gas Quadrant Members shall cast votes for candidates in a number not
193	exceeding the number of vacant EC seats.
194	- No Quadrant Member shall cast more than one vote per vacant EC seat.
195	- The candidates receiving the greatest numbers of votes shall be elected.
196	5.2 Resignation of Retail Gas Quadrant EC Members:
197	A Retail Gas Quadrant EC Member may resign from the EC by submitting a letter to the
198	Secretary of NAESB, stating that he or she is resigning and giving the effective date of the
199	resignation.
200	5.3 <u>Vacancies Other than Expiration</u> :
201	A Retail Gas Quadrant EC Member shall cease to be an EC Member upon 1) the resignation
202	of the Retail Gas Quadrant Membership of the entity of which the EC Member is a partner,
203	director, officer, or agent; 2) the lapse of the Quadrant Membership of the entity of which the
204	EC representative is a partner, director, officer, agent or employee, through a delinquency in
205	payment of NAESB dues for the Quadrant Membership; or (3) the EC Member's resignation,
206	removal, or death. An EC Member vacancy shall be filled for the remainder of that term in
207	accordance with the procedures specified in Section 5.1.
208	5.4 <u>Removal of Retail Gas Quadrant EC Members</u> :
209	Each Segment shall have the authority to remove an EC Member for cause. Prior to voting

Exhibit 3 - Prepared for the Retail Gas Quadrant's Consideration - Page 15

Proposed NAESB Retail Gas Quadrant Procedures – PRELIMINARY DRAFT 1-4-02

210 on such resolution to remove an EC Member for cause, the Segment shall give the EC 211 Member at least 30-day notice of the proposed action and the opportunity to respond. A 212 simple majority of the Segment Membership shall be required to remove an EC Member.

213

214 6.0 **RETAIL GAS QUADRANT INDUSTRY SEGMENTS**:

<u>Suppliers:</u> Persons, other than distributors, engaged in the sale of natural gas and/or pipeline capacity, including marketers (retail, wholesale), aggregators, producers, asset managers and pipelines.

<u>Distributors:</u> Persons engaged in the regulated local distribution of natural gas.

- End Users: Persons that consume natural gas, including consumer advocacy organizations and retail customer groups.
- Service Providers: Persons that provide services to the participants in the retail natural gas industry, including equipment manufacturers, equipment and service vendors, software providers, energy consultants, and other companies not otherwise declared in any other Segment.

215 7.0 MEETINGS OF THE RETAIL GAS QUADRANT:

All meetings held in association with the NAESB Retail Gas Quadrant shall be open to all interested persons. Only Retail Gas Quadrant EC Members shall have the right to vote to adopt model business practices or standards affecting only the Retail Gas Quadrant. Only Retail Gas Quadrant Voting Members have the right to ratify model business practices and standards that were adopted by the Quadrant EC and that affect only the Retail Gas Quadrant.

221 8.0 JOINT MEETINGS OF THE RETAIL GAS AND OTHER QUADRANT(S):

222 At NAESB joint Subcommittee or joint Task Force meetings, the same individual may represent

Exhibit 3 – Prepared for the Retail Gas Quadrant's Consideration – Page 15

Proposed NAESB Retail Gas Quadrant Procedures - PRELIMINARY DRAFT 1-4-02

different Segments from different Quadrants, as long as that individual declares prior to the joint
meeting the Quadrant and Segment for which he or she is casting a vote. At Retail Gas Quadrant
Subcommittee or Task Force meetings, the same individual may represent different Segments, as
long as that individual declares prior to the meeting the Retail Gas Quadrant Segment for which
he or she is casting a vote. However, at the EC and Board level, an elected representative may
hold not more than one seat, representing only one Segment within one Quadrant.

229 9.0 AMENDMENTS TO RETAIL GAS QUADRANT PROCEDURES:

In order for Retail Gas Quadrant Procedures to be amended, upon petition of at least five (5) 230 Members of the Quadrant, the Vice Chair of the EC for the Quadrant shall announce a Quadrant 231 meeting. Such announcement shall provide for at least 30-days notice. In order to transact 232 business at the Quadrant meeting, there shall be a quorum consisting of at least 1/3 of the 233 Quadrant Membership. Following such meeting, the proposed resolution adopted at the meeting 234 shall be sent out for comment, and the comments shall be distributed to Quadrant members in 235 advance of a notational vote. Any Member not choosing to vote shall be considered to have 236 voted in favor of the proposed change. In order for a proposed change to take effect, it must be 237 approved by at least 2/3 of Retail Gas Quadrant Members and 40% of each Segment's 238 Membership. 239

240

241

 242

 243

 RETAIL GAS QUADRANT SUPPLIERS SEGMENT PROCEDURES

 244

 [To be provided by the Suppliers Segment]

245 **RETAIL GAS QUADRANT DISTRIBUTORS SEGMENT PROCEDURES**

246

[To be provided by the Distributors Segment]

247 **RETAIL GAS QUADRANT END USER SEGMENTS PROCEDURES**

248

[To be provided by the End Users Segment]

249	RETAIL GAS QUADRANT SERVICE PROVIDERS SEGMENT PROCEDURES
-----	--

250 [To be provided by the Service Providers Segment]

251

TAB 6

PROPOSED NAESB SEGMENT PROCEDURES - RETAIL ELECTRIC QUADRANT

2			Draft - For Review & Discussion
3			
4			
5			
6		N	ORTH AMERICAN ENERGY INDUSTRY STANDARDS BOARD
7			RETAIL ELECTRIC QUADRANT EXHIBITS
8			
9			
10			SUPPLIER SEGMENT PROCEDURES
11			
12			
13			
14	1.0	<u>MEM</u>	BERSHIP REQUIREMENTS
15		1.1	The Supplier Segment of the Retail Electric Quadrant is comprised of those
16			parties, including service providers and marketers engaged in the competitive sale
17			of electricity to end-users. Any entity which is actively engaged in the above-
18			listed activities which wants to join NAESB may join as a Supplier Segment
19			member of the Retail Electric Quadrant.
20		1.2	NAESB Supplier Segment members become members by submitting their
21			company name, contact name, qualifying business activities, and a statement of

1

intent to join to the Executive Director of NAESB . NAESB Supplier Segment
members of the Retail Electric Quadrant become voting members of the NAESB
by paying and maintaining on a current basis their NAESB membership dues,
and by executing the revocable proxy required by the NAESB incorporation
documents.

27

28 2.0 <u>PROCESS FOR ELECTING DIRECTORS AND EXECUTIVE COMMITTEE</u> 29 <u>MEMBERS</u>

30 2.1 <u>Nominations</u>

Each voting NAESB Supplier Segment organization may submit nominations to fill the positions on the Board of Directors and nominations to fill the positions on the Executive Committee Each nominee should be qualified to represent members who are participating in the activities that comprise the Supplier Segment . To qualify, a nominee should have (a) authorization from his or her organization to serve in the position and (b) decision-making authority commensurate with the position's responsibilities.

- Nominations shall be sent to the Supplier Segment's Nominating committee. The
 Nominating committee will confirm each nominee's qualifications and willingness to
 serve. The ballot will contain the names of all qualified nominees.
- If a business activity of the Supplier Segment is not represented among the nominations,
 the Nominating committee will endeavor to solicit a qualified nominee involved in that
 activity, to be included in the ballot.
- 43 2.2. <u>Voting</u>
- 44 Voting will be a one-step process. Each NAESB Supplier Segment voting member

- 45 organization will be entitled to one ballot.
- Each organization will have the same number of votes for members of the Board of Directors and for members of the Executive Committee as there are vacancies. All voting will be at-large. There will be no cumulative voting.
- The nominee(s) who receive the highest number of votes for vacancies on the Board of 49 Directors and the Executive Committee, respectively, will serve for terms as determined 50 by the NAESB Retail Electric Quadrant Board of Directors and the Executive Committee 51 members. The next three (3) nominees to the Executive Committee who receive the 52 highest number of votes, after filling the vacancies above, will serve as alternates. The 53 Board of Directors shall have no alternates. At any given time, business activities of the 54 Supplier Segment may be represented more than once on the Board of Directors or the 55 Executive committee, or both. However, no member organization may have more than 56 one (1) representative among the Retail Electric Quadrant Supplier Segment positions 57 on the Board of Directors and Executive Committee at the same time. 58
- 59

60 3.0 PROCEDURES FOR REMOVAL VACANCY, RESIGNATION OF SEGMENT

61 **DIRECTORS AND EXECUTIVE COMMITTEE MEMBERS**

- 62 Replacement of Board of Directors or Executive Committee members for partial term of 63 office due to resignation or removal of current position holder.
- 64 3.1 <u>Board of Directors</u>

If a member resigns from a position on the Board of Directors, such member's resignation shall be effective upon the later of the date the Executive Director of NAESB receives the written notice of resignation or the effective date of the resignation set forth in the written 68 notice.

If a member of the Board of Directors dies, resigns, is terminated or transfers to another
organization not included in the Supplier Segment, such member's change in status shall
be automatic resignation and effective upon the date of such change.

If an individual member of the Board of Directors transfers to another organization included in the Supplier Segment of the Retail Electric Quadrant, such individual member shall be entitled to remain a member of the Board of Directors; provided the new organization is a voting member of NAESB.

A replacement for the member resigning as set forth above shall be selected by the same 76 procedures as outlined earlier. That is, a nominating committee shall be formed, 77 nominees solicited from Supplier Segment organizations of the Retail Electric Quadrant, 78 nominees qualified by the nominating committee and ballots distributed to NAESB 79 Supplier Segment member organizations. Each member organization shall have one (1) 80 vote and the nominee receiving the most votes shall be elected to complete the then 81 current term. Such person shall be employed by an organization that is part of the 82 Supplier Segment and shall otherwise meet the qualification set forth in Section 2.A, 83 84 above. The term of such person shall be the remainder of the term of the replaced member of the Board of Directors. 85

86

3.2 <u>Executive Committee</u>

If a member resigns from a position on the Executive Committee, such member's resignation shall be effective upon the later of the date the Executive Director of NAESB receives the written notice of resignation or the effective date of the resignation set forth in the written notice. If a member of the Executive Committee dies, retires, resigns, is terminated or is transferred to another organization not included in the Supplier Segment, such member's change in status shall be an automatic resignation and effective upon the date of such change.

If an individual member of the Executive Committee transfers to another organization
 included in the Supplier Segment of the Retail Electric Quadrant, such individual
 member shall be entitled to remain a member of the Executive Committee; provided the
 new organization is a voting member of NAESB.

A replacement for the member resigning as set forth above shall be selected by the same 99 procedures as outlined earlier. That is, a nominating committee shall be formed, 100 nominees solicited from Supplier Segment organizations, nominees qualified by the 101 nominations committee and ballots distributed to NAESB Supplier Segment members. 102 Each member organization shall have one (1) vote and the nominee receiving the most 103 votes shall be elected to complete the then current term. Such person shall be employed 104 by an organization that is part of the Supplier Segment of the Retail Electric Quadrant 105 and shall have a position similar to other members of the Executive Committee. The 106 107 term of such person shall be the remainder of the term of the replaced member of the Executive Committee. 108

109

110 4.0 SELECTION OF DESIGNATED ALTERNATES AND PROCESSES FOR THEIR

111

ASSUMPTIONS OF EXECUTIVE COMMITTEE MEMBER DUTIES

4.1 Following each election within the Supplier Segment of the Retail Electric
Quadrant to fill an EC seat, the three highest vote getters not winning a seat will be

eligible to become a Supplier Segment EC Alternate. In the event there are less than 114 three runners-up to any election, the individuals eligible to be Supplier Segment EC 115 Alternates shall be selected from among the runner(s)-up from the just completed election 116 and those individuals serving as EC alternates at the time of the just completed election; 117 whereupon, the Supplier Segment EC Alternates shall be the three individuals from this 118 combined group whose vote tallies are the highest from the most recent election in which 119 they ran for an EC seat and were a runner-up. In the event of a tie for the position of 120 runner-up, the persons whose vote totals (from the most recent election in which they ran) 121 are tied shall have their names placed in a box and the Executive Director of NAESB 122 shall, in the presence of at least two Supplier Segment EC members, draw names from 123 the box until the three EC Alternate positions are filled. 124

4.2 An EC Alternate may resign at any time. The resignation of an EC Alternate shall
result in an open position until the time of the next election.

4.3 The Segment shall maintain the list of three EC Alternates and shall inform the
NAESB Office from time to time of the identities of the eligible individuals. Following
any election and at least five days tin advance of the next scheduled EC Meeting, the list
of eligible EC Alternates (including any revisions due to new elections or resignations),
shall be forwarded to the NAESB Office.

4.4 An EC Member not attending an EC Meeting (excluding by resignation) and not
submitting a notational ballot to the NAESB Office shall identify to NAESB and to their
designated EC Alternate (in writing or facsimile) the specific Alternate they designate to
sit and vote as an EC Member of the Supplier Segment at the identified EC Meeting.
Failure of an absent EC Member not submitting a notational ballot to so identify a

specific EC Alternate shall mean that those Supplier Segment EC Members attending the
subject EC Meeting may choose (an) individuals (s) from among the eligible Supplier
Segment EC Alternates in attendance to sit and vote as (an) EC Member(s) of the
Supplier Segment during the subject EC Meeting.

- 4.5 Failure of an EC Member to attend an EC Meeting by means of that EC Member's 141 resignation shall create a vacancy which may be filled during an EC Meeting by an 142 Alternate pursuant to the procedures in this paragraph 4.5 until an election has occurred. 143 A vacant position on the EC shall be filled as soon possible under the Supplier Segment 144 election procedures set forth from time to time. In the event resignation has occurred, 145 and, there has not been sufficient time to hold an election for the vacant position; and, a 146 scheduled EC Meeting is in session; then those Supplier Segment EC Members attending 147 the subject EC Meeting may choose (an) individual(s) from among the eligible Supplier 148 Segment EC Alternates in attendance to sit and vote as (an) EC Members(s) of the 149 Supplier Segment during the subject EC Meeting. 150
- 151

152 5.0 **PROCEDURES FOR AMENDING THESE SEGMENT PROCEDURES**

The Supplier Segment of the Retail Electric Quadrant has no established process for amending these segment specific procedures but contemplates that changes, if deemed necessary, would be initiated by the Supplier Segment's Board of Director members. The Suppler Segment will promptly advise NAESB of any change to these procedures.

157

158 6.0 COMMUNICATION PROCESSES FOR MEETING NOTICES

159 The Supplier Segment intends to rely upon NAESB's own communication procedures

160		for th	e distribution of meeting notices. The Supplier Segment of the Retail Electric
161		Quad	rant believes NAESB's use of an electronic bulletin board, regular newsletter,
162		other	electronic means and adherence to standard notice procedures including distribution
163		of not	tices to all interested parties will be sufficient to advise Supplier Segment member
164		organ	izations of NAESB activities.
165			
166	7.0	EXEC	CUTIVE SUMMARY
167		7.1	To belong to the Supplier Segment, and be entitled to one (1) vote in Retail
168			Electric quadrant matters, a person or entity, including service providers and
169			marketers, must be engaged in the competitive sale of electricity to end-users.
170		7.2	Board, EC, and Alternates shall be elected by the vote of the Supplier Segment
171			membership. The slate of candidates shall be determined by a Nominating
172			Committee made up of Segment members.
173		7.3	Supplier Segment may have up to three (3) Designated Alternates for the EC.
174		7.4	The Board of Directors will have no Designated Alternates.
175		7.5	If an individual member of the EC or Board of Directors transfers to another
176			organization included in the Supplier Segment, such individual member shall be
177			entitled to remain a member of the EC or Board of Directors; provided the new
178			organization is a voting member of NAESB.

RETAIL ELECTRIC QUADRANT DISTRIBUTOR SEGMENT PROCEDURES September December 2001

1.0 **DEFINITIONS**:

- 1.1 "NAESB" means the North American Energy Standards Board
- 1.2 "REQ" means the Retail Electric Quadrant of the NAESB.
- 1.3 "Segment" means one of the co-equal membership segments of REQ. As currently defined, REQ has the following four (4) segments of the retail electric market: Suppliers, Distributors, End-Users, and Service Providers.
- 1.4 "Bylaws" means the corporate bylaws of NAESB as amended from time to time.
- 1.5 "Board" means the Board of Directors of NAESB<u>as defined in the</u> NAESB bylaws.
- 1.6 "Director" means an individual serving on the Board.
- 1.7 "EC" means the Executive Committee of NAESB<u>as defined in the NAESB bylaws</u>.
- 1.8 "Segment Member" means an individual or entity (e.g., corporation, partnership) that satisfy the requirements for membership set forth in the Bylaws of REQ, and further satisfy the qualifications for Distributor Segment membership set forth in Section 3.1 of these procedures. If the Segment Member is not an individual, the term Segment Member shall include that entity's representative (employee, officer, director, or partner) who is authorized to cast that Segment member's vote.
- 1.9 "Segment Membership" means the Segment Members collectively.
- 1.10 All other capitalized terms, if not defined in this Section, shall have the same definitions as specified in the Bylaws of NAESB or in the REQ procedures.

2.0 DESCRIPTION OF DISTRIBUTOR SEGMENT:

2.1 The Distributor Segment is one of the four (4) co-equal segments comprising REQ.

- 2.2 The Distributor Segment represents the interests of individuals and entities engaged in the local distribution of electricity.
 - a) The Distributor Segment shall meet the following criteria to ensure that the broad range of distributor interests are represented:
 - 1) A minimum of ten paid members
 - 2) There are paid members representing electric distributors from at least four (4) <u>NERC Rreliability</u> regions (as defined in 2001)

3.0 MEMBERSHIP REQUIREMENTS:

3.1 Segment Members:

Any individual or entity (e.g., corporations, partnerships) distributing electricity may elect to become a member of the Distributor Segment of REQ as long as that individual or entity pays the applicable membership fee required to be paid as set forth in Section 7.0 below. A Segment Member's voting rights shall be suspended for any period during which the member is delinquent in the payment of its membership fee.

Upon the approval of then current members of the Distributor Segment and subject to the general membership requirements of NAESB and the REQ, Trade Associations that represent the interests of electricity distributors not otherwise represented, may be admitted for membership. Such "representative" members shall have the same rights and responsibilities as any Segment Member. The criteria that Segment Members should use in evaluating such a party for membership is that by admitting the "representative" member, the interests of other Distributors who can't participate directly (such as small Co-ops or municipals) will be represented. Such approval needs to be reconfirmed every two (2) years.

Members do not extend their membership to the parent company, affiliates or subsidiaries. Separate corporate entities within a corporate organization may hold multiple memberships in the Distributor Segment, but each must join individually and pay a membership fee. Members from a single corporate organization may hold only one Executive Committee and one Board of Directors seat from the Segment. In addition, a corporate organization may have only one member on the Board of Directors representing the REQ.

3.2 Meetings:

All meetings held in association with the NAESB, the REQ or the Distributor Segment of REQ are open to any interested person or entity. Only Segment Members, however, shall have a right to vote at such meetings.

3.3 Chairperson:

The Segment Members shall elect a chairperson, who shall serve a term of one year (from January 1 to December 31) and shall preside over Distributor Segment meetings.

4.0 ELECTION, RESIGNATION, AND REMOVAL OF MEMBERS OF THE BOARD OF DIRECTORS

- 4.1 Election of Directors
 - a) The Distributor Segment will be represented on the Board by four (4) persons who shall be Segment Members.
 - b) The initial Directors representing the Distributor Segment shall be elected at the first meeting of the Segment.
 - 1) The candidates receiving the greatest number of votes shall be elected, subject to a maximum of two Directors from a single NERC reliability region, as defined in 2001.
 - 2) The two (2) initial directors Directors receiving the greatest number of votes shall serve a maximum of one two years or until replaced by Directors elected in a the second regularly scheduled election as described below. The remaining two (2) initial Directors shall serve a maximum of one or until replaced by Directors elected in the first regularly scheduled election as described below.
 - c) Scheduled Elections
 - 1) In preparation for any election of Directors, other than the initial Directors, representing the Distributor Segment, a Nominating Committee, which shall be comprised of five (5) Segment Members, shall select <u>develop</u> a slate of candidates from <u>interested</u> <u>members of</u> the Segment-<u>Membership</u>. Other nominations may be made at or before the Distributor Segment meeting at which the election shall occur.
 - 2) All persons selected by the Nominating Committee, or whose nomination is made in any other manner, expressing interest in being considered for election as

<u>a Director</u> shall <u>state in writingexpress</u> their willingness to accept <u>the responsibility all</u> <u>responsibilities</u> of serving as a Director, <u>as described</u> <u>in the Bylaws</u>, prior to the submission of their names to the Segment Membership at the election.

3) Segment Members shall cast votes for candidates in a number not exceeding the number of vacant directorships.<u>and may cast only one vote per</u> <u>candidate</u>. The candidates receiving the greatest numbers of votes shall be elected, subject to a maximum of two Directors from a single NERC reliability region, as defined in 2001.

4.2 Terms of Directors

Directors shall serve two-year terms. Terms will be staggered with two Director positions up for election in one year and two the next.

(Note: How do we align this with the built-in succession from third Vice-Chair to second, to first, to Chair in the By-laws? That would require four-year terms [at least] or the ability to replace the [first, second, third] Vice Chair with their replacement Director should they not be re-elected.)

4.3 Resignation of Directors

A Director may resign his or her directorship by submitting a letter to the Secretary of NAESB with a copy to the Chairperson of the Distributor Segment stating that he or she is resigning and giving the effective date of the resignation.

4.4 Removal:

A Director may be removed upon a majority vote of the Segment Members.

4.5 Vacancies other than Expiration of Term:

A Director shall cease to be a Director upon (1) the resignation of his or her Segment Membership or lapse of Director's Segment Membership for delinquency in payment of the membership fee (if the Director is an individual Segment Member); (2) the resignation or lapse (through a delinquency in payment of the membership fee) of the Segment Membership of the entity of which the Director is a partner, director, office, or employee; or (3) the Director's resignation, removal, conviction on a felony charge, or death. A Director vacancy shall be filled for the remainder of that term in accordance with the procedures specified in Section 4.1 (c) above.

5.0 ELECTION, RESIGNATION, AND REMOVAL OF MEMBERS OF THE EXECUTIVE COMMITTEE

- 5.1 Election of Members of the Executive Committee:
 - a) The Distributor Segment will be represented on the EC by four (4) persons who shall be individual Segment Members.
 - b) The initial Directors representing the Distributor Segment shall be elected at the first meeting of the Segment.
 - 1) The candidates receiving the greatest number of votes shall be elected, subject to a maximum of two EC Members from a single NERC reliability region, as defined in 2001.
 - 2) The two (2) initial EC Members receiving the greatest number of votes shall serve a maximum of two years or until replaced by EC Members elected in the second regularly scheduled election as described below. The remaining two (2) initial EC Members shall serve a maximum of one year or until replaced by EC Members elected in a regularly scheduled election as described below.
 - c) Scheduled Elections
 - In preparation for any election of EC Members, other than the initial EC Members, representing the Distributor Segment, a Nominating Committee, which shall be comprised of five (5) Segment Members, shall <u>select_develop</u> a slate of candidates from the <u>interested members of the</u> Segment Membership. Other nominations may be made at or before the Distributor Segment meeting at which the election shall occur.
 - 2) All persons selected by the Nominating Committee, or whose nomination is made in any other mannerexpressing interest in being elected to the Executive Committee, shall state in writingexpress their willingness to accept the responsibility responsibilities of serving as a EC Member as described in the Bylaws, prior to the submission of their names to the Segment Membership at the election.
 - Segment Members shall cast votes for candidates in a number not exceeding the number of vacant EC Memberships and may cast only one vote per candidate. The candidates receiving the greatest

numbers of votes shall be elected, subject to a maximum of two EC Members from a single NERC reliability region, as defined in 2001.

d)

5.2 Terms of Executive Committee Members Executive Committee Members shall serve two-year terms. Terms will be staggered with two ECpositions up for election in one year and two the next.

5.3 Resignation of Members of the Executive Committee: A member of the EC may resign his or her position by submitting a letter to the Secretary of NAESB with a copy to the Chairperson of the Distributor Segment stating that he or she is resigning and giving the effective date of the resignation.

5.4 Removal:

A member of the EC may be removed upon a majority vote of the Segment Members.

5.5 Vacancies other than Expiration of Term:

A member of the EC shall cease to be a member of the EC upon (1) the resignation of his or her Segment Membership or lapse of his or her Segment Membership for delinquency in payment of the membership fee (if the member of the EC is an individual Segment Member); (2) the resignation or lapse (through a delinquency in payment of the membership fee) of the Segment Membership of the entity of which the member of the EC is a partner, director, officer, or employee; or (3) the member of the EC's resignation, removal, conviction on a felony charge, or death. A vacancy shall be filled for the remainder of that term in accordance with the procedures specified in Section 5.1(c) above.

6.0 DESIGNATED ALTERNATES:

- 6.1 Selection of Designated Alternates:
 - a) The Distributors Segment will select not more than ten (10) Designated Alternates to attend and vote at meetings of the Executive Committee when one or more regular members of the Executive Committee are unable to attend a meeting of the Executive Committee.
 - b) In preparation for any election of Designated Alternates for the EC, other than the initial Designated Alternates for the EC, representing the Distributor Segment, a Nominating Committee, which shall be comprised of five (5) Segment Members, shall select a slate of candidates from the Segment Membership. Other nominations may be made at or before the Distributor Segment meeting at which the

election shall occur. Segment Members shall cast votes for candidates in a number not exceeding the number of vacant positions for Designated Alternate for the EC. The candidates receiving the greatest numbers of votes shall be elected Designated Alternates.

- c) All persons selected by the Nominating Committee, or whose nomination is made in any other manner, shall state in writing their willingness to accept the responsibility of serving as a Designated Alternate prior to the submission of their names to the Segment Membership at the election.
- d) When designated by a member of the Executive Committee to attend a meeting of the EC, a Designated Alternate shall do so unless he or she is unable to attend, in which case, the Designated Alternate shall so advise the member of the EC, and the member of the EC shall select another Designated Alternate. When attending a meeting of the EC, the Designated Alternate shall have the full voting rights of the regular member of the EC, except as may be limited by that regular member's notice in writing to the Chair of the EC, the Secretary of REQ (or the person designated to serve as the secretary of the EC), and the Designated Alternate.
- e) The Chairperson of the Distributor Segment will advise REQ's Secretary in writing of the names, addresses, telephone and facsimile numbers of the Designated Alternates for the Distributor Segment.
- 6.2 Resignation of Designated Alternates:

A Designated Alternated for the Executive Committee may resign his or her position by submitting a letter to the Secretary of REQ with a copy to the Chairperson of the Distributor Segment stating that he or she is resigning and giving the effective date of the resignation.

6.3 Removal:

A Designated Alternate may be removed upon a majority vote of the Segment Members.

6.4 Vacancy:

A Designated Alternate shall cease to be a Designated Alternate upon (1) the resignation of his or her Segment Membership or lapse of his or her Segment Membership for delinquency in payment of the membership fee (if the Designated Alternate is an individual Segment Member); (2) the resignation or lapse (through a delinquency in payment of the membership fee) of the Segment Membership of the entity of which the Designated Alternate is a partner, director, officer, or employee; or (3) the Designated Alternate's resignation, removal, or

death. A vacancy shall be filled for the remainder of that term in accordance with the procedures specified in Section 6.1 above.

7.0 DISTRIBUTOR SEGMENT FUNDING

The Distributor Segment shall be funded through allocation of the Segment's NAESB dues as established by the NAESB Board of Directors

8.0 COMMUNICATIONS PROCESSES:

- 8.1 The Distributor Segment will maintain a mailing list (and e-mail list) of it's members and will directly distribute information and notices on Segment issues to Distribution Segment Members.
 - a) The Distribution Segment will maintain a mailing list (and email list) of parties who have participated in Segment proceedings during the last 12 months and will directly distribute information and notices on Segment issues these known interested parties.
 - b) The Distributor Segment will rely upon NAESB, NRECA, APPA, EEI, and other trade associations to distribute various information and notices to other interested parties.

9.0 AMENDMENT OF DISTRIBUTOR SEGMENT PROCESSES:

9.1 Procedure for Amendment:

The Directors representing the Distributor Segment on the Board shall recommend to the Segment Membership amendments to these Procedures. Each proposed amendment shall be offered as a separate amendment, except in those situations in which an amendment will affect more than one section of these Procedures. If a proposed amendment will affect more than one section, then all affected sections will be identified.

A majority vote of the Segment Membership shall be required to approve an amendment to these procedures.

9.2 Determination of Quorum:

A quorum, for purposes of these procedures only, shall be the Segment Members present at any meeting for which prior notice has been given even if the number of Segment Members present is less than a majority of the Segment Members.
10.0 EXECUTIVE SUMMARY

- 10.1 To belong to the Distributor Segment, and be entitled to one (1) vote in REQ matters, a person or entity must be engaged in the local distribution of electricity and pay the level of membership fees set out by REQ.
- 10.2 The Chairperson of the Distributors Segment shall serve a term of one (1) calendar year.
- 10.3 Board, EC, and Alternates shall be elected by the vote of the Segment membership. The slate of candidates shall be determined by a Nominating Committee made up of Segment members.
- 10.4 Distributor Segment may have up to ten (10) Designated Alternates.
- 10.5 The Segment shall be funded out of the REQ dues..
- 10.6 Majority vote of segment membership is required to amend these procedures.
- 10.7 A quorum is defined as those present at any meeting of the Distributor Segment

1	
2	DRAFT For Discussion
3	
4	Bill Boswell's comments highlighted throughout
5	Barbara Alexander Edits 12/21/01
6	
7	
8	NORTH AMERICAN ENERGY STANDARDS BOARD
9	BYLAWS ADDENDUM
10	
11	EXHIBIT 4-D
12	
13	RETAIL ELECTRIC QUADRANT END USER SEGMENT PROCEDURES

14		
15		
16	1.0	Description of the End User Segment
17		The End User Segment of the Retail Electric Quadrant of the North
18		American Standards Board ("NAESB") is one of four (4) co-equal segments
19		comprising the retail electric market. The End User Segment membership
20		is comprised of businesses, organizations, government agencies, or
21		individuals that consume electricity at facilities served by Electric
22		Distributors, and parties who represent such consumers of electricity. The
23		End User Segment shall have a minimum of five paid members at all
24		times.
25		
26	2.0	DEFINITIONS [definitions highlighted are likely to be redundant with
27		both NAESB Bylaws and Quadrant Procedures, and need not be defined
28		here again, particularly if we move the closing statement for this Section
29		to the beginning] I think the document should be a "stand alone"
30		document that makes sense to outsiders and insiders. Therefore, I would
31		not delete the definitions and procedures that make sure that the
32		document "works" even if those provisions are duplicative of the NAESB
33		<u>Bylaws or Quadrant Procedures.</u>
34		"Agent" means an individual, partnership, firm, corporation, or other entity
35		representing the interests of a member of the End User Segment, but who,
36		itself, is not necessarily a member of the Segment.

37 "Board" means the Board of Directors of NAESB as defined in the NAESB38 bylaws.

39 "Bylaws" means the corporate bylaws of NAESB as amended from time to40 time.

41 "EC" means the Executive Committee of the Retail Electric Quadrant or

42 NAESB, as appropriate to the context, as defined in the Bylaws and these
43 Retail Electric Quadrant Procedures.

44 "Electric Consumer" means an individual, partnership, corporation, or other

- 45 entity consuming electricity at one or more facilities served by an Electric
 46 Distributor.
- 47 "Electric Distributor" means an individual, partnership, corporation, or
- 48 other entity engaged in the local distribution of electricity. Most often this
- 49 will be an investor owned utility, a municipal utility, or a cooperative utility.
- 50 "Director" means an individual serving on the Board.
- ⁵¹ "NAESB" means the North American Energy Standards Board
- 52 "NAESB Office" means the office of the Secretary of NAESB
- ⁵³ "REQ" means the Retail Electric Quadrant of NAESB.

⁵⁴ "Segment" means one of the co-equal membership segments of REQ. As

- 55 currently defined, REQ has the following four (4) segments of the retail
- 56 electric market: Suppliers, Distributors, End-Users, and Services.
- ⁵⁷ "Segment Member" means an individual or entity (e.g., corporation,
- 58 government agency or partnership) that satisfy the requirements for
- 59 membership set forth in the Bylaws of REQ, and further satisfy the

60		qualifications for End User Se	gment membership set forth in Section 3 of			
61		these procedures. If the Seg	ment Member is not an individual, the term			
62		Segment Member shall includ	e that entity's representative (employee,			
63		officer, director, or partner) w	ho is authorized to cast that Segment			
64		member's vote.				
65		"Segment Membership" mear	is the Segment Members collectively.			
66						
67	All other capitalized terms, if not defined in this Section, shall have the same					
68	defin	definitions as specified in the Bylaws of NAESB or in the REQ procedures. [Good				
69	gene	al statement, maybe should b	e at the beginning of the Section.]			
70						
71	3.0	MEMBERSHIP REQUIREME	NTS			
72		(a) Membership in the End Us	ser Segment of the Retail Electric Quadrant of			
73		the NAESB shall be open to any person or legal entity (e.g. corporation,				
74		partnership) that meets the f	ollow requirements:			
75		<u>1)</u> a)—The person or entity	meets the general membership criteria set			
76		forth in the Bylaws (A	rticle 5) and the Certificate of Incorporation			
77		(Article IV Sections 1 &	2)-:			
78		<u>2)</u> b) The person or entity m	eets the membership requirements specific to			
79		the REQ of NAESB-:				
80		<u>3)</u> e) The person or entity pa	articipates in the retail electricity marketplace			
81		in one of the following	ways:			
82		(i) as a consumer	of electricity delivered by either a regulated			

- 83 investor owned or a municipal or cooperative electric
 84 distribution company;
- (ii) as a consumer of electricity delivered by an interstate
 transmission line regulated by the Federal Energy Regulatory
 Commission in the United States or the equivalent in other
 nations; or
- (iii) as a statutorily-appointed Federal, state, territorial or district
 consumer advocate or agency; and
- 91 (1) (1) 92 The person or entity has submitted an application to the Retail 92 Electricity Quadrant, declaring End User Segment affiliation and paid 93 <u>the appropriate membership dues for the REQ End User Segment of</u> 94 <u>NAESB. An applicant for membership that represents residential</u> 95 <u>consumers may apply for a reduced dues in accordance with the</u> 96 <u>provisions of the REQ Procedures -</u>
- e) The person or entity must have paid its membership dues for the
 REQ End User Segment of NAESB. A Segment Member's voting
 rights shall be suspended for any period during which the Member is
 delinquent in the payment of its membership dues. [redundant with
 Bylaws requirements; not needed here]
- 102 f) Upon the approval of current members of the End User Segment, and 103 subject to the requirements of NAESB and the REQ, Trade Associations 104 that represent the interests of electric consumers not otherwise 105 represented, may be admitted for membership. Such Trade Association

106 members shall have the same rights and responsibilities as any Segment 107 Member, except that their membership is subject to periodic reconfirmation by the other Segment members. The criteria that Segment 108 109 Members should use in evaluating such a party for membership is that by admitting such a member, the interests of other end users who can't 110 participate directly (such as individual residential customers) will be 111 represented. Approval of this Trade Association representation needs to 112 be reconfirmed every two (2) years at a minimum, if not otherwise 113 114 provided for in the REQ Procedures contained in this Exhibit 4 to the Bylaws. [Also discussed in REQ draft Procedures --- duplicative?] 115

g)c) In the event that there are not five existing members of the End User
 Segment, at least two of whom represent residential consumers,
 membership approval may be granted and reconfirmed to Trade
 Associations by the elected members of the REQ Executive Committee
 until such time as there are five non-Trade Association members, at which
 point the non-Trade Association members will assume responsibility for
 reconfirmation of Trade Association rights to continued membership.

h)d) Corporate Members do not extend their membership to their parent
 company, affiliates or subsidiaries. Separate corporate entities within a
 corporate organization may hold multiple memberships in the End User
 Segment, but each must join individually and pay a membership fee.

i)e) All representatives of any Segment Member must be duly authorized
 by the Member entity in a letter to the Segment Chair, such

authorization is to be forwarded to the NAESB Office.

130j)f)Representatives from a single corporate organization may hold no131more than one Executive Committee and one Board of Directors132seat from the Segment.

- 133 k)g) No Segment Member may have more than one representative from
 134 their organization on the Board of Directors.
- 135

136 4.0 END USER SEGMENT ORGANIZATION

137 The End User Segment Members shall elect Members of the End User138 Segment to the following administrative and governance positions:

- a) Chair and Vice-Chair
- The Chair and Vice-Chair will preside over all meetings of the 140 141 general Segment membership separate from meetings of the REQ. 142 The Chair will appoint a Secretary of the meeting(s), which may be a standing position, to take meeting minutes and record votes. 143 144 Minutes of all meetings, including voting results, are to be reported 145 by the Chair, the Vice-Chair, or the Secretary at the Chair's discretion, to the Members and transmitted to the NAESB Office for 146 publication within 15 days of the date of the meeting. 147
- The Vice-Chair will organize and conduct all Segment elections once the Segment has commenced operation. The method of initial election of all officers is addressed in Section 6.8.
- 151 The term of office of the Chair is one year concurrent with the

152operating year of NAESB.The Chair will be succeeded in the153following year by the Vice-Chair, with a new Vice-Chair elected to fill154the vacant Vice-Chair position.There is no limit on the number of155terms any End User Segment Member may serve either as Chair or156Vice-Chair.

b) Executive Committee members

158 The Segment Membership will elect two Group A or two Group B 159 Executive Committee members in alternating years as provided for 160 in the REQ Procedures.

161 c) Directors

162The Segment Membership will elect two Group A or two Group B163Directors in alternating years as provided for in the REQ Procedures.

164

165 **5.0 MEETINGS**

All meetings held in association with NAESB, the REQ, or within the End User Segment of REQ are open to any interested person or entity. Only Segment Members, however, shall have a right to vote at such meetings on behalf of *["having this phrase here is critical to this statement being workable"]* the End User Segment.

171

5.1 Notices of meetings [drop; subsections 5.1 and 5.2 are both NAESBops *issues*]

174 Notices of meetings will be posted to the End User Segment membership

- no less than seven (7) days prior to the date of the meeting, said notices
 providing the location, date, times and agenda topics of the meeting.
- 177

178 5.2 Notification Method

Notification will be through the use of an electronic bulletin board, regular
newsletter, and adherence to standard notice a) The NAESB Office shall
advise REQ End User Segment Member organizations of NAESB activities.
As such the End User Segment intends to rely upon the NAESB Office to
distribute meeting notices, as required under Article 11 of the NAESB
Bylaws, to Segment Members and other interested parties as deemed
appropriate.

- b) If the NAESB Office cannot perform this function, then the Segment Membership will determine an agreed method of notification or reconfirm the agreement at each first meeting of the operating year of the Segment Membership.
- 190

191 5.3 Notices of Elections ["OK"{CLD -- but probably now should be dropped
192 as it is covered in Section 6}]

193 Notices of elections shall be posted as provided for in Section 6 of these194 Procedures.

195

1966.0ELECTION, RESIGNATION AND REMOVAL OF OFFICERS OF THE197END USER SEGMENT [Section 6 is an "awfully busy section", needs to

198 *be simplified*]

199 6.1 <u>Election of Directors</u>:

- a) The End User Segment of the REQ shall be represented on the
 NAESB Board of Directors by four individuals who shall be elected,
 from time to time, as required by Article 7 of the NAESB Bylaws and
 in this Exhibit 4 to those Bylaws. <u>At least one of these individuals</u>
 shall represent residential consumers.
- b) Each Director from the End User Segment of the REQ shall serve a term of two years on the NAESB Board, unless provided for otherwise in the REQ Procedures.
- 208 (i) Directors elected in a year ending in an odd digit (i.e., 2003),
 209 whose term would expire in a year ending in an odd digit
 210 (i.e., 2005), will serve as Group A Directors; there will be two
 211 Group A Director seats at any given time, subject to the
 212 conditions specified in Section 6.8.
- 213 (ii) Directors elected in a year ending in an even digit, whose
 214 term would expire in a year ending in an even digit, will serve
 215 as Group B Directors; there will be two Group B Director seats
 216 at any given time, subject to the conditions specified in
 217 Section 6.8.
- c) To be eligible to serve as an End User Segment representative on
 the NAESB Board of Directors, the individual must
- 220 (i) be a Segment Member

(ii) have a working knowledge of the NAESB process;

- (iii) be willing to commit the time and resources necessary to
 fulfill their obligations as a NAESB Board member and to meet
 the minimum threshold of participation and attendance
 established in the NAESB Bylaws
- (iv) disclose their interest, or their employer's interest, in the
 electric industry and the relationship with other entities with
 which the employer may be affiliated;
- d) No individual representing any single Segment Member shall serve
 on the Board of Directors and on the REQ Executive Committee
 concurrently.
- e) Interested individuals duly authorized to represent a Segment Member and eligible to serve on the Board of Directors shall submit their names to the NAESB Office in a form as specified in NAESB Operating Procedures or,
- (i) if such requirements are not provided for in the NAESB
 Operating Procedures, interested individuals shall submit a
 written statement to the NAESB Office that provides their
 name, title and affiliation (if any), business mail and email
 addresses, business telephone, a statement that they accept
 the conditions referred to in subsection (c)-(iii) above, and
 the statement required in subsection (c)-(iv) above.
- *f)* In preparation for any election of NAESB Directors (other than initial

244 Directors, as provided for in Section 6.8) representing the End User Segment of the REQ, a Nominating Committee consisting of five (5) 245 existing Board and EC members of the REQ End User Segment, shall 246 247 develop a slate of candidates from interested members of the Segment Membership. Other nominations may be made at or prior 248 to the close of the REQ End User Segment nomination period by any 249 Segment Member in accordance with the requirements in subsection 250 251 (e) above. All nominations must be made and conveyed in writing 252 to the NAESB Secretary no less than 31 days prior to the election date. [this really belongs in the Quadrant Procedures to keep all 253 Segments working the same way as much as possible] 254

- 255 g) The election process for REQ End User Segment representatives to256 the NAESB Board shall be
- 257 (i) coordinated by the Vice-Chair of the Segment with
 258 appropriate coordination with the Quadrant Executive
 259 Committee and the NAESB Office such that elections occur
 260 during the second week of November of each year
- 261 (ii) announced by formal notice to the Segment Membership no
 262 less than 30 days prior to the date of the election, with such
 263 notice indicating the name and affiliation of all candidates.
- h) Elections for REQ End User Segment Directors shall be subject to
 the following provisions:

266

(i)

Any End User Segment Member of the REQ with dues current

- 267 is eligible to have one duly authorized representative cast the268 Member's vote.
- 269 (ii) End User Segment Members shall cast votes for candidates in
 270 a number not exceeding the number of Director seats open
 271 for election from the End User Segment.
- 272 (iii) The candidates receiving the greatest numbers of votes shall273 be elected.
- (iv) Results of the REQ elections for the NAESB Board, including
 the returned ballots, shall be forwarded to the Secretary of
 NAESB, for the purpose of public record and validation of
 voting results, upon conclusion of any election.
- i) Any Director seats which become vacant shall be filled in accordance
 with REQ Procedures as specified in this Exhibit 4 to the Bylaws, or if
 such provisions are lacking, by special election, subject to the
 following provisions
- (i) If the vacancy occurs with less than 125 days remaining in
 the Director's term of office, the seat will remain vacant until
 filled by election of a new Director to a term commencing in
 the next operating year.
- 286 (ii) All other vacancies will be filled by special election; such
 287 special elections are to be formally noticed to the Segment
 288 Membership by the Segment Vice-Chair or Chair within 15
 289 days of when the creation of the vacancy becomes known and

held within 60 days of the creation of the vacancy.
Candidates will be designated by the normal nomination
process or selected by the remaining Board members of the
Segment from Segment Membership. Candidates elected to
fill the vacancy will serve the remainder of the term, and may
stand for re-election.

296

297

6.2 Resignation of Directors

A Director may resign his or her directorship by submitting a letter to the Secretary of NAESB with a copy to the Chair of the End User Segment of the REQ stating that he or she is resigning and giving the effective date of the resignation.

302

303 6.3 Removal of Directors

a) A NAESB Director shall cease to be a Director upon

305 (i) the resignation of REQ End User Segment Membership by the
 306 entity of which the Director serves as a representative

307 (ii) the lapse of the Segment Membership dues of the entity
308 which the Director serves as a representative, or

309 (iii) the Director's resignation, removal, or death

310 (iv) the first day of the NAESB operating year following an
311 election for the seat held by the Director, wherein the
312 Director was not re-elected by the Segment Membership as

313			declared by the NAESB Secretary.
314	b)	Rece	ipt by the NAESB Secretary of a rescission letter indicating that
315		the i	ndividual no longer represents the Member does not constitute
316		grou	nds for removal of the Director if all three of the following
317		cond	itions are satisfied:
318		(i)	the individual continues activity or employment in the End
319			User Segment of the Electric industry with another End User
320			Segment Member,
321		(ii)	the individual's new employer is a Segment Member with no
322			representatives holding seats on the Board, and
323		(iii)	the Board determines that completion of the Director's term
324			of office is in the best interest of NAESB
325	c)	A va	cancy resulting from removal of a Director shall be filled for the
326		rema	ainder of that term in accordance with Section 6.1(i) of these
327		Proce	edures.
328	[the contents of 6.3 should be probably be covered within the Quadrant		
329	Procedures document, the Bylaws or NAESBops; good to lay it out, but not a		
330	good thing to have individual Segments following significantly different		
331	approaches on this]		
332			
333	6.4 Elec	tion o	f Executive Committee Members
334	The	End Us	ser Segment of the REQ shall be represented on the NAESB EC

by four individuals who shall be elected, from time to time, as required by

- Article 10 of the NAESB Bylaws and in this Exhibit 4 to those Bylaws. <u>At</u>
 least one of these individuals shall represent residential consumers.
- a) Each EC member from the End User Segment of the REQ shall serve
 a term of two years on the REQ EC, unless provided for otherwise in
 the REQ Procedures.
- 341 (iii) Members elected in a year ending in an odd digit (i.e., 2003),
 342 whose term would expire in a year ending in an odd digit
 343 (i.e., 2005), will serve as Group A representatives; there will
 344 be two Group A representative seats at any given time,
 345 subject to the conditions specified in Section 6.8.
- (iv) Members elected in a year ending in an even digit, whose
 term would expire in a year ending in an even digit, will serve
 as Group B representatives; there will be two Group B
 representative seats at any given time, subject to the
 conditions specified in Section 6.8.
- b) To be eligible to serve as an End User Segment representative on
 the EC, the individual must
- 353 (i) be a Segment Member
- 354 (ii) have a working knowledge of the NAESB process;
- 355 (iii) be willing to commit the time and resources necessary to
 356 fulfill their obligations as a NAESB EC member and to meet
 357 the minimum threshold of participation and attendance
 358 established in the NAESB Bylaws

- 359 (iv) disclose their interest, or their employer's interest, in the
 360 electric industry and the relationship with other entities with
 361 which the employer may be affiliated;
- 362 c) No individual representing any single Segment Member shall serve
 363 on the Board of Directors and on the REQ Executive Committee
 364 concurrently.
- d) Interested individuals duly authorized to represent a Segment
 Member and eligible to serve on the EC shall submit their names to
 the NAESB Office in a form as specified in NAESB Operating
 Procedures or,
- (i) if such requirements are not provided for in the NAESB
 Operating Procedures, interested individuals shall submit a
 written statement to the NAESB Office that provides their
 name, title and affiliation (if any), business mail and email
 addresses, business telephone, a statement that they accept
 the conditions referred to in subsection (c)-(iii) above, and
 the statement required in subsection (c)-(iv) above.
- e) In preparation for any election of EC members (other than initial EC
 members, as provided for in Section 6.8) representing the End User
 Segment of the REQ, a Nominating Committee consisting of five (5)
 Board or EC members of the REQ End User Segment, shall develop
 a slate of candidates from interested members of the Segment
 Membership. Other nominations may be made at or prior to the

close of the REQ End User Segment nominating period by any
Segment Member in accordance with the requirements in subsection
(e) above. All nominations must be made and conveyed in writing
to the NAESB Secretary no less than 31 days prior to the election
date.

- f) The election process for REQ End User Segment representatives to
 the REQ EC shall be
- (iii) coordinated by the Vice-Chair of the Segment with
 appropriate coordination with the REQ EC and the NAESB
 Office such that elections occur during the second week of
 November of each year
- (iv) announced by formal notice to the Segment Membership no
 less than 30 days prior to the date of the election, with such
 notice indicating the name and affiliation of all candidates.
- 396 g) Elections for REQ End User Segment EC members shall be subject to397 the following provisions:
- 398 (i) Any End User Segment Member of the REQ with dues current
 399 is eligible to have one duly authorized representative cast the
 400 Member's vote.
- 401 (ii) End User Segment Members shall cast votes for candidates in
 402 a number not exceeding the number of EC seats open for
 403 election by the End User Segment.
- 404 (iii) The candidates receiving the greatest numbers of votes shall

- 405 be elected.
- 406 (iv) Results of the elections for the REQ EC, including the returned
 407 ballots, shall be forwarded to the Secretary of NAESB, for the
 408 purpose of public record and validation of voting results, upon
 409 conclusion of any election.
- h) Any EC seats becoming vacant shall be filled in accordance with REQ
 Procedures as specified in this Exhibit 4 to the Bylaws, or if such
 provisions are lacking, by a special election subject to the following
 provisions
- 414 (i) If the vacancy occurs with less than 125 days remaining in
 415 the EC member's term of office, the seat will remain vacant
 416 until filled by election of a new EC member to a term
 417 commencing in the next operating year, and a Designated
 418 Alternate will serve until expiration of the term.
- (ii) All other vacancies will be filled by special election; such 419 special elections are to be formally noticed to the Segment 420 421 Membership by the Segment Vice-Chair or Chair within 15 422 days of when the creation of the vacancy becomes known and held within 60 days of the creation of the vacancy. 423 Candidates will be designated by the normal nomination 424 process or selected by the remaining EC members of the 425 Segment from Segment Membership. Candidates elected to 426 fill the vacancy will serve the remainder of the term, and may 427

428				stand for re-election.
429			(iii)	Until an election is held and a new EC member is elected, a
430				Designated Alternate will serve in the open EC seat.
431	[the	highlig	phted p	parts of this section are redundant with what should be in the
432	Quad	rant Pr	ocedur	res; unnecessary]
433				
434	6.5	Resig	Inatio	n of EC Representatives
435		An EC	c mem	ber may resign his or her seat by submitting a letter to the
436		Secre	tary of	NAESB with a copy to the Chair of the End User Segment of
437		the RI	EQ stat	ting that he or she is resigning and giving the effective date of
438		the re	signati	ion.
439	[all of the remaining portions of Section 6 belong in the Quadrant Procedures so			
440	that all Segments operate the same in regards to these issues]			
441				
442	6.6	Remo	oval of	EC Representatives
443		a)	An EC	member shall cease to be a member of the EC upon
444			(i)	the resignation of REQ End User Segment Membership by the
445				entity of which the EC member serves as a representative
446			(ii)	the lapse of the Segment Membership dues of the entity
447				which the EC member serves as a representative,
448			(iii)	the EC member's resignation, removal, or death
449			(iv)	the first day of the NAESB operating year following an
450				election for the seat held by the EC member, wherein the

451			member was not re-elected by the Segment Membership as
452			declared by the NAESB Secretary
453			(iv) receipt by the NAESB Secretary of a rescission letter
454			indicating that the individual no longer represents the
455			Segment Member
456		b)	A vacancy resulting from removal of an EC member shall be filled for
457			the remainder of that term in accordance with Section 6.4(h) of
458			these Procedures.
459			
460	6.7	Desi	gnation of Alternates for Executive Committee Members
461		<u>a)</u>	Selection of Designated Alternates:
462			The End User Segment will select no more than three (3), and no
463			les than one Designated Alternate EC representative in each election
464			year, allowing for up to three (3) Designated Alternates overall.
465		b)	Designated Alternates will be selected by the same election that
466			determines EC members, with those unsuccessful EC member
467			candidate(s) receiving the greatest number of votes becoming a
468			Designated Alternate in the event that there already are or are
469			expected to be less than three Designated Alternates from the End
470			User Segment in the coming year.
471		c)	The first, second and third Designated Alternate will be determined
472			by
473			(i) the order in which they become named as a Designated

- 474 Alternate, as indicated by the first year of their designation 475 and
- 476 (ii) the order of voting results.
- 477 d) A Designated Alternate will attend and vote at meetings of the EC of the REQ or of the larger EC of NAESB, when one or more regular EC 478 Members from the End User Segment are unable to attend a 479 meeting of the EC and when the absent member or the Vice-Chair of 480 481 the REQ EC requests their attendance as a Designated Alternate. 482 Selection of which Alternate will serve will proceed through the order of first, second or third Designated Alternate, with each indicating 483 484 whether they can or cannot fulfill the requirement to serve at the 485 subject meeting.

486

487 6.8 Transitional Election of Segment Officers

The requirements of the NAESB Bylaws for a minimum of 5 Members per 488 489 Segment and 40 Members per Quadrant do not provide for enough 490 members to satisfy the constraints on election of Directors and EC members as stated in earlier subsections of this Section 6. Consequently, 491 492 until the End-User Segment has at least eight (8) Members, the number of Directors and EC Members elected by the Segment Membership within the 493 Quadrant will be halved, to 2 Directors and 2 EC members with no dilution 494 of the Segment's balanced voting within the Quadrant. 495

496

- The following provisions will apply and supercede the requirements of
 Sections 6.1 through 6.6 until the End-User Segment has enrolled a
 minimum of eight (8) members.
- 500

501 If not otherwise provided for in the REQ Procedures, the following 502 provisions will apply during the initial operating period of the End User 503 Segment until the Segment has at least eight (8) paid Members.

- a) Term of Office of Initial Directors
- 505 The initial group of two (2) Directors will both serve a 2 year term of 506 office as either Group A or Group B Directors, as determined by the 507 expiration year of their original term of office.
- 508 b) Interim Elections of Directors
- 509 Within 60 days of receipt of notification from the NAESB Office that 510 the End-User Segment has achieved eight (8) members, an election 511 shall be held in order to select of two additional Directors to serve 512 alongside the initially elected Directors
- 513(i)the term of office of the second group elected at this time will514expire one year later than the term of the initially elected515Directors, thereby allowing for two Group A and two Group B
- 516 Directors from the Segment.
- 517 c) Term of Office of Initial Executive Committee Members

- 518 The initial group of EC members will serve a 2 year term of office as
- 519 either Group A or Group B members, as determined by the

520 expiration year of their original term of office.

- 521 d) Interim Elections of EC Members
- 522 Within 60 days of receipt of notification from the NAESB Office that 523 the End-User Segment has achieved eight (8) members an election 524 shall be held in order to select of two additional EC Members to 525 serve alongside the initially elected Directors
- 526 (i) The term of office of the second group elected at this time will
 527 expire one year later than the term of the initially elected EC
 528 members, thereby allowing for two Group A and two Group B
 529 EC members from each Segment.
- (ii) The previously elected Chair and Vice-Chair of the Quadrant
 EC will continue in their official role until either their reelection at the end of their term or removal from office in
 accordance with the NAESB Bylaws, Quadrant Procedures or
 these Procedures.
- 535

536 **7.0 AMENDMENT OF SEGMENT PROCEDURES**

Amendment of these REQ End User Segment Procedures shall be subject to approval by a 90% majority of the Segment Members. *[at first blush, this seemed awfully high to Bill; as we discussed the undesirability of having Segments change their procedures frequently, he came around to* 541 maybe 80-90% is an appropriate number; up to the Segment].



RETAIL ELECTRIC QUADRANT SERVICES SEGMENT PROCEDURES

1.0 Table of Contents

1.0 Table of Contents	. 1
2.0 Definitions	
3.0 Description of the Services Segment	.2
4.0 Services Segment Membership Requirements	
5.0 Services Segment Funding	
6.0 Services Segment Organization	3
6.1 Chair and Vice-Chair	3
6.2 Executive Committee (EC) Members	4
6.3 Directors	4
7.0 Determination of Quorum	
8.0 Meetings	
9.0 Services Segment Notifications	.4
9.1 Notices of Meetings	4
9.2 Notification Method	4
9.3 Notices of Elections	5
10.0 Election, Resignation, and Removal of Services Segment Officers	
10.1 Election of Directors	5
10.2 Election of EC Members	8
	11
	11
	2
	13
11.0 Amendment of Services Segment Procedures	3

2.0 Definitions

Board - means the Board of Directors of NAESB as defined in the NAESB bylaws.

Bylaws - means the corporate bylaws of NAESB as amended from time to time.

<u>EC</u> - means the Executive Committee of the Retail Electric Quadrant or NAESB, as appropriate to the context, as defined in the Bylaws and these Retail Electric Quadrant Procedures.

Director - means an individual serving on the Board.

NAESB - means the North American Energy Standards Board

NAESB Office - means the office of the Secretary of NAESB

REQ - means the Retail Electric Quadrant of NAESB.

<u>Segment</u> - means one of the co-equal membership segments of REQ. As currently defined, REQ has the following four (4) segments of the retail electric market: Suppliers, Distributors, End-Users, and Services.

<u>Segment Member</u> - means an individual or entity (e.g., corporation, government agency or partnership) that satisfies

NAESB Bylaws Addendum Exhibit 4-B Retail Electric Quadrant Services Segment Procedures Draft Last Revised: 01/14/02

the requirements for membership set forth in the Bylaws of REQ, and further satisfies the qualifications for Services Segment membership set forth in Section 3 of these procedures. If the Segment Member is not an individual, the term Segment Member shall include that entity's representative (employee, officer, director, or partner) who is authorized to cast that Segment member's vote.

<u>Segment Membership</u> - means the Segment Members collectively.

All other capitalized terms, if not defined in this Section, shall have the same definitions as specified in the Bylaws of NAESB or in the REQ procedures.

3.0 Description of the Services Segment

The Services Segment of the Retail Electric Quadrant of the North American Energy Standards Board ("NAESB") is one of four (4) co-equal segments comprising the retail electric market. The Services Segment membership is comprised of businesses that provide services to suppliers and distributors of electricity or an entity not otherwise declared in another segment. The Services Segment shall have a minimum of five paid members at all times.

4.0 Services Segment Membership Requirements

- Membership in the Services Segment of the Retail Electric Quadrant of the NAESB shall be open to any entity (e.g., corporation, partnership or proprietorship) that meets the following requirements:
 - a) The entity meets the general membership criteria set forth in the Bylaws (Article 5) and the Certificate of Incorporation (Article IV Sections 1 & 2).
 - b) The entity meets the membership requirements specific to the REQ of NAESB.
 - c) The entity provides services to participants in the retail electric industry.
 - d) The entity is not otherwise declared in another segment.
 - e) The entity has submitted an application to the REQ, declaring Services Segment affiliation.
 - f) Corporate Members do not extend their membership to their parent company, affiliates or subsidiaries. Separate

corporate entities within a corporate organization may hold multiple memberships in the Services Segment, but each must join individually and pay a membership fee.

- g) All representatives of any Services Segment Member must be duly authorized by the Member entity in a letter to the Segment Chair, such authorization is to be forwarded to the NAESB Office.
- *h*) Representatives from a single corporate organization may hold no more than one Executive Committee and one Board of Directors seat from the Segment.
- i) No Segment Member may have more than one representative from their organization on the Board of Directors.
- j) The entity satisfies the requirements for Membership set forth in Article 5 of the Bylaws of NAESB, and Section 3 of the Retail Electric Quadrant Procedures.

5.0 Services Segment Funding

The Services Segment shall be funded through allocation of the Segment's NAESB dues as established by the NAESB Board of Directors

6.0 Services Segment Organization

The Services Segment Members shall elect Members of the Services Segment to the following administrative and governance positions:

6.1 Chair and Vice-Chair

- The Chair and Vice-Chair will preside over all meetings of the general Services Segment membership separate from meetings of the REQ. The Chair will appoint a Secretary of the meeting(s), which may be a standing position, to take meeting minutes and record votes. Minutes of all meetings, including voting results, are to be reported by the Chair, the Vice-Chair, or the Secretary at the Chair's discretion, to the Members and transmitted to the NAESB Office for publication within 15 days of the date of the meeting.
- The Vice-Chair will organize and conduct all Segment elections once the Segment has commenced operation. The method of initial election of all officers is addressed in Section 8.6

Transitional Election of Officers of this document.

The term of office of the Chair is one year concurrent with the operating year of NAESB. The Chair will be succeeded in the following year by the Vice-Chair, with a new Vice-Chair elected to fill the vacant Vice-Chair position. There is no limit on the number of terms any Services Segment Member may serve either as Chair or Vice-Chair.

6.2 Executive Committee (EC) Members

Four (4) persons who are individual Segment Members will represent the Services Segment on the REQ EC. The Segment Membership will elect Executive Committee members as provided for in the REQ Procedures and Section 10.2 Election of EC Members of this document.

6.3 Directors

Four (4) persons who are individual Segment Members will represent the Services Segment on the REQ Board of Directors. The Segment Membership will elect Directors as provided for in the REQ Procedures and Section 10.1 Election of Board of Directors of this document.

7.0 Determination of Quorum

A quorum for purposes of amending these procedures, electing officers, or voting on any other Services Segment issues shall be at least 51% of the Segment Members.

8.0 Meetings

All meetings held in association with the Services Segment of REQ are open to any interested company or organization. Only Segment Members, however, shall have a right to vote at such meetings.

9.0 Services Segment Notifications

9.1 Notices of Meetings

Notices of meetings will be posted to the Services Segment membership no less than seven (7) days prior to the date of the meeting, said notices providing the location, date, times and agenda topics of the meeting.

9.2 Notification Method

The Services Segment Chair will appoint a Segment Member to maintain a mailing list (and e-mail list) of Segment Members and will directly distribute information and notices on Segment meetings and issues to Services Segment Members.

Notification may additionally be provided by any of the means referenced in Article 11 of the NAESB Bylaws.

9.3 Notices of Elections

Notices of elections shall be posted as provided for in Sections 10.1 Election of Directors and 10.2 Election of EC Members of these Procedures.

10.0 Election, Resignation, and Removal of Services Segment Officers

10.1 Election of Directors

- a) The Services Segment of the REQ shall be represented on the NAESB Board of Directors by four individuals who shall be elected, from time to time, as required by Article 7 of the NAESB Bylaws and in this Exhibit 4 to those Bylaws.
- b) Each Director from the Services Segment of the REQ shall serve a term of two years on the NAESB Board, except as provided for in Section 10.6 Transitional Election of Officers of this procedure or unless provided for otherwise in the REQ Procedures.
 - Directors elected in a year ending in an odd digit (i.e., 2003), whose term would expire in a year ending in an odd digit (i.e., 2005), will serve as Group A Directors; there will be two Group A Director seats at any given time.
 - (ii) Directors elected in a year ending in an even digit, whose term would expire in a year ending in an even digit, will serve as Group B Directors; there will be two Group B Director seats at any given time.
- c) To be eligible to serve as a Services Segment representative on the NAESB Board of Directors, the individual must
 - (i) be a Services Segment Member

- (ii) have a working knowledge of the NAESB process;
- (iii) be willing to commit the time and resources necessary to fulfill their obligations as a NAESB Board member and to meet the minimum threshold of participation and attendance established in the NAESB Bylaws
- (iv) disclose the legal name for their employer and if applicable, its ownership, sister companies also under the umbrella of its owner, and a list of other entities for which the employer is engaged in a legal partnership or joint venture.
- (v) provide a copy of their employer's mission statement or a summary of their business objectives along with a brief statement regarding which segment these objectives best serve;
- d) No individual representing any single Segment Member shall serve on the Board of Directors and on the REQ Executive Committee concurrently.
- e) Interested individuals duly authorized to represent a Segment Member and eligible to serve on the Board of Directors shall submit their names to the NAESB Office in a form as specified in NAESB Operating Procedures or,
 - (i) if such requirements are not provided for in the NAESB Operating Procedures, interested individuals shall submit a written statement to the NAESB Office that provides their name, title and affiliation (if any), business mail and email addresses, business telephone, a statement that they accept the conditions referred to in subsection (c)-(iii) above, and the statement required in subsection (c)-(iv) above.
- f) In preparation for any election of NAESB Directors (other than initial Directors, as provided for in Section 10.6) representing the Services Segment of the REQ, a Nominating Committee consisting of five (5) existing Board and EC members of the REQ Services Segment, shall be appointed by the Services Segment Chair. This committee shall develop a slate of candidates from interested members of the Segment Membership. Other nominations may be made at or prior to the

h)

close of the REQ Services Segment nomination period by any Segment Member in accordance with the requirements in subsection (e) above. All nominations must be made and conveyed in writing to the NAESB Secretary no less than 31 days prior to the election date.

- g) The election process for REQ Services Segment representatives to the NAESB Board shall be
 - coordinated by the Vice-Chair of the Segment with appropriate coordination with the Quadrant Executive Committee and the NAESB Office such that elections occur during the second week of November of each year
 - (ii) announced by formal notice to the Segment Membership no less than 30 days prior to the date of the election, with such notice indicating the name and affiliation of all candidates.

Elections for REQ Services Segment Directors shall be subject to the following provisions:

- (i) Any Services Segment Member of the REQ with dues current is eligible to cast a vote.
- (*ii*) Services Segment Members shall cast votes for candidates in a number not exceeding the number of Director seats open for election from the Services Segment.
- (iii) The candidates receiving the greatest numbers of votes shall be elected.
- (iv) Results of the REQ elections for the NAESB Board, including the returned ballots, shall be forwarded to the Secretary of NAESB, for the purpose of public record and validation of voting results, upon conclusion of any election.
- Any Director seats which become vacant shall be filled in accordance with REQ Procedures as specified in this Exhibit 4 to the Bylaws, or if such provisions are lacking, by special election, subject to the following provisions
 - (i) If the vacancy occurs with less than 125 days remaining in the Director's term of office, the seat will remain vacant until filled by election of a new

Director to a term commencing in the next operating year.

(ii) All other vacancies will be filled by special election; such special elections are to be formally noticed to the Segment Membership by the Segment Vice-Chair or Chair within 15 days of when the creation of the vacancy becomes known and held within 60 days of the creation of the vacancy. Candidates will be designated by the normal nomination process or selected by the remaining Board members of the Segment from Segment Membership. Candidates elected to fill the vacancy will serve the remainder of the term, and may stand for re-election.

10.2 Election of EC Members

- a) The Services Segment of the REQ shall be represented on the NAESB EC by four individuals who shall be elected, from time to time, as required by Article 10 of the NAESB Bylaws and in this Exhibit 4 to those Bylaws.
- b) Each EC member from the Services Segment of the REQ shall serve a term of two years on the REQ EC, unless provided for otherwise in the REQ Procedures.
 - (i) Members elected in a year ending in an odd digit (i.e., 2003), whose term would expire in a year ending in an odd digit (i.e., 2005), will serve as Group A representatives; there will be two Group A representative seats at any given time.
 - (ii) Members elected in a year ending in an even digit, whose term would expire in a year ending in an even digit, will serve as Group B representatives; there will be two Group B representative seats at any given time.
 - c) To be eligible to serve as a Services Segment representative on the EC, the individual must
 - (i) be a Segment Member
 - (ii) have a working knowledge of the NAESB process;
 - (iii) be willing to commit the time and resources necessary to fulfill their obligations as a NAESB EC

member and to meet the minimum threshold of participation and attendance established in the NAESB Bylaws

- (iv) disclose the legal name for their employer and if applicable, its ownership, sister companies also under the umbrella of its owner, and a list of other entities for which the employer is engaged in a legal partnership or joint venture
- (v) provide a copy of their employer's mission statement or a summary of their business objectives along with a brief statement regarding which segment these objectives best serve;
- d) No individual representing any single Segment Member shall serve on the Board of Directors and on the REQ Executive Committee concurrently.
- e) Interested individuals duly authorized to represent a Segment Member and eligible to serve on the EC shall submit their names to the NAESB Office in a form as specified in NAESB Operating Procedures or,
 - (i) if such requirements are not provided for in the NAESB Operating Procedures, interested individuals shall submit a written statement to the NAESB Office that provides their name, title and affiliation (if any), business mail and email addresses, business telephone, a statement that they accept the conditions referred to in subsection (c)-(iii) above, and the statement required in subsection (c)-(iv) above.
- f) In preparation for any election of EC members (other than initial EC members, as provided for in Section 10.6) representing the Services Segment of the REQ, a Nominating Committee consisting of five (5) Board or EC members of the REQ Services Segment, shall be appointed by the Services Segment Chair. This committee will develop a slate of candidates from interested members of the Segment Membership. Other nominations may be made at or prior to the close of the REQ Services Segment nominating period by any Segment Member in accordance with the requirements in subsection (e) above. All nominations must be made and conveyed in writing to the NAESB
Secretary no less than 31 days prior to the election date.

- g) The election process for REQ Services Segment representatives to the REQ EC shall be
 - (i) coordinated by the Vice-Chair of the Segment with appropriate coordination with the REQ EC and the NAESB Office such that elections occur during the second week of November of each year
 - (ii) announced by formal notice to the Segment Membership no less than 30 days prior to the date of the election, with such notice indicating the name and affiliation of all candidates.
- h) Elections for REQ Services Segment EC members shall be subject to the following provisions:
 - (i) Any Services Segment Member of the REQ with dues current is eligible to have one duly authorized representative cast the Member's vote.
 - (ii) Services Segment Members shall cast votes for candidates in a number not exceeding the number of EC seats open for election by the Services Segment.
 - (iii) The candidates receiving the greatest numbers of votes shall be elected.
 - (iv) Results of the elections for the REQ EC, including the returned ballots, shall be forwarded to the Secretary of NAESB, for the purpose of public record and validation of voting results, upon conclusion of any election.
- Any EC seats becoming vacant shall be filled in accordance with REQ Procedures as specified in this Exhibit 4 to the Bylaws, or if such provisions are lacking, by a special election subject to the following provisions
 - (i) If the vacancy occurs with less than 125 days remaining in the EC member's term of office, the seat will remain vacant until filled by election of a new EC member to a term commencing in the next operating year, and a Designated Alternate will serve until expiration of the term.

NAESB Bylaws Addendum Exhibit 4-B Retail Electric Quadrant Services Segment Procedures Draft Last Revised: 01/14/02

- (ii) All other vacancies will be filled by special election; such special elections are to be formally noticed to the Segment Membership by the Segment Vice-Chair or Chair within 15 days of when the creation of the vacancy becomes known and held within 60 days of the creation of the vacancy. Candidates will be designated by the normal nomination process or selected by the remaining EC members of the Segment from Segment Membership. Candidates elected to fill the vacancy will serve the remainder of the term, and may stand for reelection.
- (iii) Until an election is held and a new EC member is elected, a Designated Alternate will serve in the open EC seat.

10.3 Resignation of Directors or EC Members

A Director or EC Member may resign his or her position by submitting a letter to the Secretary of NAESB with a copy to the Chair of the Services Segment of the REQ stating that he or she is resigning and giving the effective date of the resignation.

10.4 Removal of Directors or EC Members

- a) A NAESB Director or EC Member shall cease to hold that position upon
 - (i) the resignation of REQ Services Segment Membership by the company of which the Director or EC Member serves as a representative
 - (ii) the lapse of the Segment Membership dues of the entity which the Director or EC Member serves as a representative, or
 - (iii) the Director or EC Member's resignation, removal, or death
 - (iv) the first day of the NAESB operating year following an election for the seat held by the Director or EC Member, wherein the Director or EC Member was not re-elected by the Segment Membership as declared by the NAESB Secretary.
- b) Receipt by the NAESB Secretary of a rescission letter indicating that the individual no longer represents the

Member does not constitute grounds for removal of the Director if all three of the following conditions are satisfied:

- the individual continues activity or employment in the Services Segment of the Electric industry with another Services Segment Member,
- (ii) the individual's new employer is a Segment Member with no representatives holding seats on the Board, and
- (iii) the Board determines that completion of the Director's term of office is in the best interest of NAESB
- c) A vacancy resulting from removal of a Director shall be filled for the remainder of that term in accordance with Section 10.1(i) of these Procedures.

10.5 Designation of Alternates for EC

- a) Selection of Designated Alternates: The Services Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to three (3) Designated Alternates overall.
- b) Designated Alternates will be selected by the same election that determines EC members, with those unsuccessful EC member candidate(s) receiving the greatest number of votes becoming a Designated Alternate in the event that there already are or are expected to be less than three Designated Alternates from the Services Segment in the coming year.
- c) The first, second and third Designated Alternate will be determined by
 - the order in which they become named as a Designated Alternate, as indicated by the first year of their designation and
 - (ii) the order of voting results.
- d) A Designated Alternate will attend and vote at meetings of the EC of the REQ or of the larger EC of NAESB, when one or more regular EC Members from the Services Segment are unable to attend a meeting of the EC and when the absent member or the Vice-Chair of

the REQ EC requests their attendance as a Designated Alternate. Selection of which Alternate will serve will proceed through the order of first, second or third Designated Alternate, with each indicating whether they can or cannot fulfill the requirement to serve at the subject meeting.

10.6 Transitional Election of Officers

- a) The initial Directors and EC Members representing the Services Segment shall be elected at the first meeting of the Segment.
- b) The candidates receiving the greatest number of votes shall be elected.
- c) The two (2) initial Directors and EC Members receiving the greatest number of votes shall serve for two years in Group B positions as described in Sections 10.1 (b) and 10.2 (b) of this procedure. The remaining two (2) initial Directors and EC Members shall serve for one year in Group A positions as described in Sections 10.1 (b) and 10.2 (b) of this procedure.

11.0 Amendment of Services Segment Procedures

The Directors representing the Services Segment on the Board shall recommend to the Segment Membership amendments to these Procedures. Each proposed amendment shall be offered as a separate amendment, except in those situations in which an amendment will affect more than one section of these Procedures. If a proposed amendment will affect more than one section, then all affected sections will be identified.

A majority vote of the Segment Membership shall be required to approve an amendment to these procedures. **TAB 7**

PROPOSED NAESB SEGMENT PROCEDURES - RETAIL GAS QUADRANT

2			Draft For Review & Discussion
3			
4			
5			
6		N	ORTH AMERICAN ENERGY INDUSTRY STANDARDS BOARD
7			RETAIL GAS QUADRANT EXHIBITS
8			
9			
10			SUPPLIER SEGMENT PROCEDURES
11			
12			
13			
14	1.0	MEN	IBERSHIP REQUIREMENTS
15		1.1	The Supplier Segment of the Retail Gas Quadrant is comprised of those parties,
16			including service providers and marketers engaged in the competitive sale of gas
17			to end-users. Any entity which is actively engaged in the above-listed activities
18			which wants to join NAESB may join as a Supplier Segment member of the
19			Retail Gas Quadrant.
20		1.2	NAESB Supplier Segment members become members by submitting their
21			company name, contact name, qualifying business activities, and a statement of

1

intent to join to the Executive Director of NAESB . NAESB Supplier Segment
 members of the Retail Gas Quadrant become voting members of the NAESB by
 paying and maintaining on a current basis their NAESB membership dues, and by
 executing the revocable proxy required by the NAESB incorporation documents.

26

27 2.0 <u>PROCESS FOR ELECTING DIRECTORS AND EXECUTIVE COMMITTEE</u> 28 MEMBERS

29 2.1 <u>Nominations</u>

Each voting NAESB Supplier Segment organization may submit nominations to fill the positions on the Board of Directors and nominations to fill the positions on the Executive Committee Each nominee should be qualified to represent members who are participating in the activities that comprise the Supplier Segment . To qualify, a nominee should have (a) authorization from his or her organization to serve in the position and (b) decision-making authority commensurate with the position's responsibilities.

Nominations shall be sent to the Supplier Segment's Nominating committee. The Nominating committee will confirm each nominee's qualifications and willingness to serve. The ballot will contain the names of all qualified nominees.

If a business activity of the Supplier Segment is not represented among the nominations,
the Nominating committee will endeavor to solicit a qualified nominee involved in that
activity, to be included in the ballot.

42 2.2. <u>Voting</u>

43 Voting will be a one-step process. Each NAESB Supplier Segment voting member
44 organization will be entitled to one ballot.

45 46

47

Each organization will have the same number of votes for members of the Board of Directors and for members of the Executive Committee as there are vacancies. All voting will be at-large. There will be no cumulative voting.

The nominee(s) who receive the highest number of votes for vacancies on the Board of 48 Directors and the Executive Committee, respectively, will serve for terms as determined 49 by the NAESB Retail Gas Ouadrant Board of Directors and the Executive Committee 50 members. The next three (3) nominees to the Executive Committee who receive the 51 highest number of votes, after filling the vacancies above, will serve as alternates. The 52 Board of Directors shall have no alternates. At any given time, business activities of the 53 Supplier Segment may be represented more than once on the Board of Directors or the 54 Executive committee, or both. However, no member organization may have more than 55 one (1) representative among the Retail Gas Quadrant Supplier Segment positions on the 56 Board of Directors and Executive Committee at the same time. 57

58

3.0 **PROCEDURES FOR REMOVAL VACANCY, RESIGNATION OF SEGMENT** 59

60

DIRECTORS AND EXECUTIVE COMMITTEE MEMBERS

Replacement of Board of Directors or Executive Committee members for partial term of 61 office due to resignation or removal of current position holder. 62

- 3.1 Board of Directors 63
- If a member resigns from a position on the Board of Directors, such member's resignation 64 shall be effective upon the later of the date the Executive Director of NAESB receives the 65 written notice of resignation or the effective date of the resignation set forth in the written 66 notice. 67

68

69

70

If a member of the Board of Directors dies, resigns, is terminated or transfers to another organization not included in the Supplier Segment, such member's change in status shall be automatic resignation and effective upon the date of such change.

If an individual member of the Board of Directors transfers to another organization included in the Supplier Segment of the Retail Gas Quadrant, such individual member shall be entitled to remain a member of the Board of Directors; provided the new organization is a voting member of NAESB.

A replacement for the member resigning as set forth above shall be selected by the same 75 procedures as outlined earlier. That is, a nominating committee shall be formed, 76 nominees solicited from Supplier Segment organizations of the Retail Gas Quadrant, 77 nominees qualified by the nominating committee and ballots distributed to NAESB 78 Supplier Segment member organizations. Each member organization shall have one (1) 79 vote and the nominee receiving the most votes shall be elected to complete the then 80 current term. Such person shall be employed by an organization that is part of the 81 Supplier Segment and shall otherwise meet the qualification set forth in Section 2.A, 82 above. The term of such person shall be the remainder of the term of the replaced 83 84 member of the Board of Directors.

85 3.2 <u>Executive Committee</u>

If a member resigns from a position on the Executive Committee, such member's resignation shall be effective upon the later of the date the Executive Director of NAESB receives the written notice of resignation or the effective date of the resignation set forth in the written notice.

90

If a member of the Executive Committee dies, retires, resigns, is terminated or is

transferred to another organization not included in the Supplier Segment, such member's
change in status shall be an automatic resignation and effective upon the date of such
change.

If an individual member of the Executive Committee transfers to another organization included in the Supplier Segment of the Retail Gas Quadrant, such individual member shall be entitled to remain a member of the Executive Committee; provided the new organization is a voting member of NAESB.

A replacement for the member resigning as set forth above shall be selected by the same 98 procedures as outlined earlier. That is, a nominating committee shall be formed, 99 nominees solicited from Supplier Segment organizations, nominees qualified by the 100 nominations committee and ballots distributed to NAESB Supplier Segment members. 101 Each member organization shall have one (1) vote and the nominee receiving the most 102 votes shall be elected to complete the then current term. Such person shall be employed 103 by an organization that is part of the Supplier Segment of the Retail Gas Quadrant and 104 shall have a position similar to other members of the Executive Committee. The term of 105 such person shall be the remainder of the term of the replaced member of the Executive 106 107 Committee.

108

109 4.0 SELECTION OF DESIGNATED ALTERNATES AND PROCESSES FOR THEIR

110

ASSUMPTIONS OF EXECUTIVE COMMITTEE MEMBER DUTIES

4.1 Following each election within the Supplier Segment of the Retail Gas Quadrant
to fill an EC seat, the three highest vote getters not winning a seat will be eligible to
become a Supplier Segment EC Alternate. In the event there are less than three runners-

up to any election, the individuals eligible to be Supplier Segment EC Alternates shall be 114 selected from among the runner(s)-up from the just completed election and those 115 individuals serving as EC alternates at the time of the just completed election; 116 whereupon, the Supplier Segment EC Alternates shall be the three individuals from this 117 combined group whose vote tallies are the highest from the most recent election in which 118 they ran for an EC seat and were a runner-up. In the event of a tie for the position of 119 runner-up, the persons whose vote totals (from the most recent election in which they ran) 120 are tied shall have their names placed in a box and the Executive Director of NAESB 121 shall, in the presence of at least two Supplier Segment EC members, draw names from 122 the box until the three EC Alternate positions are filled. 123

4.2 An EC Alternate may resign at any time. The resignation of an EC Alternate shall
result in an open position until the time of the next election.

4.3 The Segment shall maintain the list of three EC Alternates and shall inform the
NAESB Office from time to time of the identities of the eligible individuals. Following
any election and at least five days tin advance of the next scheduled EC Meeting, the list
of eligible EC Alternates (including any revisions due to new elections or resignations),
shall be forwarded to the NAESB Office.

4.4 An EC Member not attending an EC Meeting (excluding by resignation) and not
submitting a notational ballot to the NAESB Office shall identify to NAESB and to their
designated EC Alternate (in writing or facsimile) the specific Alternate they designate to
sit and vote as an EC Member of the Supplier Segment at the identified EC Meeting.
Failure of an absent EC Member not submitting a notational ballot to so identify a
specific EC Alternate shall mean that those Supplier Segment EC Members attending the

- subject EC Meeting may choose (an) individuals (s) from among the eligible Supplier
 Segment EC Alternates in attendance to sit and vote as (an) EC Member(s) of the
 Supplier Segment during the subject EC Meeting.
- 4.5 Failure of an EC Member to attend an EC Meeting by means of that EC Member's 140 resignation shall create a vacancy which may be filled during an EC Meeting by an 141 Alternate pursuant to the procedures in this paragraph 4.5 until an election has occurred. 142 A vacant position on the EC shall be filled as soon possible under the Supplier Segment 143 election procedures set forth from time to time. In the event resignation has occurred, 144 and, there has not been sufficient time to hold an election for the vacant position; and, a 145 scheduled EC Meeting is in session; then those Supplier Segment EC Members attending 146 the subject EC Meeting may choose (an) individual(s) from among the eligible Supplier 147 Segment EC Alternates in attendance to sit and vote as (an) EC Members(s) of the 148 Supplier Segment during the subject EC Meeting. 149
- 150

151 5.0 **PROCEDURES FOR AMENDING THESE SEGMENT PROCEDURES**

- The Supplier Segment of the Retail Gas Quadrant has no established process for amending these segment specific procedures but contemplates that changes, if deemed necessary, would be initiated by the Supplier Segment's Board of Director members. The Suppler Segment will promptly advise NAESB of any change to these procedures.
- 156

157 6.0 COMMUNICATION PROCESSES FOR MEETING NOTICES

158The Supplier Segment intends to rely upon NAESB's own communication procedures159for the distribution of meeting notices. The Supplier Segment of the Retail Gas

160		Quadr	ant believes NAESB's use of an electronic bulletin board, regular newsletter,							
161		other	other electronic means and adherence to standard notice procedures including distribution							
162		of not	of notices to all interested parties will be sufficient to advise Supplier Segment member							
163		organi	organizations of NAESB activities.							
164										
165	7.0	EXEC	CUTIVE SUMMARY							
166		7.1	To belong to the Supplier Segment, and be entitled to one (1) vote in Retail Gas							
167			quadrant matters, a person or entity, including service providers and marketers,							
168			must be engaged in the competitive sale of gas to end-users.							
169		7.2	Board, EC, and Alternates shall be elected by the vote of the Supplier Segment							
170			membership. The slate of candidates shall be determined by a Nominating							
171			Committee made up of Segment members.							
172		7.3	Supplier Segment may have up to three (3) Designated Alternates for the EC.							
173		7.4	The Board of Directors will have no Designated Alternates.							
174		7.5	If an individual member of the EC or Board of Directors transfers to another							
175			organization included in the Supplier Segment, such individual member shall be							
176			entitled to remain a member of the EC or Board of Directors; provided the new							
177			organization is a voting member of NAESB.							

Proposed NAESB RGQ Distributors Segment ProceduresDRAFT 01/25/02

1 2		RETAIL GAS QUADRANT DISTRIBUTORS SEGMENT PROCEDURES
3 4	1.0	DESCRIPTION OF THE DISTRIBUTORS SEGMENT:
5		The Distributors Segment of the Retail Gas Quadrant of the North American Energy Standards
6		Board ("NAESB") is one of four (4) co-equal Segments comprising the retail natural gas market.
7		The Distributors Segment represents the interests of entities engaged in the regulated or publicly
8		owned local distribution of natural gas.
9		
10	2.0	DISTRIBUTORS SEGMENT MEMBERSHIP REQUIREMENTS:
11		Membership in the Distributors Segment of the Retail Gas Quadrant of NAESB shall be open to
12		persons that meet the following requirements:
13		a. The person meets the general membership criteria set forth in the NAESB Bylaws (Article 5)
14		and the NAESB Certificate of Incorporation (Article IV Sections 1 & 2).
15		b. The person meets the membership requirements specific to the Retail Gas Quadrant of
16		NAESB.
17		c. The person is engaged in the regulated or publicly owned local distribution of natural gas.
18		d. The person has submitted an application to the Retail Gas Quadrant, declaring its Segment
19		affiliation.
20		
21	3.0	AMENDMENT OF SEGMENT PROCEDURES:
22		Amendment of these Retail Gas Quadrant Distributors Segment Procedures shall be subject to
23		approval by 2/3 of the NAESB Retail Gas Quadrant Distributors Segment Members.
24		
25	4.0	DISSEMINATION OF INFORMATION:
26		The Distributors Segment shall designate a point of contact for their Segment to disseminate

Prepared for the Retail Gas Quadrant Distributors Segment's Consideration - Page2

- 27 various information and notices to the Segment Membership, its Quadrant, other Quadrants, and
- 28 the NAESB office regarding issues specific to the Distributors Segment.

1	
2	DRAFT For Discussion
3	
4	Bill Boswell's comments highlighted throughout
5	Barbara Alexander Edits 12/21/01
6	
7	
8	NORTH AMERICAN ENERGY STANDARDS BOARD
9	BYLAWS ADDENDUM
10	
11	EXHIBIT 3-D
12	
13	RETAIL NATURAL GAS QUADRANT END USER SEGMENT
14	PROCEDURES

15		
16		
17	1.0	Description of the End User Segment
18		The End User Segment of the Retail Natural Gas Quadrant of the North
19		American Standards Board ("NAESB") is one of four (4) co-equal segments
20		comprising the retail natural gas market. The End User Segment
21		membership is comprised of businesses, organizations, government
22		agencies, or individuals that consume natural gas at facilities served by
23		Natural Gas Distributors, and parties who represent such consumers of
24		natural gas. The End User Segment shall have a minimum of five paid
25		members at all times.
26		
26 27	2.0	DEFINITIONS [definitions highlighted are likely to be redundant with
	2.0	DEFINITIONS [definitions highlighted are likely to be redundant with both NAESB Bylaws and Quadrant Procedures, and need not be defined
27	2.0	
27 28	2.0	both NAESB Bylaws and Quadrant Procedures, and need not be defined
27 28 29	2.0	both NAESB Bylaws and Quadrant Procedures, and need not be defined here again, particularly if we move the closing statement for this Section
27 28 29 30	2.0	both NAESB Bylaws and Quadrant Procedures, and need not be defined here again, particularly if we move the closing statement for this Section to the beginning] <u>I think the document should be a "stand alone"</u>
27 28 29 30 31	2.0	both NAESB Bylaws and Quadrant Procedures, and need not be defined here again, particularly if we move the closing statement for this Section to the beginning] <u>I think the document should be a "stand alone"</u> <u>document that makes sense to outsiders and insiders. Therefore, I would</u>
 27 28 29 30 31 32 	2.0	both NAESB Bylaws and Quadrant Procedures, and need not be defined here again, particularly if we move the closing statement for this Section to the beginning] <u>I think the document should be a "stand alone"</u> <u>document that makes sense to outsiders and insiders. Therefore, I would</u> <u>not delete the definitions and procedures that make sure that the</u>
 27 28 29 30 31 32 33 	2.0	both NAESB Bylaws and Quadrant Procedures, and need not be defined here again, particularly if we move the closing statement for this Section to the beginning] <u>I think the document should be a "stand alone"</u> <u>document that makes sense to outsiders and insiders. Therefore, I would</u> <u>not delete the definitions and procedures that make sure that the</u> <u>document "works" even if those provisions are duplicative of the NAESB</u>
 27 28 29 30 31 32 33 34 	2.0	both NAESB Bylaws and Quadrant Procedures, and need not be defined here again, particularly if we move the closing statement for this Section to the beginning]_I think the document should be a "stand alone" document that makes sense to outsiders and insiders. Therefore, I would not delete the definitions and procedures that make sure that the document "works" even if those provisions are duplicative of the NAESB Bylaws or Quadrant Procedures.

- 38 "Board" means the Board of Directors of NAESB as defined in the NAESB39 bylaws.
- "Bylaws" means the corporate bylaws of NAESB as amended from time totime.
- 42 "EC" means the Executive Committee of the Retail Natural Gas Quadrant
- 43 or NAESB, as appropriate to the context, as defined in the Bylaws and
- 44 these Retail Natural Gas Quadrant Procedures.
- 45 "Director" means an individual serving on the Board.
- 46 "NAESB" means the North American Energy Standards Board
- 47 "NAESB Office" means the office of the Secretary of NAESB
- 48 "Natural Gas Consumer" means an individual, partnership, corporation, or
- 49 other entity consuming natural gas at one or more facilities served by a
- 50 Natural Gas Distributor.
- ⁵¹ "Natural Gas Distributor" means an individual, partnership, corporation, or
- 52 other entity engaged in the local distribution of natural gas. Most often
- 53 this will be an investor owned utility or a municipal utility.
- ⁵⁴ "RGQ" means the Retail Natural Gas Quadrant of NAESB.
- ⁵⁵ "Segment" means one of the co-equal membership segments of RGQ. As
- 56 currently defined, RGQ has the following four (4) segments of the retail
- 57 natural gas market: Suppliers, Distributors, End-Users, and Services.
- ⁵⁸ "Segment Member" means an individual or entity (e.g., corporation,
- 59 government agency or partnership) that satisfy the Requirements for
- 60 membership set forth in the Bylaws of RGQ, and further satisfy the

61	qualifications for End User Segment membership set forth in Section 3 of
62	these procedures. If the Segment Member is not an individual, the term
63	Segment Member shall include that entity's representative (employee,
64	officer, director, or partner) who is authorized to cast that Segment
65	member's vote.
66	"Segment Membership" means the Segment Members collectively.
67	
68	All other capitalized terms, if not defined in this Section, shall have the same
69	definitions as specified in the Bylaws of NAESB or in the RGQ procedures. [Good
70	general statement, maybe should be at the beginning of the Section.]
71	
72	3.0 MEMBERSHIP REQUIREMENTS
73	(a) Membership in the End User Segment of the Retail Natural Gas
74	Quadrant of the NAESB shall be open to any person or legal entity (e.g.
75	corporation, partnership) that meets the following requirements:
76	<u>1)</u> a)—The person or entity meets the general membership criteria set
77	forth in the Bylaws (Article 5) and the Certificate of Incorporation
78	(Article IV Sections 1 & 2)-:
79	2) The person or entity meets the membership Requirements specific
80	to the RGQ of NAESB-
81	<u>3)</u> e) The person or entity participates in the retail natural gas
82	marketplace in one of the following ways:
83	(i) as a consumer of natural gas delivered by either a regulated

84	investor	owned	or	а	municipal	natural	gas	distribution
85	company	· •						

- 86 (ii) as a consumer of natural gas delivered by an interstate
 87 transmission pipeline regulated by the Federal Energy
 88 Regulatory Commission in the United States or the equivalent
 89 in other nations; or
- 90 (iii) as a statutorily-appointed Federal, state, territorial or district
 91 consumer advocate or agency; and
- 92 (+)4) The person or entity has submitted an application to the Retail
 93 Natural Gas Quadrant, declaring End User Segment affiliation_and
 94 paid the appropriate membership dues for the RGQ_End_User
 95 Segment of NAESB. An applicant for membership that represents
 96 residential consumers may apply for a reduced dues in accordance
 97 with the provisions of the RGQ_Procedures -
- e) The person or entity must have paid its membership dues for the
 RGQ End User Segment of NAESB. A Segment Member's voting
 rights shall be suspended for any period during which the Member is
 delinquent in the payment of its membership dues. [redundant with
 Bylaws Requirements; not needed here]
- 103 f(b) Upon the approval of current members of the End User Segment, and 104 subject to the Requirements of NAESB and the RGQ, Trade Associations 105 that represent the interests of natural gas consumers not otherwise 106 represented, may be admitted for membership. Such Trade Association

107 members shall have the same rights and responsibilities as any Segment 108 Member, except that their membership is subject to periodic 109 reconfirmation by the other Segment members. The criteria that Segment 110 Members should use in evaluating such a party for membership is that by admitting such a member, the interests of other end users who can't 111 participate directly (such as individual residential customers) will be 112 represented. Approval of this Trade Association representation needs to 113 be reconfirmed every two (2) years at a minimum, if not otherwise 114 115 provided for in the RGQ Procedures contained in this Exhibit 4 to the Bylaws. [Also discussed in RGQ draft Procedures --- duplicative?] 116

117 g)c) In the event that there are not five existing members of the End User
118 Segment, at least two of whom represent residential consumers,
119 membership approval may be granted and reconfirmed to Trade
120 Associations by the elected members of the RGQ Executive Committee
121 until such time as there are five non-Trade Association members, at which
122 point the non-Trade Association members will assume responsibility for
123 reconfirmation of Trade Association rights to continued membership.

h)d) Corporate Members do not extend their membership to their parent
 company, affiliates or subsidiaries. Separate corporate entities within a
 corporate organization may hold multiple memberships in the End User
 Segment, but each must join individually and pay a membership fee.

i)e) All representatives of any Segment Member must be duly authorized
 by the Member entity in a letter to the Segment Chair, such

131 i)f) Representatives from a single corporate organization may hold no more than one Executive Committee and one Board of Directors 132 133 seat from the Segment. (k) No Segment Member may have more than one representative from 134 their organization on the Board of Directors. 135 136 END USER SEGMENT ORGANIZATION 4.0 137 138 The End User Segment Members shall elect Members of the End User Segment to the following administrative and governance positions: 139 Chair and Vice-Chair 140 a) The Chair and Vice-Chair will preside over all meetings of the 141 142 general Segment membership separate from meetings of the RGQ. 143 The Chair will appoint a Secretary of the meeting(s), which may be a standing position, to take meeting minutes and record votes. 144 145 Minutes of all meetings, including voting results, are to be reported 146 by the Chair, the Vice-Chair, or the Secretary at the Chair's discretion, to the Members and transmitted to the NAESB Office for 147 publication within 15 days of the date of the meeting. 148

authorization is to be forwarded to the NAESB Office.

130

The Vice-Chair will organize and conduct all Segment elections once the Segment has commenced operation. The method of initial election of all officers is addressed in Section 6.8.

152 The term of office of the Chair is one year concurrent with the

operating year of NAESB. The Chair will be succeeded in the following year by the Vice-Chair, with a new Vice-Chair elected to fill the vacant Vice-Chair position. There is no limit on the number of terms any End User Segment Member may serve either as Chair or Vice-Chair.

b) Executive Committee members

159 The Segment Membership will elect two Group A or two Group B 160 Executive Committee members in alternating years as provided for 161 in the RGQ Procedures.

162 c) Directors

163The Segment Membership will elect two Group A or two Group B164Directors in alternating years as provided for in the RGQ Procedures.

165

166 **5.0 MEETINGS**

All meetings held in association with NAESB, the RGQ, or within the End User Segment of RGQ are open to any interested person or entity. Only Segment Members, however, shall have a right to vote at such meetings on behalf of *["having this phrase here is critical to this statement being workable"]* the End User Segment.

172

173 5.1 Notices of meetings [drop; subsections 5.1 and 5.2 are both NAESBops
174 issues]

175 Notices of meetings will be posted to the End User Segment membership

- no less than seven (7) days prior to the date of the meeting, said notices
 providing the location, date, times and agenda topics of the meeting.
- 178

179 5.2 Notification Method

Notification will be through the use of an electronic bulletin board, regular
newsletter, and adherence to standard notice a) The NAESB Office shall
advise RGQ End User Segment Member organizations of NAESB activities.
As such the End User Segment intends to rely upon the NAESB Office to
distribute meeting notices, as required under Article 11 of the NAESB
Bylaws, to Segment Members and other interested parties as deemed
appropriate.

- b) If the NAESB Office cannot perform this function, then the Segment Membership will determine an agreed method of notification or reconfirm the agreement at each first meeting of the operating year of the Segment Membership.
- 191

192 5.3 Notices of Elections ["OK"{CLD -- but probably now should be dropped
193 as it is covered in Section 6}]

194 Notices of elections shall be posted as provided for in Section 6 of these195 Procedures.

196

ELECTION, RESIGNATION AND REMOVAL OF OFFICERS OF THE
 END USER SEGMENT [Section 6 is an "awfully busy section", needs to

199 *be simplified*]

200 6.1 <u>Election of Directors</u>:

- a) The End User Segment of the RGQ shall be represented on the
 NAESB Board of Directors by four individuals who shall be elected,
 from time to time, as required by Article 7 of the NAESB Bylaws and
 in this Exhibit 4 to those Bylaws. <u>At least one of these individuals</u>
 shall represent residential consumers.
- b) Each Director from the End User Segment of the RGQ shall serve a term of two years on the NAESB Board, unless provided for otherwise in the RGQ Procedures.
- 209 (i) Directors elected in a year ending in an odd digit (i.e., 2003),
 210 whose term would expire in a year ending in an odd digit
 211 (i.e., 2005), will serve as Group A Directors; there will be two
 212 Group A Director seats at any given time, subject to the
 213 conditions specified in Section 6.8.
- 214 (ii) Directors elected in a year ending in an even digit, whose
 215 term would expire in a year ending in an even digit, will serve
 216 as Group B Directors; there will be two Group B Director seats
 217 at any given time, subject to the conditions specified in
 218 Section 6.8.
- c) To be eligible to serve as an End User Segment representative on
 the NAESB Board of Directors, the individual must
- (i) be a Segment Member

(ii) have a working knowledge of the NAESB process;

- (iii) be willing to commit the time and resources necessary to
 fulfill their obligations as a NAESB Board member and to meet
 the minimum threshold of participation and attendance
 established in the NAESB Bylaws
- 227 (iv) disclose their interest, or their employer's interest, in the 228 natural gas industry and the relationship with other entities 229 with which the employer may be affiliated;
- d) No individual representing any single Segment Member shall serve
 on the Board of Directors and on the RGQ Executive Committee
 concurrently.
- e) Interested individuals duly authorized to represent a Segment
 Member and eligible to serve on the Board of Directors shall submit
 their names to the NAESB Office in a form as specified in NAESB
 Operating Procedures or,
- (i) if such Requirements are not provided for in the NAESB
 Operating Procedures, interested individuals shall submit a
 written statement to the NAESB Office that provides their
 name, title and affiliation (if any), business mail and email
 addresses, business telephone, a statement that they accept
 the conditions referred to in subsection (c)-(iii) above, and
 the statement required in subsection (c)-(iv) above.
- *f)* In preparation for any election of NAESB Directors (other than initial

245 Directors, as provided for in Section 6.8) representing the End User Segment of the RGQ, a Nominating Committee consisting of five (5) 246 existing Board and EC members of the RGQ End User Segment, 247 248 shall develop a slate of candidates from interested members of the Segment Membership. Other nominations may be made at or prior 249 to the close of the RGQ End User Segment nomination period by any 250 Segment Member in accordance with the Requirements in 251 subsection (e) above. All nominations must be made and conveyed 252 253 in writing to the NAESB Secretary no less than 31 days prior to the election date. [this really belongs in the Quadrant Procedures to 254 keep all Segments working the same way as much as possible] 255

- 256 g) The election process for RGQ End User Segment representatives to257 the NAESB Board shall be
- (i) coordinated by the Vice-Chair of the Segment with
 appropriate coordination with the Quadrant Executive
 Committee and the NAESB Office such that elections occur
 during the second week of November of each year
- 262 (ii) announced by formal notice to the Segment Membership no
 263 less than 30 days prior to the date of the election, with such
 264 notice indicating the name and affiliation of all candidates.
- 265 h) Elections for RGQ End User Segment Directors shall be subject to266 the following provisions:

267

(i) Any End User Segment Member of the RGQ with dues current

- 268 is eligible to have one duly authorized representative cast the269 Member's vote.
- 270 (ii) End User Segment Members shall cast votes for candidates in
 271 a number not exceeding the number of Director seats open
 272 for election from the End User Segment.
- 273 (iii) The candidates receiving the greatest numbers of votes shall274 be elected.
- (iv) Results of the RGQ elections for the NAESB Board, including
 the returned ballots, shall be forwarded to the Secretary of
 NAESB, for the purpose of public record and validation of
 voting results, upon conclusion of any election.
- i) Any Director seats which become vacant shall be filled in accordance
 with RGQ Procedures as specified in this Exhibit 4 to the Bylaws, or
 if such provisions are lacking, by special election, subject to the
 following provisions
- (i) If the vacancy occurs with less than 125 days remaining in
 the Director's term of office, the seat will remain vacant until
 filled by election of a new Director to a term commencing in
 the next operating year.
- 287 (ii) All other vacancies will be filled by special election; such
 288 special elections are to be formally noticed to the Segment
 289 Membership by the Segment Vice-Chair or Chair within 15
 290 days of when the creation of the vacancy becomes known and

held within 60 days of the creation of the vacancy.
Candidates will be designated by the normal nomination
process or selected by the remaining Board members of the
Segment from Segment Membership. Candidates elected to
fill the vacancy will serve the remainder of the term, and may
stand for re-election.

297

298

6.2 Resignation of Directors

A Director may resign his or her directorship by submitting a letter to the Secretary of NAESB with a copy to the Chair of the End User Segment of the RGQ stating that he or she is resigning and giving the effective date of the resignation.

303

304 6.3 Removal of Directors

a) A NAESB Director shall cease to be a Director upon

306 (i) the resignation of RGQ End User Segment Membership by the
 307 entity of which the Director serves as a representative

308 (ii) the lapse of the Segment Membership dues of the entity
309 which the Director serves as a representative, or

310 (iii) the Director's resignation, removal, or death

311 (iv) the first day of the NAESB operating year following an
312 election for the seat held by the Director, wherein the
313 Director was not re-elected by the Segment Membership as

314 declared by the NAESB Secretary. b) Receipt by the NAESB Secretary of a rescission letter indicating that 315 the individual no longer represents the Member does not constitute 316 grounds for removal of the Director if all three of the following 317 conditions are satisfied: 318 (i) the individual continues activity or employment in the End 319 User Segment of the natural gas industry with another End 320 User Segment Member, 321 322 (ii) the individual's new employer is a Segment Member with no representatives holding seats on the Board, and 323 (iii) the Board determines that completion of the Director's term 324 of office is in the best interest of NAESB 325 A vacancy resulting from removal of a Director shall be filled for the 326 c) 327 remainder of that term in accordance with Section 6.1(i) of these Procedures. 328 [the contents of 6.3 should be probably be covered within the Quadrant 329 330 Procedures document, the Bylaws or NAESBops; good to lay it out, but not a good thing to have individual Segments following significantly different 331 approaches on this] 332 333 6.4 Election of Executive Committee Members 334 335 The End User Segment of the RGQ shall be represented on the NAESB EC

by four individuals who shall be elected, from time to time, as RGQuired by

336

Page 15 of 25

- Article 10 of the NAESB Bylaws and in this Exhibit 4 to those Bylaws. <u>At</u>
 least one of these individuals shall represent residential consumers.
- a) Each EC member from the End User Segment of the RGQ shall serve
 a term of two years on the RGQ EC, unless provided for otherwise in
 the RGQ Procedures.
- 342 (iii) Members elected in a year ending in an odd digit (i.e., 2003),
 343 whose term would expire in a year ending in an odd digit
 344 (i.e., 2005), will serve as Group A representatives; there will
 345 be two Group A representative seats at any given time,
 346 subject to the conditions specified in Section 6.8.
- (iv) Members elected in a year ending in an even digit, whose
 term would expire in a year ending in an even digit, will serve
 as Group B representatives; there will be two Group B
 representative seats at any given time, subject to the
 conditions specified in Section 6.8.
- b) To be eligible to serve as an End User Segment representative on
 the EC, the individual must
- 354 (i) be a Segment Member
- 355 (ii) have a working knowledge of the NAESB process;
- 356 (iii) be willing to commit the time and resources necessary to
 357 fulfill their obligations as a NAESB EC member and to meet
 358 the minimum threshold of participation and attendance
 359 established in the NAESB Bylaws

- 360 (iv) disclose their interest, or their employer's interest, in the
 361 natural gas industry and the relationship with other entities
 362 with which the employer may be affiliated;
- 363 c) No individual representing any single Segment Member shall serve
 364 on the Board of Directors and on the RGQ Executive Committee
 365 concurrently.
- d) Interested individuals duly authorized to represent a Segment
 Member and eligible to serve on the EC shall submit their names to
 the NAESB Office in a form as specified in NAESB Operating
 Procedures or,
- (i) if such Requirements are not provided for in the NAESB
 Operating Procedures, interested individuals shall submit a
 written statement to the NAESB Office that provides their
 name, title and affiliation (if any), business mail and email
 addresses, business telephone, a statement that they accept
 the conditions referred to in subsection (c)-(iii) above, and
 the statement required in subsection (c)-(iv) above.
- e) In preparation for any election of EC members (other than initial EC members, as provided for in Section 6.8) representing the End User
 Segment of the RGQ, a Nominating Committee consisting of five (5)
 Board or EC members of the RGQ End User Segment, shall develop
 a slate of candidates from interested members of the Segment
 Membership. Other nominations may be made at or prior to the

close of the RGQ End User Segment nominating period by any
Segment Member in accordance with the Requirements in
subsection (e) above. All nominations must be made and conveyed
in writing to the NAESB Secretary no less than 31 days prior to the
election date.

- f) The election process for RGQ End User Segment representatives to
 the RGQ EC shall be
- 390 (iii) coordinated by the Vice-Chair of the Segment with
 391 appropriate coordination with the RGQ EC and the NAESB
 392 Office such that elections occur during the second week of
 393 November of each year
- (iv) announced by formal notice to the Segment Membership no
 less than 30 days prior to the date of the election, with such
 notice indicating the name and affiliation of all candidates.
- 397 g) Elections for RGQ End User Segment EC members shall be subject
 398 to the following provisions:
- 399 (i) Any End User Segment Member of the RGQ with dues current
 400 is eligible to have one duly authorized representative cast the
 401 Member's vote.
- 402 (ii) End User Segment Members shall cast votes for candidates in
 403 a number not exceeding the number of EC seats open for
 404 election by the End User Segment.
- 405 (iii) The candidates receiving the greatest numbers of votes shall

```
406 be el
```

be elected.

- 407 (iv) Results of the elections for the RGQ EC, including the
 408 returned ballots, shall be forwarded to the Secretary of
 409 NAESB, for the purpose of public record and validation of
 410 voting results, upon conclusion of any election.
- h) Any EC seats becoming vacant shall be filled in accordance with RGQ
 Procedures as specified in this Exhibit 4 to the Bylaws, or if such
 provisions are lacking, by a special election subject to the following
 provisions
- (i) If the vacancy occurs with less than 125 days remaining in
 the EC member's term of office, the seat will remain vacant
 until filled by election of a new EC member to a term
 commencing in the next operating year, and a Designated
 Alternate will serve until expiration of the term.
- (ii) All other vacancies will be filled by special election; such 420 421 special elections are to be formally noticed to the Segment 422 Membership by the Segment Vice-Chair or Chair within 15 days of when the creation of the vacancy becomes known and 423 held within 60 days of the creation of the vacancy. 424 Candidates will be designated by the normal nomination 425 process or selected by the remaining EC members of the 426 Segment from Segment Membership. Candidates elected to 427 fill the vacancy will serve the remainder of the term, and may 428

429				stand for re-election.			
430			(iii)	Until an election is held and a new EC member is elected, a			
431				Designated Alternate will serve in the open EC seat.			
432	[the highlighted parts of this section are redundant with what should be in the						
433	Quad	rant Pr	ocedur	res; unnecessary]			
434							
435	6.5	Resig	Inatio	n of EC Representatives			
436		An EC	c mem	ber may resign his or her seat by submitting a letter to the			
437		Secre	tary of	NAESB with a copy to the Chair of the End User Segment of			
438		the R	GQ sta	ting that he or she is resigning and giving the effective date of			
439		the re	signat	ion.			
440	[all of the remaining portions of Section 6 belong in the Quadrant Procedures so						
441	that a	all Segr	ments	operate the same in regards to these issues]			
442							
443	6.6	Remo	oval of	EC Representatives			
444		a)	An EC	member shall cease to be a member of the EC upon			
445			(i)	the resignation of RGQ End User Segment Membership by the			
446				entity of which the EC member serves as a representative			
447			(ii)	the lapse of the Segment Membership dues of the entity			
448				which the EC member serves as a representative,			
449			(iii)	the EC member's resignation, removal, or death			
450			(iv)	the first day of the NAESB operating year following an			
451				election for the seat held by the EC member, wherein the			

452			member was not re-elected by the Segment Membership as
453			declared by the NAESB Secretary
454			(iv) receipt by the NAESB Secretary of a rescission letter
455			indicating that the individual no longer represents the
456			Segment Member
457		b)	A vacancy resulting from removal of an EC member shall be filled for
458			the remainder of that term in accordance with Section 6.4(h) of
459			these Procedures.
460			
461	6.7	Desi	gnation of Alternates for Executive Committee Members
462		<u>a)</u>	Selection of Designated Alternates:
463			The End User Segment will select no more than three (3), and no
464			les than one Designated Alternate EC representative in each election
465			year, allowing for up to three (3) Designated Alternates overall.
466		b)	Designated Alternates will be selected by the same election that
467			determines EC members, with those unsuccessful EC member
468			candidate(s) receiving the greatest number of votes becoming a
469			Designated Alternate in the event that there already are or are
470			expected to be less than three Designated Alternates from the End
471			User Segment in the coming year.
472		c)	The first, second and third Designated Alternate will be determined
473			by
474			(i) the order in which they become named as a Designated
- 475 Alternate, as indicated by the first year of their designation 476 and
- 477 (ii) the order of voting results.
- 478 d) A Designated Alternate will attend and vote at meetings of the EC of the RGQ or of the larger EC of NAESB, when one or more regular EC 479 Members from the End User Segment are unable to attend a 480 meeting of the EC and when the absent member or the Vice-Chair of 481 482 the RGQ EC requests their attendance as a Designated Alternate. 483 Selection of which Alternate will serve will proceed through the order of first, second or third Designated Alternate, with each indicating 484 485 whether they can or cannot fulfill the requirement to serve at the 486 subject meeting.

487

488 6.8 Transitional Election of Segment Officers

489 The Requirements of the NAESB Bylaws for a minimum of 5 Members per 490 Segment and 40 Members per Quadrant do not provide for enough 491 members to satisfy the constraints on election of Directors and EC members as stated in earlier subsections of this Section 6. Consequently, 492 493 until the End-User Segment has at least eight (8) Members, the number of Directors and EC Members elected by the Segment Membership within the 494 Quadrant will be halved, to 2 Directors and 2 EC members with no dilution 495 of the Segment's balanced voting within the Quadrant. 496

497

- The following provisions will apply and supercede the requirements of Sections 6.1 through 6.6 until the End-User Segment has enrolled a minimum of eight (8) members.
- 501

502 If not otherwise provided for in the RGQ Procedures, the following 503 provisions will apply during the initial operating period of the End User 504 Segment until the Segment has at least eight (8) paid Members.

- a) Term of Office of Initial Directors
- 506 The initial group of two (2) Directors will both serve a 2 year term of 507 office as either Group A or Group B Directors, as determined by the 508 expiration year of their original term of office.
- 509 b) Interim Elections of Directors
- 510 Within 60 days of receipt of notification from the NAESB Office that 511 the End-User Segment has achieved eight (8) members, an election 512 shall be held in order to select of two additional Directors to serve 513 alongside the initially elected Directors
- 514(i)the term of office of the second group elected at this time will515expire one year later than the term of the initially elected516Directors, thereby allowing for two Group A and two Group B
- 517 Directors from the Segment.
- 518 c) Term of Office of Initial Executive Committee Members

- 519 The initial group of EC members will serve a 2 year term of office as either Group A or Group B members, as determined by the 520 expiration year of their original term of office. 521 Interim Elections of EC Members 522 d) Within 60 days of receipt of notification from the NAESB Office that 523 the End-User Segment has achieved eight (8) members an election 524 shall be held in order to select of two additional EC Members to 525 serve alongside the initially elected Directors 526 527 (i) The term of office of the second group elected at this time will expire one year later than the term of the initially elected EC 528 members, thereby allowing for two Group A and two Group B 529 EC members from each Segment. 530 The previously elected Chair and Vice-Chair of the Quadrant 531 (ii) EC will continue in their official role until either their re-532 election at the end of their term or removal from office in 533 534 accordance with the NAESB Bylaws, Quadrant Procedures or 535 these Procedures.
 - 536

537 **7.0 AMENDMENT OF SEGMENT PROCEDURES**

Amendment of these RGQ End User Segment Procedures shall be subject to approval by a 90% majority of the Segment Members. *[at first blush, this seemed awfully high to Bill; as we discussed the undesirability of having Segments change their procedures frequently, he came around to* 542 maybe 80-90% is an appropriate number; up to the Segment].



RETAIL GAS QUADRANT SERVICES SEGMENT PROCEDURES

1.0 Table of Contents

1.0 Table of Contents	. 1
2.0 Definitions	
3.0 Description of the Services Segment	. 2
4.0 Services Segment Membership Requirements	
5.0 Services Segment Funding	. 3
6.0 Services Segment Organization	. 3
6.1 Chair and Vice-Chair	3
6.2 Executive Committee (EC) Members	4
6.3 Directors	4
7.0 Determination of Quorum	
8.0 Meetings	
9.0 Services Segment Notifications	. 4
9.1 Notices of Meetings	4
9.2 Notification Method	4
9.3 Notices of Elections	5
10.0 Election, Resignation, and Removal of Services Segment Officers	
10.1 Election of Directors	5
10.2 Election of EC Members	8
	11
	11
	12
	13
11.0 Amendment of Services Segment Procedures	13

2.0 Definitions

Board - means the Board of Directors of NAESB as defined in the NAESB bylaws.

Bylaws - means the corporate bylaws of NAESB as amended from time to time.

<u>EC</u> - means the Executive Committee of the Retail Gas Quadrant or NAESB, as appropriate to the context, as defined in the Bylaws and these Retail Gas Quadrant Procedures.

Director - means an individual serving on the Board.

NAESB - means the North American Energy Standards Board

NAESB Office - means the office of the Secretary of NAESB

RGQ - means the Retail Gas Quadrant of NAESB.

<u>Segment</u> - means one of the co-equal membership segments of RGQ. As currently defined, RGQ has the following four (4) segments of the retail Gas market: Suppliers, Distributors, End-Users, and Services.

<u>Segment Member</u> - means an individual or entity (e.g., corporation, government agency or partnership) that satisfies

the requirements for membership set forth in the Bylaws of RGQ, and further satisfies the qualifications for Services Segment membership set forth in Section 3 of these procedures. If the Segment Member is not an individual, the term Segment Member shall include that entity's representative (employee, officer, director, or partner) who is authorized to cast that Segment member's vote.

<u>Segment Membership</u> - means the Segment Members collectively.

All other capitalized terms, if not defined in this Section, shall have the same definitions as specified in the Bylaws of NAESB or in the RGQ procedures.

3.0 Description of the Services Segment

The Services Segment of the Retail Gas Quadrant of the North American Energy Standards Board ("NAESB") is one of four (4) co-equal segments comprising the retail Gas market. The Services Segment membership is comprised of businesses that provide services to suppliers and distributors of gas or an entity not otherwise declared in another segment. The Services Segment shall have a minimum of five paid members at all times.

4.0 Services Segment Membership Requirements

- Membership in the Services Segment of the Retail Gas Quadrant of the NAESB shall be open to any entity (e.g., corporation, partnership or proprietorship) that meets the following requirements:
 - a) The entity meets the general membership criteria set forth in the Bylaws (Article 5) and the Certificate of Incorporation (Article IV Sections 1 & 2).
 - b) The entity meets the membership requirements specific to the RGQ of NAESB.
 - c) The entity provides services to participants in the retail Gas industry.
 - d) The entity is not otherwise declared in another segment.
 - e) The entity has submitted an application to the RGQ, declaring Services Segment affiliation.
 - f) Corporate Members do not extend their membership to their parent company, affiliates or subsidiaries. Separate

corporate entities within a corporate organization may hold multiple memberships in the Services Segment, but each must join individually and pay a membership fee.

- g) All representatives of any Services Segment Member must be duly authorized by the Member entity in a letter to the Segment Chair, such authorization is to be forwarded to the NAESB Office.
- Representatives from a single corporate organization may hold no more than one Executive Committee and one Board of Directors seat from the Segment.
- i) No Segment Member may have more than one representative from their organization on the Board of Directors.
- j) The entity satisfies the requirements for Membership set forth in Article 5 of the Bylaws of NAESB, and Section 3 of the Retail Gas Quadrant Procedures.

5.0 Services Segment Funding

The Services Segment shall be funded through allocation of the Segment's NAESB dues as established by the NAESB Board of Directors

6.0 Services Segment Organization

The Services Segment Members shall elect Members of the Services Segment to the following administrative and governance positions:

6.1 Chair and Vice-Chair

- The Chair and Vice-Chair will preside over all meetings of the general Services Segment membership separate from meetings of the RGQ. The Chair will appoint a Secretary of the meeting(s), which may be a standing position, to take meeting minutes and record votes. Minutes of all meetings, including voting results, are to be reported by the Chair, the Vice-Chair, or the Secretary at the Chair's discretion, to the Members and transmitted to the NAESB Office for publication within 15 days of the date of the meeting.
- The Vice-Chair will organize and conduct all Segment elections once the Segment has commenced operation. The method of initial election of all officers is addressed in Section 8.6

Transitional Election of Officers of this document.

The term of office of the Chair is one year concurrent with the operating year of NAESB. The Chair will be succeeded in the following year by the Vice-Chair, with a new Vice-Chair elected to fill the vacant Vice-Chair position. There is no limit on the number of terms any Services Segment Member may serve either as Chair or Vice-Chair.

6.2 Executive Committee (EC) Members

Four (4) persons who are individual Segment Members will represent the Services Segment on the RGQ EC. The Segment Membership will elect Executive Committee members as provided for in the RGQ Procedures and Section 10.2 Election of EC Members of this document.

6.3 Directors

Four (4) persons who are individual Segment Members will represent the Services Segment on the RGQ Board of Directors. The Segment Membership will elect Directors as provided for in the RGQ Procedures and Section 10.1 Election of Board of Directors of this document.

7.0 Determination of Quorum

A quorum for purposes of amending these procedures, electing officers, or voting on any other Services Segment issues shall be at least 51% of the Segment Members.

8.0 Meetings

All meetings held in association with the Services Segment of RGQ are open to any interested company or organization. Only Segment Members, however, shall have a right to vote at such meetings.

9.0 Services Segment Notifications

9.1 Notices of Meetings

Notices of meetings will be posted to the Services Segment membership no less than seven (7) days prior to the date of the meeting, said notices providing the location, date, times and agenda topics of the meeting.

9.2 Notification Method

The Services Segment Chair will appoint a Segment Member to maintain a mailing list (and e-mail list) of Segment Members and will directly distribute information and notices on Segment meetings and issues to Services Segment Members.

Notification may additionally be provided by any of the means referenced in Article 11 of the NAESB Bylaws.

9.3 Notices of Elections

Notices of elections shall be posted as provided for in Sections 10.1 Election of Directors and 10.2 Election of EC Members of these Procedures.

10.0 Election, Resignation, and Removal of Services Segment Officers

10.1 Election of Directors

- a) The Services Segment of the RGQ shall be represented on the NAESB Board of Directors by four individuals who shall be elected, from time to time, as required by Article 7 of the NAESB Bylaws and in this Exhibit 4 to those Bylaws.
- b) Each Director from the Services Segment of the RGQ shall serve a term of two years on the NAESB Board, except as provided for in Section 10.6 Transitional Election of Officers of this procedure or unless provided for otherwise in the RGQ Procedures.
 - Directors elected in a year ending in an odd digit (i.e., 2003), whose term would expire in a year ending in an odd digit (i.e., 2005), will serve as Group A Directors; there will be two Group A Director seats at any given time.
 - (ii) Directors elected in a year ending in an even digit, whose term would expire in a year ending in an even digit, will serve as Group B Directors; there will be two Group B Director seats at any given time.
- c) To be eligible to serve as a Services Segment representative on the NAESB Board of Directors, the individual must
 - (i) be a Services Segment Member

- (ii) have a working knowledge of the NAESB process;
- (iii) be willing to commit the time and resources necessary to fulfill their obligations as a NAESB Board member and to meet the minimum threshold of participation and attendance established in the NAESB Bylaws
- (iv) disclose the legal name for their employer and if applicable, its ownership, sister companies also under the umbrella of its owner, and a list of other entities for which the employer is engaged in a legal partnership or joint venture.
- (v) provide a copy of their employer's mission statement or a summary of their business objectives along with a brief statement regarding which segment these objectives best serve;
- d) No individual representing any single Segment Member shall serve on the Board of Directors and on the RGQ Executive Committee concurrently.
- e) Interested individuals duly authorized to represent a Segment Member and eligible to serve on the Board of Directors shall submit their names to the NAESB Office in a form as specified in NAESB Operating Procedures or,
 - (i) if such requirements are not provided for in the NAESB Operating Procedures, interested individuals shall submit a written statement to the NAESB Office that provides their name, title and affiliation (if any), business mail and email addresses, business telephone, a statement that they accept the conditions referred to in subsection (c)-(iii) above, and the statement required in subsection (c)-(iv) above.
- f) In preparation for any election of NAESB Directors (other than initial Directors, as provided for in Section 10.6) representing the Services Segment of the RGQ, a Nominating Committee consisting of five (5) existing Board and EC members of the RGQ Services Segment, shall be appointed by the Services Segment Chair. This committee shall develop a slate of candidates from interested members of the Segment Membership. Other nominations may be made at or prior to the

h)

close of the RGQ Services Segment nomination period by any Segment Member in accordance with the requirements in subsection (e) above. All nominations must be made and conveyed in writing to the NAESB Secretary no less than 31 days prior to the election date.

- g) The election process for RGQ Services Segment representatives to the NAESB Board shall be
 - coordinated by the Vice-Chair of the Segment with appropriate coordination with the Quadrant Executive Committee and the NAESB Office such that elections occur during the second week of November of each year
 - (ii) announced by formal notice to the Segment Membership no less than 30 days prior to the date of the election, with such notice indicating the name and affiliation of all candidates.

Elections for RGQ Services Segment Directors shall be subject to the following provisions:

- (i) Any Services Segment Member of the RGQ with dues current is eligible to cast a vote.
- (*ii*) Services Segment Members shall cast votes for candidates in a number not exceeding the number of Director seats open for election from the Services Segment.
- (iii) The candidates receiving the greatest numbers of votes shall be elected.
- (iv) Results of the RGQ elections for the NAESB Board, including the returned ballots, shall be forwarded to the Secretary of NAESB, for the purpose of public record and validation of voting results, upon conclusion of any election.
- Any Director seats which become vacant shall be filled in accordance with RGQ Procedures as specified in this Exhibit 4 to the Bylaws, or if such provisions are lacking, by special election, subject to the following provisions
 - (i) If the vacancy occurs with less than 125 days remaining in the Director's term of office, the seat will remain vacant until filled by election of a new

Director to a term commencing in the next operating year.

(ii) All other vacancies will be filled by special election; such special elections are to be formally noticed to the Segment Membership by the Segment Vice-Chair or Chair within 15 days of when the creation of the vacancy becomes known and held within 60 days of the creation of the vacancy. Candidates will be designated by the normal nomination process or selected by the remaining Board members of the Segment from Segment Membership. Candidates elected to fill the vacancy will serve the remainder of the term, and may stand for re-election.

10.2 Election of EC Members

- a) The Services Segment of the RGQ shall be represented on the NAESB EC by four individuals who shall be elected, from time to time, as required by Article 10 of the NAESB Bylaws and in this Exhibit 4 to those Bylaws.
- b) Each EC member from the Services Segment of the RGQ shall serve a term of two years on the RGQ EC, unless provided for otherwise in the RGQ Procedures.
 - (i) Members elected in a year ending in an odd digit (i.e., 2003), whose term would expire in a year ending in an odd digit (i.e., 2005), will serve as Group A representatives; there will be two Group A representative seats at any given time.
 - (ii) Members elected in a year ending in an even digit, whose term would expire in a year ending in an even digit, will serve as Group B representatives; there will be two Group B representative seats at any given time.
 - c) To be eligible to serve as a Services Segment representative on the EC, the individual must
 - (i) be a Segment Member
 - (ii) have a working knowledge of the NAESB process;
 - (iii) be willing to commit the time and resources necessary to fulfill their obligations as a NAESB EC

member and to meet the minimum threshold of participation and attendance established in the NAESB Bylaws

- (iv) disclose the legal name for their employer and if applicable, its ownership, sister companies also under the umbrella of its owner, and a list of other entities for which the employer is engaged in a legal partnership or joint venture
- (v) provide a copy of their employer's mission statement or a summary of their business objectives along with a brief statement regarding which segment these objectives best serve;
- d) No individual representing any single Segment Member shall serve on the Board of Directors and on the RGQ Executive Committee concurrently.
- e) Interested individuals duly authorized to represent a Segment Member and eligible to serve on the EC shall submit their names to the NAESB Office in a form as specified in NAESB Operating Procedures or,
 - (i) if such requirements are not provided for in the NAESB Operating Procedures, interested individuals shall submit a written statement to the NAESB Office that provides their name, title and affiliation (if any), business mail and email addresses, business telephone, a statement that they accept the conditions referred to in subsection (c)-(iii) above, and the statement required in subsection (c)-(iv) above.
- f) In preparation for any election of EC members (other than initial EC members, as provided for in Section 10.6) representing the Services Segment of the RGQ, a Nominating Committee consisting of five (5) Board or EC members of the RGQ Services Segment, shall be appointed by the Services Segment Chair. This committee will develop a slate of candidates from interested members of the Segment Membership. Other nominations may be made at or prior to the close of the RGQ Services Segment nominating period by any Segment Member in accordance with the requirements in subsection (e) above. All nominations must be made and conveyed in writing to the NAESB

Secretary no less than 31 days prior to the election date.

- g) The election process for RGQ Services Segment representatives to the RGQ EC shall be
 - (i) coordinated by the Vice-Chair of the Segment with appropriate coordination with the RGQ EC and the NAESB Office such that elections occur during the second week of November of each year
 - (ii) announced by formal notice to the Segment Membership no less than 30 days prior to the date of the election, with such notice indicating the name and affiliation of all candidates.
- h) Elections for RGQ Services Segment EC members shall be subject to the following provisions:
 - (i) Any Services Segment Member of the RGQ with dues current is eligible to have one duly authorized representative cast the Member's vote.
 - (ii) Services Segment Members shall cast votes for candidates in a number not exceeding the number of EC seats open for election by the Services Segment.
 - (iii) The candidates receiving the greatest numbers of votes shall be elected.
 - (iv) Results of the elections for the RGQ EC, including the returned ballots, shall be forwarded to the Secretary of NAESB, for the purpose of public record and validation of voting results, upon conclusion of any election.
- Any EC seats becoming vacant shall be filled in accordance with RGQ Procedures as specified in this Exhibit 4 to the Bylaws, or if such provisions are lacking, by a special election subject to the following provisions
 - (i) If the vacancy occurs with less than 125 days remaining in the EC member's term of office, the seat will remain vacant until filled by election of a new EC member to a term commencing in the next operating year, and a Designated Alternate will serve until expiration of the term.

NAESB Bylaws Addendum Exhibit 4-B Retail Gas Quadrant Services Segment Procedures Draft Last Revised: 01/14/02

- (ii) All other vacancies will be filled by special election; such special elections are to be formally noticed to the Segment Membership by the Segment Vice-Chair or Chair within 15 days of when the creation of the vacancy becomes known and held within 60 days of the creation of the vacancy. Candidates will be designated by the normal nomination process or selected by the remaining EC members of the Segment from Segment Membership. Candidates elected to fill the vacancy will serve the remainder of the term, and may stand for reelection.
- (iii) Until an election is held and a new EC member is elected, a Designated Alternate will serve in the open EC seat.

10.3 Resignation of Directors or EC Members

A Director or EC Member may resign his or her position by submitting a letter to the Secretary of NAESB with a copy to the Chair of the Services Segment of the RGQ stating that he or she is resigning and giving the effective date of the resignation.

10.4 Removal of Directors or EC Members

- a) A NAESB Director or EC Member shall cease to hold that position upon
 - (i) the resignation of RGQ Services Segment Membership by the company of which the Director or EC Member serves as a representative
 - (ii) the lapse of the Segment Membership dues of the entity which the Director or EC Member serves as a representative, or
 - (iii) the Director or EC Member's resignation, removal, or death
 - (iv) the first day of the NAESB operating year following an election for the seat held by the Director or EC Member, wherein the Director or EC Member was not re-elected by the Segment Membership as declared by the NAESB Secretary.
- b) Receipt by the NAESB Secretary of a rescission letter indicating that the individual no longer represents the

Member does not constitute grounds for removal of the Director if all three of the following conditions are satisfied:

- (i) the individual continues activity or employment in the Services Segment of the Gas industry with another Services Segment Member,
- (ii) the individual's new employer is a Segment Member with no representatives holding seats on the Board, and
- (iii) the Board determines that completion of the Director's term of office is in the best interest of NAESB
- c) A vacancy resulting from removal of a Director shall be filled for the remainder of that term in accordance with Section 10.1(i) of these Procedures.

10.5 Designation of Alternates for EC

- a) Selection of Designated Alternates: The Services Segment will select no more than three (3), and no less than one Designated Alternate EC representative in each election year, allowing for up to three (3) Designated Alternates overall.
- b) Designated Alternates will be selected by the same election that determines EC members, with those unsuccessful EC member candidate(s) receiving the greatest number of votes becoming a Designated Alternate in the event that there already are or are expected to be less than three Designated Alternates from the Services Segment in the coming year.
- c) The first, second and third Designated Alternate will be determined by
 - the order in which they become named as a Designated Alternate, as indicated by the first year of their designation and
 - (ii) the order of voting results.
- d) A Designated Alternate will attend and vote at meetings of the EC of the RGQ or of the larger EC of NAESB, when one or more regular EC Members from the Services Segment are unable to attend a meeting of the EC and when the absent member or the Vice-Chair of

the RGQ EC requests their attendance as a Designated Alternate. Selection of which Alternate will serve will proceed through the order of first, second or third Designated Alternate, with each indicating whether they can or cannot fulfill the requirement to serve at the subject meeting.

10.6 Transitional Election of Officers

- a) The initial Directors and EC Members representing the Services Segment shall be elected at the first meeting of the Segment.
- b) The candidates receiving the greatest number of votes shall be elected.
- c) The two (2) initial Directors and EC Members receiving the greatest number of votes shall serve for two years in Group B positions as described in Sections 10.1 (b) and 10.2 (b) of this procedure. The remaining two (2) initial Directors and EC Members shall serve for one year in Group A positions as described in Sections 10.1 (b) and 10.2 (b) of this procedure.

11.0 Amendment of Services Segment Procedures

The Directors representing the Services Segment on the Board shall recommend to the Segment Membership amendments to these Procedures. Each proposed amendment shall be offered as a separate amendment, except in those situations in which an amendment will affect more than one section of these Procedures. If a proposed amendment will affect more than one section, then all affected sections will be identified.

A majority vote of the Segment Membership shall be required to approve an amendment to these procedures.

TAB 8

NEXT STEPS

[Letterhead of Prospective Member]

Ms. Rae McQuade Executive Director Gas Industry Standards Board 1100 Louisiana Suite 3625 Houston, TX 77002

Re: Letter of Intent to Join North American Energy Standards Board

Dear Ms. McQuade:

[Name of company] hereby gives notice of its intent to join the North American Energy Standards Board ("NAESB") in recognition of the following actions taken by the Board of Directors of the Gas Industry Standards Board ("GISB"):

- 1. Adoption of a resolution at its September 19, 2001 Board meeting changing the scope of GISB in its Certificate of Incorporation to include development of standards for wholesale and retail electric transactions;
- 2. Adoption of a resolution at its December 5, 2001 Board meeting changing the name of GISB to the North American Energy Standard Board to reflect the expanded mission of the organization as of January 1, 2002; and
- 3. Adoption of revised bylaws at its December 5, 2001 Board meeting to reflect the change of name and broader mission of the organization as NAESB in substantially the form provided to the GISB Board for the December 5, 2001 meeting, as published on GISB's website.

[Name of company] agrees to tender its membership dues of \$5,000, along with a designation of a Quadrant and Segment in which it would like to be a member within 30 days of the date of this letter, upon the occurrence of: (a) ratification by the GISB membership of the changes to the GISB Certificate of Incorporation described in Items 1 and 2 above to transform the organization into NAESB, and (b) filing of such changes with the Secretary of State of Delaware. We understand that ratification should be completed by January 1, 2002 and that the filing of the

amended and restated Certificate of Incorporation of NAESB with the Secretary of State of Delaware will take place immediately thereafter.

Very truly yours,

(Name)

(Title)

(Name of Company)

NORTH AMERICAN ENERGY STANDARDS BOARD MEMBERSHIP APPLICATION

Name of Membership:	Date of Membership		
. (Legal name for NAESB records)		
Mailing Address:			
City, State, Zip:			
Name of Contact Person:	Title:		
(Not applicable to personal memberships)			
Mailing Address:	Telephone #:		
(If different than above)	Fax #:		
Email Address:	_		
***Please indicate appropriate segment of industry covered by this membership: (one per membership fee)			
Quadrant Segment			
(Wholesale Gas, Electric – Retail Gas, Electric)			
Quadrant Segment			
(Wholesale Gas, Electric – Retail Gas, Electric)			
***Membership Fee: billing in the amount of \$5000 (per membership) will be sent to you. Payment is requested within 30 days of billing.			
Mail to: North American Energy Standard Board, 1100 Louisiana, Suite 3625, Houston, Texas 77002 Fax to: 713-356-0067 – email: naesb@aol.com			
OR: American Express Visa Master Card	Card # Exp Date		
Signature:	Date:		

NOTICE. To complete this application, please execute and return the attached form of revocable proxy) required by Article IV, §2 of the Certificate of Incorporation of NAESB, which requires each voting member, "as a condition of membership, ….. to execute a revocable appointment in a form approved by the Board of Directors authorizing a designated proxy to vote in support of any of the proposals adopted by the Board of Directors described in Article V, §3 of this Certificate of Incorporation which may be submitted to a vote of the general membership." Article V, §3 deals with votes to amend the Certificate of Incorporation or to determine the continuation of the corporation after December 31, 2004.

NORTH AMERICAN ENERGY STANDARDS BOARD, INC.

REVOCABLE PROXY

In accordance with Article IV, Section 2 of the Certificate of Incorporation of the North American Energy Standards Board, Inc., a Delaware non-stock corporation (the "Corporation"), the undersigned member of the Corporation hereby appoints the Chairman of the Board of Directors of the Corporation and/or its Executive Director ("Proxy") as the proxy for the undersigned, with full power of substitution, to attend any and all meetings of the members of the Corporation called for the purpose of voting on any proposals adopted by the Board of Directors described in Article V, Section 3, of the Corporation's Certificate of Incorporation (dealing with amendment of the Certificate of Incorporation or the continuation of the Corporation after December 31, 2004), and to vote in favor of any such proposals in the name of the undersigned at any and all meetings of the members or any adjournment thereof, except as provided below. The undersigned hereby revokes all prior proxies for such meetings, and ratifies and confirms all that the Proxy may lawfully do or cause to be done by virtue hereof.

This proxy shall be valid and in full force, and shall not terminate or expire, until revoked by the undersigned; <u>provided</u>, <u>however</u>, that the undersigned may attend any and all meetings of the members of the Corporation and vote in lieu of the Proxy on any proposals described in Article V, Section 3, of the Corporation's Certificate of Incorporation. If the undersigned is present and votes on any such proposals, such vote will be valid for all purposes but will not serve to revoke this proxy as to any subsequent vote of members on any proposal described in Article V, Section 3, of the Corporation's Certificate of Incorporation.

Executed this _____ day of _____2002.

ATTEST:

Name of Member Company

Secretary

Voting Representative's Signature

Print Name

Title

NAESB QUADRANTS & SEGMENTS FORMATION CONTACT

NAESB Quadrants

Retail Electric

Wholesale Electric, Retail Electric Technology

Retail Gas

Charlie Driggs 302-429-3194 charlie.driggs@conectiv.com

Craig Goodman 202-333-3288 cgoodman@energymarketers.com

Mariam Arnaout 202-824-7127 marnaout@aga.org

Wholesale Gas

Jim Buccigross 508-238-0345 jhb@8760.com

NAESB Retail Electric Segments

Retail Electric Suppliers

Retail Electric Distributors

Retail Electric End Users

Retail Electric Services

Eric Ackerman 202-508-5528 eackerman@eei.org

Misty Ann Khan 713-463-9661 khan@telerx.com Jim Minneman 610-774-5774 jmminneman@pplweb.com

NAESB Retail Gas Segments

Retail Gas Suppliers

Retail Gas Distributors

Retail Gas End Users

Retail Gas Services

Mariam Arnaout 202-824-7127 marnaout@aga.org

Misty Ann Khan 713-463-9661 khan@telerx.com Jim Minneman 610-774-5774 jmminneman@pplweb.com